

BODY MASS INDEX AND ITS EFFECT ON PLANTAR PRESSURE IN OVERWEIGHT AND OBESE ADULTS

A Thesis submitted by

Suhad Kareem Rahi Al-Magsoosi

B.Sc., M.Sc.

For the award of

Doctor of Philosophy

2019

ABSTRACT

The proportion of overweight or obese adults is creating a growing problem throughout the world. Overweight and obesity have a significant influence on gait, and often cause difficulty. There is evidence to suggest that being overweight or obese places adults at a greater risk of developing foot complications such as osteoarthritis, tendonitis, plantar fasciitis, and foot ulcers. Increasingly, pressure ulcers have become a serious health problem. The purpose of this research is to investigate the effect of body weight on the feet, and to investigate the use of simulated body mass to study the effect of variable body mass on the foot plantar in adults aged 24 to 50 years of age while walking at a self-selected pace. A series of studies were undertaken to achieve the above purpose. The research involved: 1) assessing dynamic foot plantar pressure characteristics in adults who are normal weight, overweight or obese; 2) studying the gait impact of increased simulated body weight (SBW); and 3) evaluating the spatial relationship between the trace of the centroid of the area of contact with heel strike, midstance, and toe-off phases for the SBW groups. F-Scan in-shoe systems were utilised to gather the foot pressure data.

The first study sought to investigate the effect of different body mass index (BMI) levels on plantar pressure distribution during walking, collection in fifteen voluntary participants were recruited. The BMI participants were divided into three groups (healthy, overweight and obese). The foot was divided into ten regions: heel (H), midfoot (MF), first metatarsal head (1MH), second metatarsal head (2MH), third metatarsal head (3MH), fourth metatarsal head (4MH), fifth metatarsal head (5MH), hallux (1stT), second toe (2ndT), and third to fifth toes (3rd-5thT). For each region, the following parameters were calculated: force (F), contact area (CA), contact pressure (CP), pressure time integral (PTI) and peak pressure (PP). The mean of the three repetitions of each subject was computed, and statistical procedures were performed with these mean \pm standard deviation (SD) values. This study showed that the obese group had higher plantar pressure parameter values compared to the other two groups (overweight and healthy) for the ten different foot regions. The study observed significant changes in the parameters in the H and MHs (e.g. 2MH and 3MH) foot regions. The forefoot appears to be more sensitive to weight-related pressure under the foot than the rearfoot. Findings from this study indicate that being overweight or

obese increases foot pressure measures, even for individuals with similar body features. Higher BMI values correlate with a higher load on the foot during walking in males. These findings have implications for pain and discomfort in the lower extremity in the obese while participating in activities of daily living such as walking.

The second study investigated the effect of the research methodology involving the simulation of body weight (SBW) with additional weight, adding 10, 20, 30 kg to each participant's body weight on plantar pressures. The sample comprised 31 adult males; each subject walked four times. The first walk was without any external weight (NBW, 0 kg), the second walk was with a weight of 10 kg, the third walk was with a weight of 20 kg and the last walk with a weight of 30 kg in the vest. The foot was divided into ten regions and for each region, the parameters were calculated the same way as the first study. At the end of this study it should be noted that SBW groups subjected to load have shown changes in foot plantar measure values compared to the NBW group. Most of the differences were found under H, MHs, 1stT and MF regions in the most clinically relevant parameters in SBW groups compared to the control group; the SBW groups showed higher values of plantar pressure. The results of the ICC showed a generally good to an excellent level of reliability, the quality of which was dependent on the regions of the foot and the variables investigated with SBW loads. This experiment pointed out that an insole pressure system is a reliable tool for evaluating foot plantar forces and pressures throughout the walk. The plantar pressure measures can be used in relative assessments, as the measures of repeatability are favourable for the measures and foot zones generally utilised in the study of people with clinical problems like neuropathic diabetics.

In the final study, associations were investigated of the centroid (coordinates x-axis and y-axis) of the area of contact captured between normal (NBW) and simulated body weight (SBW) changes. The same 31 adult males who enrolled with the SBW tests were used to collect the centroid of the area of contact with the surface. This was located by calculating the geometric centre of a set of cloud points having the lowest z coordinate value. In this part, a foot pressure sensing insole was used to calculate the moment of heel strike, midstance and toe-off phases. Data were analysed descriptively (mean \pm SD only). The outcome of this study, relating to specific individual characteristics of the centroid trace of the plantar contact area was

compared with the heel strike, midstance, and toe-off phases for the SBW group with the NSBW group. X-axis and y-axis coordinates in the heel strike, midstance and toe-off phases under SBW with 30, 20, 10 kg had higher mean values compared to NSW. The x-axis and y-axis coordinates had mean values of 11.76, 9.68, and 7.76 mm; while the y-axis coordinates had mean values of 11.96, 9.89, and 8.18 mm. Moreover, x-axis and y-axis coordinates were assessed in the midstance phase under SBW with 30, 20, 10 kg with means of 6.59, 5.48, and 4.50 mm; while the y-axis coordinates had mean values of 6.38, 5.41, and 4.41 mm. In addition, x-axis and yaxis coordinates were assessed in the toe-off phase under SBW (30, 20, 10 kg) with mean values of 11.56, 9.67, and 7.97 mm; while the y-axis coordinates had mean values of 11.51, 9.39, 8.02 mm, respectively. X-axis and y-axis coordinates had mean values in relation to NBW in three phases: heel strike of 5.47 and 6.15; midstance of 2.99 and 3.05; and toe-off of 6.04 and 5.82, respectively. The x-locate and y-locate change can be calculate the change in rotation of the ankle joint. As the data was normalised according to the total time taken for the loading phase of the gait, the y-locational change was due partly to the extra weight, which could increase the time of lifting the foot. Therefore, the results showed that the x-locate and ylocate change can help to calculate the change in the rotation of the ankle joint.

The project has shown that it is possible to demonstrate that obese people will, throughout their lives, adopt ways to effectively execute a particular activity. This finding provides a foundation for future clinical trials which could assist in preventing foot complications and could assist in the design of appropriate interventions to promote healthy outcomes for these adults. The simulated body weight resulted in a variation in plantar pressure distribution. Because the human foot adapts itself to any simulated condition, knowledge of the variation of pressure distributions of both feet can provide input for suitable guidelines for biomedical engineers. To promote the prevention of likely injury to the feet of overweight and obese people, the results of this study demonstrate the need to develop strategies which could include the building of an insole (orthosis) that absorbs foot plantar pressure.

Certification of Thesis

This thesis is entirely the work of <u>Suhad Kareem Rahi Al-Magsoosi</u> except where otherwise acknowledged. The work is original and has not previously been submitted for any other award, except where acknowledged.

Dr. Albert Chong

Principal Supervisor: _____

Dr. Zahra Gharineiat

Associate Supervisor: _____

Student's and supervisors' signatures of endorsement are held at the University.

ACKNOWLEDGEMENTS

I am highly grateful to my thesis advisors Dr Albert K. Chong, and Dr Zahra Gharineiat for their advice, guidance, mentorship and persistent hard work in shaping my professional growth and my appreciation to them for their insightful remarks, valuable comments and ideas for all these years. I am extremely thankful to them for their invaluable suggestions and constant encouragement.

This research was made possible through the financial assistance of the Iraqi Government represented by the Ministry of Higher Education and Scientific Research (MoHESR).

Many thanks must go to Mr Kaled Aljebur, Dr Jasim Ahmed AL-Baghdadi, Dr Mahmood Hussain, Dr Shahab Abdulla, Mr Ali Al-Kharaz for the significant contribution and permanent assistance at all stages of the research and experimental work.

I also wish to acknowledge the editorial support provided by the English editor Dr Barbara Harmes and also my supervisory team in reviewing the earlier versions of this thesis for grammar and syntax.

I would like to thank the administrative staff at the University of Southern Queensland a special thanks to Mrs Juanita Ryan, Lesley Astbury, and Mrs Rebecca Darr for their various forms of support during my PhD study.

I would like to thank the Australian Commonwealth government's contribution for the financial support through the Research Training Program (RTP) -fees offset.

I also thank all of the participants in the various measurement in my studies. None of the studies would have been possible without them. It was an immense pleasure to meet you and the fleeting discussions we had shall always be treasured.

Finally, I would like to dedicate this dissertation to my family: my parents, sisters and brothers who have been a continual source of inspiration, support and encouragement. Without them all I may not have successfully made it to the end of this journey.

CONTRIBUTIONS TO THEORY AND PRACTICE FROM THIS RESEARCH

This research covers a relatively wide range of topics relating to overweight and obesity and lower limb foot plantar sole issues, sustainability, causes, consequence and insole system technique. They have an increased susceptibility to overuse injuries and the tendency to withdraw from weight management programs that involve physical activity. Therefore, we believe that before overweight/obese adults join any physical activity weight-management program. A summary of the main contributions to practice arising from this research include:

- An in-depth understanding of the foot and ankle in terms of balance and injury.
- Data collection from 46 subjects with different body weight (normal, overweight and obese) and normal self-select gait.
- Consideration of the efficiency of insole systems.
- A novel approach to simulating the long-term relationships between body weight and foot plantar pattern. This simulation approach appears to be robust and may be used to assist further studies in this space and to assist in decision-making.
- Practical recommendations (based on laboratory experiments) for foot protection and management to increase human health.
- The outcomes of this study can be used to enable further research into areas like injury prevention and balance monitoring in walking based on weight.

LIST OF RELATED PUBLICATIONS

JOURNAL

Suhad K. R. Al-Magsoosi & Albert K. Chong, 2019. 'Foot loading pattern variations between normal weight, overweight, and obese adults aged 24 to 50 years'. *Journal of Biosciences and Medicines*. Vol.07 No.05, pp.34-49. DOI: 10.4236/jbm.2019.75007.

Albert K. Chong & **Suhad K. R. Al-Magsoosi**, 2019. 'Development of an Index for Drop-Foot Severity of DPN Patients'. *Journal of Biosciences and Medicines*. Vol.7 No.5, PP. 61-64. DOI: 10.4236/jbm.2019.75009.

CONFERENCE

Al-Magsoosi, S.K. & Chong, A.K., 2019, March. Advanced Foot Pressure Technology for Demonstrating the Influence of Over-weight and Obesity on Ankle Joint. In Proceedings of the ICISS 2019 Proceedings of the 2019 2nd International Conference on Information Science and Systems. Tokyo, Japan. (pp. 195-199). ACM. DOI>10.1145/3322645.3322661.

TABLE OF CONTENTS

| AB | STRACT | i |
|----------|---|---|
| AC | KNOWLEDGEMENTS | v |
| CO RE | NTRIBUTIONS TO THEORY AND PRACTICE FROM THIS | S ix |
| LIS | ST OF RELATED PUBLICATIONS | X |
| TA | BLE OF CONTENTS | xi |
| LIS | T OF FIGURES | XV |
| LIS | ST OF TABLES | xviii |
| AB | BREVIATIONS | xxi |
| 1. | INTRODUCTION | 1 |
| 2. | 1.1. Introduction 1.2. General overview | 1 1 3 4 4 6 7 8 11 11 12 12 13 13 15 |
| | 2.1. Introduction 2.2. Overweight and obesity | 15 19 20 20 24 27 29 31 32 |
| | | |

| | 2.5. Planta | ır pressure | 34 |
|----|---------------|--|----|
| | 2.5.1. | Plantar load distribution: an overview | 36 |
| | 2.5.2. | Foot plantar pressure mask or anatomical regions | 39 |
| | 2.5.3. | Plantar pressure distribution | 42 |
| | 2.6. Pressu | are measurement techniques | 48 |
| | 2.6.1. | Pressure and ground reaction forces measuring systems | 48 |
| | 2.6.2. | Foot pressure measurement systems | 49 |
| | 2.6.3. | In-shoe pressure measurements | 51 |
| | 2.6.4. | Sensor calibration | 55 |
| | 2.7. Discu | ssion | 56 |
| | 2.8. Chapt | er summary | 57 |
| 3. | METHO | DOLOGY | 59 |
| | 3.1 Introd | luction | 59 |
| | 3.2 Instru | mentation and software | 59 |
| | 3.2.1. | In-shoe sensor | 59 |
| | 3.2.2. | Shoes | 63 |
| | 3.2.3. | F-Scan software | 63 |
| | 3.2.4. | Wall stadiometer | 64 |
| | 3.2.5. | Body composition scanner scale | 64 |
| | 3.2.6. | IBM SPSS statistics | 65 |
| | 3.2.7. | Iron mass | 65 |
| | 3.2.8. | Army vest | 66 |
| | 3.3 Unive | rsity approval and experiments | 66 |
| | 3.3.1 | Ethical approval | 66 |
| | 3.3.2 | Ethical issues | 67 |
| | 3.3.3 | Participants | 67 |
| | 3.4 Exper | iment design | 71 |
| | 3.4.1 | Experiment 1: foot loading pattern variations between healthy, | |
| | overweigh | t, and obese adults | 71 |
| | 3.4.2 | Experiment 2: simulation body weight protocol and | |
| | determination | tion of plantar pressure | 72 |
| | 3.4.3 | Experiment 3: profiling the centroid of the area of contact | |
| | captured | 75 | |
| | 3.5 The n | ethod used in data analysis | 77 |
| | 3.5.1 | The coefficients of variation: | 77 |
| | 3.5.2 | Analysis of variance | 77 |
| | 3.5.3 | Intra-class correlation coefficients | 78 |
| | 3.6 Analy | sis of experiments | 81 |
| | 3.6.1 | Analysis of experiment 1 | 81 |
| | 3.6.2 | Analysis of experiment 2 | 81 |
| | 3.6.3 | Analysis of experiment 3 | 82 |
| | 3.7 Assur | nptions | 82 |
| | 3.8 Chapt | er summary | 83 |
| 4. | RESULT | OF EXPERIMENT 1 | 85 |
| | 4.1 Introd | | |
| | 411 F | Force and contact area measurements | 05 |
| | 1.1.1.1 | sice and conduct area mousurements | 05 |

| | 4.1.2. 0 | Contact pressure and pressure-time integrals | 89 |
|----|-------------|--|-------|
| | 4.1.3. P | eak pressure | 92 |
| | 4.2 Discu | ssion | 93 |
| | 4.2.1. | Force and contact area | 94 |
| | 4.2.2. | Contact pressure and pressure-time integrals | 96 |
| | 4.2.3. | Peak pressure | 97 |
| | 4.3 Chapt | er summary | 98 |
| _ | | | |
| 5. | RESULT | OF EXPERIMENT 2 | 99 |
| | 5.1. Introd | uction | 99 |
| | 5.1.1. | Force | . 102 |
| | 5.1.2. | Contact area | . 105 |
| | 5.1.3. | Contact pressure | 107 |
| | 5.1.4. | Pressure time integral measurement | 109 |
| | 5.1.5. | Peak pressure | 111 |
| | 5.2. Repea | tability of the F-Scan system | 112 |
| | 5.2.1. | Force variable | 113 |
| | 5.2.2. | Contact area variable | |
| | 5.2.3. | Contact pressure variable | |
| | 5.2.4 | Pressure time integral variable | 119 |
| | 5.2.5 | Peak pressure | 121 |
| | 5.3. Discu | ssion | 123 |
| | 5.3.1 | Force variable | 123 |
| | 5.3.2 | Contact area variable | 124 |
| | 533 | Contact pressure variable | 124 |
| | 534 | Pressure time integral variable | 124 |
| | 535 | Peak pressure variable | 125 |
| | 536 | Repeatability of the F-Scan system | 126 |
| | 5.4. Chapt | er summary | 127 |
| | | | |
| 6. | RESULT | OF EXPERIMENT 3 | . 129 |
| | 6.1. Introd | uction | 129 |
| | 6.2. The c | entroid of heel strike contact surface with the simulated body | |
| | weight 130 | | |
| | 6.2.1. | X-axis coordinate of heel strike | . 130 |
| | 6.2.2. | Y-axis coordinate of heel-strike | 132 |
| | 6.3. The c | centroid of midstance contact surface for the simulated body | |
| | weight 134 | | |
| | 6.3.1. | X-axis coordinate of midstance | 134 |
| | 6.3.2. | Y-axis coordinate of midstance | 136 |
| | 6.4. The c | entroid of toe-off contact surface for simulated body weight | |
| | 6.4.1 | X-axis coordinate of toe-off | |
| | 6.4.2 | Y-axis coordinate of toe-off | |
| | 6.5 Discu | ssion | . 142 |
| | 66 Centre | bid of the foot plantar during gait | 142 |
| | 661 | Centroid of heel strike contact surface for the simulated body | . 174 |
| | weight | 142 | |
| | weight | 1 14 | |

| | 6.6.2. The centroid of midstance contact surface for the simulated | |
|-----|--|------|
| | ody weight | 143 |
| | 6.6.3. The centroid of toe-off contact surface within the simulation | |
| | ody weight | 143 |
| | 5.7. Chapter summary | 143 |
| | | |
| 7. | CONCLUSIONS | 145 |
| | | |
| | 1. Introduction | 145 |
| | 2. Overall conclusions | .147 |
| | '.3. Limitations of the research | 148 |
| | .4. Future research | 149 |
| DFF | DENCES | 151 |
| КСГ | RENCES | 131 |
| APP | NDICES | 175 |
| | Appendices A - Human Research Ethics Committee] | 175 |
| | Appendices B - Consent Form] | 176 |
| | Appendices C - The calibration procedure] | 177 |
| | representation procedure j | |

LIST OF FIGURES

| Figure 2.1. Prevalence of obesity worldwide in males 15 to 100 years old | | |
|--|--|--|
| in 2010. This figure represents the prevalence of obesity $(BMI > 30)$ | | |
| kg/m ²) among males on a global scale in the year 2010 (Haidar & | | |
| Cosman, 2011)17 | | |
| Figure 2.2. Classification of BMIs human body requirements for each | | |
| group (from CDC, 2014) | | |
| Figure 2.3. The foot connects with the lower leg at the talus, tibia, and | | |
| fibula interface (Lochner, 2013) | | |
| Figure 2.4. View of the dorsal bones of the rearfoot, midfoot, and | | |
| forefoot (adapted from (Oatis, 2009) | | |
| Figure 2.5. Three different types of human foot arch and the contact area | | |
| of each type (Kraushaar, 2005) | | |
| Figure 2.6. The normal gait cycle of an 8-year old boy (Perttunen, 2002) 32 | | |
| Figure 2.7. Systems in normal gait laboratory, utilizing photocells for | | |
| synchronizing both camera system and force platforms. Further, | | |
| accelerometers, electromyography, and pressure insoles can be used | | |
| (Lyytinen et al., 2014) | | |
| Figure 2.8. The plantar pressure platform registers several steps while | | |
| the patient walks barefoot a- foot plantar pressure system by Zebris | | |
| Medical GmbH. b- plantar pressure sensor emed® by Novel (cited by | | |
| Razak et al., 2012) | | |
| Figure 2.9. Layers of the plantar skin | | |
| Figure 2.10. Mean and SD for the peak plantar pressure during walking | | |
| (Perttunen, 2002) | | |
| Figure 2.11. Illustration of the variety of foot regions | | |
| Figure 2.12. Ten anatomical regions of the foot defined for plantar | | |
| pressure distribution analysis. M01: Heel, M02: Medial midfoot, M03: | | |
| 1 MT Head, M04: 2 MT Head, M05: 3 MT Head, M06: 4 MT Head, | | |
| M07: 5 MT Head, M08: Hallux, M09: M10: Toes (Putti et al., 2008) 42 | | |
| Figure 2.13. (a) Ground reaction forces measuring system; (b) Pressure | | |
| measuring systems (Chong et al., 2015; Al-Magsoosi & Chong, 2019) 49 | | |

| Figure 2.14. The matrix of force resistance sensors (Chong et al., 2015) 50 |
|--|
| Figure 2.15. In-shoe pressure systems (Chong et al., 2015) |
| Figure 2.16. Platform, walkway or mat systems (Orlin & McPoil, 2000; |
| AL-Baghdadia et al., 2015)55 |
| Figure 3.1. In-shoe pressure system (pressure sensor). (A) Flexible, thin |
| in-shoe sensors. (B) Inserting the insole system inside the shoe (C) |
| Represents how each participant was set up with the F-Scan system |
| prior to each calibration and recording sensors. (D) Represents a |
| typical plantar pressure recording obtained using the insole system, |
| where high pressure areas can be easily identified |
| Figure 3.2. FScan software 63 |
| Figure 3.3. Wall stadiometer |
| Figure 3.4. Body composition scanner scale |
| Figure 3.5. IBM SPSS statistics version 25 65 |
| Figure 3.6. Show the piece of metal |
| Figure 3.7. Army vest jacket |
| Figure 3.8. Plantar foot divided into ten regions |
| Figure 3.9. Participants were grouped within three categories based on |
| their BMI: (A) healthy, (B) overweight, and (C) obese72 |
| Figure 3.10. Modification of volunteer body weight by adding weight |
| evenly distributed anteriorly and posteriorly using vests. State one |
| walk without any additional weight; state two (10 kg); State three (20 |
| kg); State four (30 kg): (A) Front view, (B) Back74 |
| Figure 3.11. Determining the centroid of the heel strike during gait. The |
| one-step sequence of gait cycle phases which start from heel strike, |
| midstance and toe-off phases77 |
| Figure 4.1. The relationship between force (N) measure and BMI |
| (kg/m²) |
| Figure 4.2. The relationship between the contact area (cm ²) measure and |
| BMI (kg/m ²) |
| Figure 4.3. The relationship between contact pressure (KPa) measure |
| and BMI (kg/m ²)91 |
| Figure 4.4. The relationship between pressure time integral (kPa/s) |
| measure and BMI (kg/m ²) |

| Figure 4.5. Bar chart for healthy weight, overweight and obese, | | |
|---|-----|--|
| representing the variation in the ten foot regions | 93 | |
| Figure 5.1. The average effect of simulated body weight on foot plantar | | |
| pressure (A- force, B- contact area, C- contact pressure, D- pressure | | |
| time integral, and E- peak pressure) | 102 | |
| Figure 5.2. Comparison of peak pressure parameter among simulated | | |
| body weight under four weighted simulations for ten regions of the | | |
| foot. The next section focuses on the coefficients of repeatability for | | |
| the insole system | 111 | |
| | | |

LIST OF TABLES

| Table 3.1. Physical characteristics of the subjects mean and SD |
|---|
| Table 3.2. Participant information for each of the three groups of |
| participants (Experiment 1)72 |
| Table 3.3. Participant characteristics of the study population |
| Table 3.4. Different cases for calculation ICC depending on model and |
| form of ICC79 |
| Table 4.1. Summary of the dynamic force (N) and contact area (cm ²) |
| data for the normal weight (n=5), overweight (n=5) and obese (n=5) |
| [Mean (SD) of three trials]86 |
| Table 4.2. Summary of the contact pressure (kPa) and pressure-time |
| integral (kPa/s) data for the healthy weight (n=5), overweight (n=5) |
| and obese (n=5) [Mean (SD) of three trials]90 |
| Table 5.1. Force measure with increased body weight for ten regions of |
| the foot |
| Table 5.2. Contact area measure with increased body weight for ten |
| regions of the foot106 |
| Table 5.3. Contact pressure measure with increased body weight for ten |
| regions of the foot108 |
| Table 5.4. Pressure time integral measure with increased body weight for |
| ten regions of the foot110 |
| Table 5.5. Intra-class correlation coefficients result for F-Scan system |
| regarding force variable114 |
| Table 5.6. Intra-class correlation coefficients result for F-Scan system |
| regarding contact area at baseline116 |
| Table 5.7. Intra-class correlation coefficients result for F-Scan system |
| regarding contact pressure at baseline118 |
| Table 5.8. Intra-class correlation coefficients result for F-Scan system |
| regarding pressure time integral at baseline120 |
| Table 5.9. Intra-class correlation coefficients result for F-Scan regarding |
| peak pressure at baseline122 |

| Table 6.1. The mean value and SD for left and right foot in Heel strike | | |
|---|-------|--|
| phase (x-axis coordinate, mm) | . 131 | |
| Table 6.2. The mean value and SD for left and right foot in Heel strike | | |
| phase (y-axis coordinate, mm) | 133 | |
| Table 6.3. The mean value and SD for left and right foot with midstance | | |
| phase (x-axis coordinate, mm) | . 135 | |
| Table 6.4. The of mean value and SD for left and right foot in midstance | | |
| phase (y- axis coordinate, mm) | . 137 | |
| Table 6.5. The mean value and SD for left and right foot in toe-off phase | | |
| (x-axis coordinate, mm) | .139 | |
| Table 6.6. The mean value and SD for left and right foot in toe-off phase | | |
| (y-axis coordinate, mm) | . 141 | |
| | | |

ABBREVIATIONS

| AIHW | Australian Institute of Health and Welfare |
|------------------------------------|--|
| ANOVA | Analysis of variance |
| BMI | Body Mass Index |
| CA | Contact area |
| CDC | Centres for Disease Control and Prevention |
| cm ² | Square Centimetre |
| СОМ | Centre of mass |
| COP | Centre of pressure |
| CP | Contact pressure |
| CV | The coefficients of variation |
| СТ | Contact time |
| F | Force |
| FTI | Force-time integrals |
| GRF | Ground reaction force |
| Н | Heel |
| H ₇ | Hertz |
| | Interclass Correlation Coefficient |
| IDD | Instant of peak pressure |
| Π I Kα | Kilogram |
| kDa | Kilopascals pressure measurement unit |
| k a k Da s | Kilopascals second |
| I f | L aft foot |
| | First metatarsal head |
| 2MH | Sacond metatarsal head |
| 2MH | Third metatarsal head |
| | Fourth mototorsal head |
| 4MH 5MH | Fourin metatarsal head |
| m | Motro |
| III ME | Midfoot |
| Moy | Maximum |
| mm | Millimotro |
| 111111 P | Number |
| II N/am^2 | Number |
| N/CIII | Newton |
| | Nermal hady weight 0 kg |
| | Normal body weight, 0 kg |
| | Peak force |
| PP: | Peak pressure |
| P1 DTL | Pressure time |
| PII: | Pressure time integral |
| P-value | level of significance within a statistical hypothesis test |
| RI | Right foot |
| r-value | Correlations value |
| SBW | Simulate body weight (by additional weight 10, 20, 30 kg) |
| abaa 2D | Standard deviation |
| | Statistical Package for the Social Sciences |
| | Hallux |
| 2 nd I | Second toe |
| 3 ¹⁴ -5 ¹¹ T | 3–5Toes |
| WHO | World Health Organization |
| % | Percentage |

1. INTRODUCTION

1.1. Introduction

This chapter will provide an outline of the background and research motivation by defining the research gaps and the research aims and objectives. It will also explore contribution of this research to the relevant body of literature. The chapter concludes with a brief thesis outline.

1.2. General overview

The increasing number of overweight and obese adults has become a significant public health concern worldwide and is associated with both increasing health care costs and clinical problems (Sheffler et al., 2014; Anandacoomarasamy et al., 2007). Increasing levels of obesity are associated with rising rates of diabetes (Mokdad et al., 2003). Furthermore, 41 million children under the age of five years and, over 340 million children and adolescents aged five to nineteen years, are overweight or obese (WHO, 2016).

Overweight and obesity are associated with numerous other medical conditions such as painful feet, flat feet, Charcot foot, foot pathology (ulcer), a high incidence of osteoarthritis, and symptomatic complaints in the joints of the lower extremities (Wearing et al., 2006; Stuck et al., 2008; Riskowski et al., 2011; Periyasamy et al., 2012).

Body weight is considered to have the most significant impact on the structure and functioning of the foot. Overweight and obesity and their influence on foot arches have been dealt with, especially in the context of flat foot incidence (Mickle et al., 2006; Villarroya et al., 2009; Chang et al., 2010).

A high prevalence of disabling foot pain and the incidence of foot deformities, poor foot health, and the presence of other pain like lower back, hip, leg, and knee pain, have been associated with adults who are overweight or obese (Butterworth et al., 2012; Mickle and Steele, 2015). More recent investigations by Butterworth et al. (2012) have found that the foot is not immune to the effects of

obesity, with a recent systematic review of 25 studies involving 93,224 participants concluding that obesity is strongly associated with non-specific foot pain in the general population.

A possible explanation for the relationship between obesity and foot pain is that excess body weight leads to greater mechanical loading of the foot. Indeed, a link between increased force and pressure under the foot and obesity has been reported (Hills et al., 2001). A case–controlled study of 80 subjects with chronic heel pain syndrome and 80 age- and gender-matched control subjects found that those with chronic heel pain syndrome were three times more likely to be obese and four times more likely be flatfooted (Irving et al., 2007).

Recently, foot and ankle pathologies such as plantar fasciitis (Riddle et al., 2003) and plantar heel pain (Irving et al., 2007; Prichasuk, 1994), have become more prevalent in adults due to increased body weight. Some research studies have investigated the relationship between plantar pressure and foot pain, finding a clear association between them. It has been demonstrated that load carriage can adversely affect a number of physiological parameters, such as gait (Birrell & Haslam, 2009; LaFiandra et al., 2003; Smith et al., 2006). Some research studies in western countries have revealed that plantar pressure is increased in obese adults whose body mass index (BMI) is $> 30 \text{kg/m}^2$ (Gravante et al., 2003; Hills et al., 2002; Birtane & Tuna, 2004). Around the world, it is estimated that two million people present with chronic plantar heel pain each year. For these people, the number one cause is plantar fasciitis (Thing et al., 2012). One of the concerns with the high prevalence of obesity is its association with an increased risk of falling. Each year, obese adults fall almost twice as frequently (27%) as their non-obese counterparts (15%) (Wu & Madigan, 2014). An important reason for falling in obese people is the loss of balance due to disrupted plantar sensitivity of the foot (Gravante et al., 2003).

Overweight and obesity influence plantar pressure distribution (Mickle et al., 2006; Phethean & Nestr, 2012; da Rochaet al., 2014). These effects, caused by obesity, are explained by the effect of weight on the foot structure. The literature includes studies reporting that excessive weight and increased BMI are related to a decrease in the arch of the foot and pes planus, changes in the plantar fat pad,

biomechanical changes in the feet, and increased plantar pressure while standing and walking (Butterworth et al., 2012; Song et al., 2015).

The foot of an adult who is overweight or obese may differ in structure and function compared with the foot of a healthy weight individual due to alterations in morphology, soft tissue properties and functional capability (Dowling et al., 2001; Hills et al., 2002; Riddiford-Harland et al., 2011). Specifically, lower longitudinal arch heights (Gilmour & Burns, 2001; Gravante et al., 2003; Mickle et al., 2006) and greater foot lengths and girths (Mickle & Steele, 2015; Mickle et al., 2006) are evident in the feet of adults and children who are obese compared with healthy groups. As some research does not differentiate between overweight and obese individuals, the morphology and function of the feet of adults who are overweight are yet to be widely and thoroughly investigated independently from those of obese adults (Mauch et al., 2008).

Obesity also results in an increase in the ground contact area of the feet, altering the plantar pressure distribution and inducing pressure peaks in certain parts of the foot (Lee, 2009, Dowling et al., 2001). Peak plantar pressure is the highest pressure registered on each part of the foot, and its evaluation provides a strong clinical tool for understanding the structural and functional consequences of obesity (Filippin et al., 2007). The most negative consequences of being overweight or obese are that individuals often exhibit greater posture deficits (Gilleard & Smith, 2007; Greve et al., 2007) and gait alterations (Ko et al., 2010; Segal et al., 2009; Lai et al., 2008). The following sections discuss different gait analysis components.

1.3. Overview of clinical gait analysis (gait characteristics)

This section discusses gait analysis components. Comprehensive gait analysis is generally used for the assessment of a patient with a movement disorder. There are three components to gait analysis:

The first component is the spatiotemporal data: Spatiotemporal data provide a basic quantitative description of time and distance parameters during walking that involve: stride length, step length, stride width, step width, cadence, stride time, step time, walking speed (velocity) and phases (stance/swing).

The second component is kinematic parameters: Kinematic parameters give information about body motion, position, moments, and acceleration of body segments (joint angles like trunk angle, hip angle, knee angle and ankle angle). Track or trace coordinates (X, Y, and Z) and orientation are also considered. These are important because they reveal the significant effect of obesity on mechanics during walking.

The third component is kinetic parameters: Consideration of kinetic parameters includes kinetic factors; the term to describe gait with the force acting on the body including evaluating joint loading by calculating ground reaction force (GRF), pressure patterns, joint powers, joint forces momentum and torque (Winter, 2005). Therefore, spatiotemporal, kinematic and kinetic parameters have been researched as the main parts of the human gait. A thorough understanding of obesity on gait parameters is becoming more important as its effect on health status is gaining increasing significance. The next section will discuss how obesity affects walking.

1.4. Alterations in gait due to overweight and obesity

Numerous researchers have investigated the impact of higher body mass and BMI on gait characteristics (spatiotemporal, kinematic and kinetic), generally.

1.4.1 Spatiotemporal

First, spatiotemporal studies have shown there to be biomechanical differences between adults of healthy weight and those who are obese:

- Changes to the temporal-spatial gait parameters include decreased or shorter stride length (Hills & Parker, 1991; Spyropoulos et al., 1991; de Souza et al., 2005; Russell et al., 2010; Browning & Kram, 2007; Ko et al., 2010; Runhaar et al., 2011; Lai et al., 2008; DeVita & Hortobágyi, 2003; Messier, 2010) which has been related to reduced plantar flexion after initial contact for weight acceptance (Spyropoulos et al., 1991; Browning & Kram, 2007);
- Increased body mass height contributes to shorter step lengths (de Souza et al., 2005; Da Silva-Hamu et al., 2013; Kirtley, 2006; DeVita & Hortobágyi, 2003);
- Greater stride width (Hills & Parker, 1991), most likely due to increased inner thigh adipose tissue;

- Greater step width (Spyropoulos et al., 1991; de Souza et al., 2005; Browning& Kram, 2007; Hills & Parker, 1991; Huang et al., 2013; Dufek et al., 2012; Ko et al., 2010; Sarkar et al., 2011). Which can mainly be attributed to excessive adipose tissue between the obese individual's inner thighs, altering the angular components of the gait. For instance, step width was twice as wide as for the non-obese participant (Browning & Kram, 2007; Hills & Parker, 1991; Sarkar et al., 2011; Spyropoulos et al., 1991);
- A lower cadence of walking (Hills & Parker, 1991; Gill, 2015; Da Silva-Hamu et al., 2013; Kirtley, 2006);
- Slower velocity of walking (Chow et al., 2005; Ehlen et al.,2011; Da Silva-Hamu, et al., 2013; Kirtley, 2006; Hills &Parker, 1991; Malatesta, et al., 2009; Freedman Silvernail, et al., 2013; Ko et al., 2010; Hergenroeder et al., 2011; Spyropoulos et al., 1991; de Souza et al., 2005; Segal et al., 2009; Wearing et al., 2006; Runhaar et al., 2011; Lai et al., 2008; DeVita & Hortobágyi, 2003; Messier, 2010; Browning et al., 2013; Browning et al., 2009; Browning & Kram, 2007; England & Granata, 2007; Silvernail et al., 2013; Russell et al., 2010);
- Longer stance phase duration (Hills & Parker, 1991; Dufek et al., 2012; Lai et al., 2008; Spyropoulos et al., 1991; DeVita & Hortobágyi, 2003; Amiri et al., 2015, Browning & Kram, 2007; Lai et al., 2008). Which is important to increase stability therefore, obese adults tend to spend more time in the stance phase, with reduced time spent in the swing phase duration (Martin & Nelson, 1986; Dufek et al., 2012; Spyropoulos et al., 1991; Błaszczyk et al., 2011; Browning & Kram, 2007; DeVita & Hortobágyi, 2003);
- Longer period of double support phase compared to the healthy group (Martin & Nelson, 1986; Spyropoulos et al., 1991; Chow et al., 2005; Błaszczyk et al., 2011; DeVita & Hortobágyi, 2003; Browning & Kram, 2007; Lai et al., 2008; LaRoche et al., 2011; Hills & Parker, 1991; Huang et al., 2013; Dufek et al., 2012).

All of these make obese individuals walk more slowly than healthy individuals. The next section investigates the kinematic differences found for healthy weight and obese participants.

1.4.2. Kinematic factors

Kinematic factors are important because they reveal the significant effect of obesity on walking mechanics. During walking, obese adults have a more extended knee during early stance along with a greater pelvic obliquity during late stance (Lerner et al., 2014):

- The range of motion (ROM) is limited at the knee, hip, and ankle during walking (Gill, 2015; Ling et al., 2009; Lyytinen et al., 2013; Park et al., 2010);
- During level walking, significantly greater peak tibiofemoral contact forces are experienced due to greater muscle forces (Lerner et al., 2014; Haight et al., 2014);
- Differences can be detected in the hip joint (Spyropoulos et al., 1991; DeVita & Hortobágyi, 2003; Lai et al., 2008; Lerner et al., 2014), the knee joint (Browning & Kram, 2007; Lai et al., 2008; Segal et al., 2009; Russell et al., 2010; Lerner et al., 2014), and the ankle joint (Spyropoulos et al., 1991; Messier et al., 1994; de Souza et al., 2005; Lai et al., 2008; Vismara et al., 2007);
- Differences in the sagittal and frontal plane lower extremity joint mechanics like knee joint moments (Browning & Kram, 2007) and greater frontal plane joint moments for both overweight or obese (Gushue et al., 2005; Shultz et al., 2009; Shultz et al., 2014);
- Reduced forward trunk flexion (Sibella et al., 2003; Galli et al., 2000; Gilleard et al., 2002; Gilleard et al., 2008), greater hip extension moments (DeVita & Hortobágyi, 2003; Browning& Kram, 2007; Lai et al., 2008; McMillan et al., 2010), increased hip abduction (Wills, 2004; Ko et al., 2010; Spyropoulos et al., 1991; Lai et al., 2008; McMillan et al., 2010; McMillan et al., 2009; Shultz et al., 2009), smaller knee torque, and lower peak knee flexion angles (DeVita & Hortobagyi, 2003; Foti et al., 2000; Gushue et al., 2005; Ko et al., 2010), and higher peak extended knee moments through initial stance (DeVita & Hortobágyi, 2003; Browning & Kram, 2007; Lerner et al., 2014; McMillan et al., 2010; Gushue et al., 2010; Gushue et al., 2014; McMillan et al., 2010; Gushue et al., 2003);
- During level walking, obese adults experience significantly higher peak tibiofemoral contact and greater muscle forces through the gait, thus causing gait alterations (Lerner et al., 2014; Haight et al., 2014; Zeng et al., 2017);

- Diminishing muscle strength with obesity (Cau et al., 2014; Ponta et al., 2014);
- Decreasing plantar flexion angles during stance and swing that is an adaptation to keep a similar speed to that of healthy adults (Spyropoulos et al., 1991; Lai et al., 2008; Browning & Kram, 2007; DeVita & Hortobágyi, 2003);
- Adopted or altered foot structure (Butterworth et al., 2014), so the obese walk with flatter feet (Hills & Parker, 1991), and they have a greater toe out/foot progression angle (Messier et al., 1994; de Souza et al., 2005);
- Reduced peak knee extensor moment scaled to body mass is substantially decreased in obese subjects than in normal-weight individuals (Spyropoulos et al., 1991; DeVita & Hortobágyi, 2003; Lai et al., 2008; Browning & Kram, 2007; Messier, 2010; Runhaar et al., 2011).

1.4.3. Kinetics

Finally, kinetic factors are important because they reveal the significant effect of obesity walking. During walking, obese adults have a greater pelvic obliquity during late stance (Lai et al., 2008; Lerner et al., 2014), as well as increasing torque at the knee joint (Gilleard et al., 2008; Sibella et al., 2003). Leading ligament looseness and muscle strength reduction in ankle joints have been related to excess body weight. Excess body weight is also usually distinguished through lower strength (Hulens et al., 2001).

Increasing power at the knee joint (Ko et al., 2010; DeVita & Hortobágyi, 2003) and greater peak knee extensor moments occur at the knee joint during gait (Silvernail et al., 2013; Ko et al., 2010; Lai et al., 2008; Segal et al., 2009; Russell et al., 2010; Vismara et al., 2007; DeVita & Hortobágyi, 2003). Peak vertical greater absolute GRF (de Castro et al., 2014; Browning & Kram, 2007) and increased vertical GRF have been linked to an increased risk of injury and joint degeneration (Butler et al., 2003). There is also a positive correlation between peak GRF and body mass (Messier, 2010). For example, as body mass increases (greater than 40 kg/m²) people have a smaller number of strides as a result of the reverse link amidst the mean number of steps and BMI. Furthermore, healthy weight subjects have higher peak intensity of physical activity with less exertion than obese subjects (King et al., 2008), reduced impact on proximal leg joints (such as knee joint and hip joint) (DeVita & Hortobágyi, 2003; Ko et al., 2010), reduction in ankle dorsiflexion, higher

peak plantar pressure (Butterworth et al., 2014) and greater muscle forces (Lerner et al., 2014; Haight et al., 2014).

Researchers have obtained a great deal of precise knowledge by investigating biomechanical parameters such as gait cycle, time phase, the center of pressure and pace during gait. While aspects of gait have been extensively studied in elderly patients with obesity, examining the centre of pressure would permit a more thorough evaluation of biomechanical parameters, providing more useful and accurate information on the anatomical structures acting in and on the foot during walking. People who are overweight and obese compensate for instability and muscle weakness by changing their walking style, and depend on different body movements to move their excess mass. This study explores the lack of information about functional gait imposed by a small change in body weight. Conversely, a reduction in obesity will lead to improved balance and a lower risk of falling (Aaboe et al., 2011; Song et al., 2015).

Overweight and obesity affect foot function and morphology, and the weight placed on the human foot can play a role in determining both the biomechanical behaviour and the function of the foot (Guiotto et al., 2013). The foot is one of the most complex structures of the human body, comprising multiple active and passive components which provide important support and functionality to the rest of the human body (Richie Jr, 2007). Feet, as the body's base of support, continually withstand GRF through the normal activities of daily living. Several researchers have proposed that the excessive increases in weight-bearing forces caused by obesity, might be damaging to the lower limbs of adults, like the feet (Hills et al., 2002; Wearing et al., 2006). Obesity and overweight have a detrimental impact on the structure and function of the foot, and numerous foot issues and injuries have been connected to defects in the foot (e.g., ulcers in the diabetic foot) that may be caused by excessive loading on a certain part of the plantar surface (Bus et al., 2005).

1.5. Plantar pressure measures

Individuals with foot pain have been shown to have altered plantar pressure measures compared to those without foot pain (Mickle et al., 2010). Plantar pressure measures provide a sophisticated way of assessing gait adaptations in those with foot and ankle

pathologies (Lobet et al., 2012; McKay et al., 2017). Plantar pressure measures can identify differences in loading patterns and the distribution of pressure acting on the foot during walking (McKay et al., 2017). Both the platform system and insole systems may be utilized to measure plantar pressures, providing high resolution options due to a large number of sensors, thereby providing accurate pressure maps of the foot during gait (Lobet et al., 2012). Plantar pressure analysis provides a reliable method of diagnosing foot and ankle pathologies, including osteoarthritis (Wafai et al., 2015). Additionally, plantar pressure measures have been shown to identify differences across patient groups, particularly for those who frequently fall or are at higher risk of falling (Gray et al., 2014; Moghadam et al., 2011). Not only do plantar pressure analyses provide a way of assessing differences in gait between individuals with ankle osteoarthritis and healthy controls, they also provide a way of assessing the effect of interventions.

The longitudinal arch in the human foot has muscles and ligaments that provide support for maintaining the arch, and these muscles and ligaments mimic the mechanism of an elastic band by storing the energy as they are stretched, and releasing energy as they return to their original state, thus assisting with the propulsion of the body during movement. Increases in body weight affect the ligaments and soft tissue in the arch which lead to a loss of elasticity the arch (Dowling et al., 2001).

A considerable number of research and clinical evaluations have used foot plantar loading assessments because plantar load distribution gives an insight into plantar loading attributes through functional activities like walking and running (Orlin & McPoil, 2000). In addition, studying foot plantar loading is important for the understanding of human foot injury mechanisms, foot biomechanics, foot stress, and the neuropathic foot (Perttunen, 2002; Castro et al., 2013). Plantar pressure assessment can determine loading features on particular plantar zones of the foot (Rosenbaum & Becker, 1997; Chong et al., 2013). In fact, foot plantar load distribution studies are of particular interest as they provide information about the GRF or plantar pressure acting on foot regions, and the physical interactions between the foot and the supporting surface (Orlin & McPoil, 2000). Furthermore, the human foot plays a vital role in the body's kinetic variables/parameters as it serves as a base of support (Mickle et al., 2006; Yan et al., 2013). Finally, plantar foot loading has been extensively observed in both walking (Vereecke et al., 2003; Giacomozzi & Martelli, 2006; Mao et al., 2006; Menz & Morris, 2006) and stationary conditions (Mao et al., 2006).

Scientific clarification on the correlation of increased body mass and the health of the plantar foot is yet to be made. Feet are the only part of the body that are directly connected to the ground during walking. The foot plays a significant role in alleviating the strike coming from the ground while protecting joints from mechanical energy, and it plays an important role in stabilizing and moving the body (Thomas et al., 2011).

Specifically, the longitudinal arch of the foot plays a key role in absorbing and distributing the high ground reaction forces due to its direct influence on such varied aspects as foot function, stability, pain and the predisposition to injury (McCrory et al., 1997; Dowling et al., 2001). Investigation of plantar load distribution allows the diagnosis and analysis of GRFs, or pressures, acting within the foot regions during standing and movement (Menz & Morris, 2006).

A study of human gait involves detailed examinations of various characteristics such as physical and pressure measurement which have been utilized for a better understanding of human gait strategies. In particular, expert analysis that uses various techniques to study human gait analysis and assess abnormal human gait is most frequently based on the pressure measuring system (sensory) or on video recordings (vision) (Prakash et al., 2018). Physical and pressure characteristics are assessed for a better understanding of human gait patterns. A number of techniques are available for analysing gait. These are based on pressure sensors that are placed on the subject's body, on the floor, or inside the insole of shoes (Alaqtash et al., 2011; Muro-de-la Herran et al., 2014). Pressure sensors are used to obtain the kinetics of the subject's movement and are being divided into two types: force platforms and pressure systems (Robertson et al., 2013). Both systems are used by clinicians and researchers and rely on quantifying the centre of pressure of various foot regions (Wearing et al., 2012; Cousins et al., 2013). They provide information about the foot; its structure and plantar loading characteristics when standing, as well as during gait (Mathieson et al., 1999; Tong & Kong, 2013). They may also be based on a video camera to capture human body movement by placing direct or indirect markers on the subject's body (Lee et al., 2014).

Several authors have investigated the functioning of feet using pressure sensors. They selected different parameters in their trials: different numbers of regions of the foot (from two to ten), age group, weight, and variables that are used to describe plantar pressure loading including peak pressure (PP), peak force (PF), pressure time (PT), pressure time integral (PTI), centre of pressure (COP) and centre of mass (COM) (Cousins et al., 2012; Cousins et al., 2013; Butterworth et al., 2015; Song et al., 2015). This research indicates the useful contribution of pressure sensors in providing insights into foot function and foot structure.

The most important technique used to study gait in this work is the in-shoe pressure F-Scan[©] (Tekscan) which can provide valuable clinical information about the management of individuals at risk of developing flat foot, foot ulcerations or Charcot foot by providing information about their footprint (Riddiford-Harland et al., 2011; Periyasamy et al., 2012). Therefore, in-shoe pressure F-Scan[©] (Tekscan) is the most important technique utilized to study gait in this work. The goal of this work is to provide a more accurate picture of one aspect of the diverse effects of simulated body weight, and to examine its connection to plantar pressure patterns in light of socio-demographic data.

1.6. Research gap

With the increased prevalence of overweight and obesity in adults, it is essential that further work be conducted to assess the full effect of various adult weight categories on the foot by evaluating plantar pressure patterns. The plantar pressure patterns will be analysed considering age and BMI, to provide a more comprehensive understanding of the connection between simulated body weight and the insole system. Therefore, this study focuses on investigating the plantar loading differences between groups of adults aged 24 to 50 years.

1.7. Research aim

The aim of this study is to investigate the effect of body weight on the feet and to investigate the use of simulated body mass to study the effect of variable body mass on the foot plantar in adults aged 24 to 50 years of age, while walking at a self-selected pace.

1.8. Research objectives

To achieve the overall aim of this research, the following objectives were defined as:

- Assessing dynamic foot plantar pressure characteristics in adults who are normal weight, overweight or obese during level walking using the insole pressure system, and comparing these values (the findings) with those of a healthy control group;
- Studying the impact of simulated body weight (SBW), used to increase the weight from healthy to overweight and obese, by comparing higher body weight which simulated body weight (SBW) (by wearing a vest with 10, 20, and 30 kg weight) group with the normal body weight (NBW, 0 kg);
- 3. Evaluating the spatial relationship between the trace of the centroid of the area of contact with a heel strike, midstance, and toe-off phases of simulated body weight (SBW) groups, using the in-shoe plantar pressure system.

1.9. Research questions

The research questions were:

- 1. Are there any noticeable changes in plantar pressure load between healthy, overweight and obese adults?
- 2. How do changes in weight (simulated body weight) influence plantar pressure patterns?
- 3. What is the most suitable plantar pressure data to provide an understanding of the relationship between simulated BMIs and gait characteristics such as force, contact area, contact pressure, plantar pressure integrals, and peak pressure measures?
- 4. What are the spatial correlations between the trace of the centroid of the area of contact of the plantar surface and the trace of the center of pressure captured by the pressure mat during gait respectively?

1.10. Significance of the research

Completing this research may bring a number of important benefits. These are:

- 1. Understanding how body weight changes pressure/force distribution and plantar surface during gait;
- Understanding how simulated body weight changes the dynamic data of foot pressure;
- 3. Evaluating the innovative techniques used at USQ in: (a) the determination of the change of x and y locations at various incidence angles of plantar pressure data; and (b) the detection impact of increasing body weight on plantar pressure measurements namely pressure time integral, peak pressure, pressure contact area and average pressure.
- 1.11. Thesis structure

A brief outline of each chapter is given below:

After the current chapter, a literature review is presented in Chapter 2, introducing the background which concentrates on the prevalence of overweight and obesity as well as the causes and consequences of overweight and obesity, and reviews human gait analysis. The anatomy of the human foot and the main sections of the foot connected to the problems, are also considered, as are plantar load and pressure measurement techniques. Chapter 3 focuses on the practical aspects of the three experiments and their corresponding methodologies to extract the body weight on gait. The results and discussion for the first experiment which relates to objective one are provided in Chapter 4, which proposes a definition of obese adults compared with overweight and healthy individuals. Chapter 5 presents the integrated discussion for the second experiment, which relates to objective two and provides an explanation for the simulated body weight, and the effect on plantar foot measures. Chapter 6 focuses on the results and discussion for the third experiment, which relates to third objective on enhancing the x-axis and y-axis with coordinate values that represent simulated body weight with to three significant stages of the load phase: 1) heel strike; 2) midstance; and 3) toe-off, for both feet. The conclusion, presented in **Chapter 7**, brings together all the chapters. It summarizes the main findings from the experiments together with suggestions for future work.
2. LITERATURE REVIEW

2.1. Introduction

The literature review is presented in the following order: Section one reviews overweight and obesity, Section two considers human gait, Section three discusses the human foot, Section four considers plantar pressure and Section five explores pressure measurement. The literature review then concludes with a brief discussion and summary.

2.2. Overweight and obesity

During the past ten years, the prevalence of overweight and obesity has been increasing rapidly worldwide. Moreover, it has become more common, not only in adults, but also in adolescents and children (Zimmet et al., 2014). Overweight and obese adults have become significant public health concerns worldwide (Sheffler et al., 2014). In 2016, the World Health Organization (WHO) indicated that, from a global perspective, nearly two billion (aged 18 years and older) of the population are considered overweight and, of these, over 650 million are clinically obese. Thirty-nine percent of people aged eighteen years and over have been found to be overweight; thirteen percent were obese. More disturbingly, obesity is on the rise. The worldwide prevalence of overweight or obese males has increased from 28.8 to 36.9% between 1980 and 2013, whilst that of females rose from 29.8% to 38.2% during the same period. This means the rate of obesity has quadrupled in adults and doubled in children in the past 30 years (Ogden et al., 2014).

The prevalence of obesity has increased dramatically over the past three decades in virtually every country. **Figure 2.1** illustrates the global epidemiology of obesity in adult males. According to the Centre for Disease Control and Prevention, in the United States alone, there are more than 78.6 million obese adults (CDC, 2014). Currently, in the United States, approximately 33.9% and 38% of American adults are overweight and obese, respectively (Fryar et al., 2014). Alarmingly, American obesity rates have nearly doubled for adults and tripled in children since

the 1980s (Obesity Rates & Trends Overview, 2015). In Australia, data from a recent Australian Institute of Health and Welfare report shows the problem has escalated to the point where nearly two-thirds (63%) of adults are overweight and/or obese (of both sexes) with 28% categorised as obese, and more than one in four children are also considered to be obese (Abo, 2015).

Being obese presents higher health risks than being overweight (AIHW, 2016; Australian Government, 2017), but both present potential health problems. The obese population has a higher risk of suffering ulcers because of extra pressure in the vascular system (Firat, 2017). Obesity and diabetes are associated with each other in that, and obesity plays a significant role in the cause of diabetes (Leong & Wilding, 1999). Obese people have extra adipose tissue, and the existence of adipose tissue increases the pressure and pain sensitivity threshold. Some studies determined that high plantar pressure was related to obesity and that plantar pressure was related to plantar sensitivity (Wu & Madigan, 2014). Plantar pressure is frequently measured to solve foot problems in patients with pain, diabetes, or rheumatism (Lee et al., 1996). The increased contact area was not found to be sufficient to compensate for the high forces generated during walking, which cause higher plantar pressures with increased stress (Mickle et al., 2006). Therefore, obese individuals feel less pain when a force, such as pricking with a needle on their forearm, is applied (Scheinfeld, 2004). Feeling less pain and sensitivity causes pressure ulcers. It can be seen also in elderly people for the same reason, and in diabetics whose disease is a consequence of obesity.



Figure 2.1. Prevalence of obesity worldwide in males 15 to 100 years old in 2010. This figure represents the prevalence of obesity (BMI > 30 kg/m^2) among males on a global scale in the year 2010 (Haidar & Cosman, 2011)

The significant physiological and biomechanical complications related to overweight and obesity have generated substantial economic costs because both conditions are serious, complex, and may be lifelong. One aspect of these cost is that the diseases associated with overweight and obesity place a major burden on the economy through an increased use of the health care system and treatment costs. Other aspects will be mentioned below. In 2014, the global economic impact from obesity was approximately \$2.0 trillion, or 2.8% of the global gross domestic product, roughly equivalent to the global impact of smoking or armed violence, war, or terrorism (Dobbs et al., 2014). In the United States, approximately \$147 billion per year is used up on obesity-connected health care costs (CDC, 2011). In Australia, the total cost of obesity to society and governments has been estimated at \$58 billion (in 2011-2012), which includes a direct cost of \$8.6 billion for productivity costs due to short and long-term employment impact \$3.6 billion, health system costs of \$2 billion and carer costs of \$1.9 billion. In addition, lost wellbeing disability and the early death consequence of obesity and its effects cost the country \$49.9 billion (AIHW, 2017). Therefore, overweight and obesity were the second highest contributors to the burden of disease in Australia (Sainsbury & Zhang, 2012).

Prevalence of foot pain and pathology can be disabling, leading to more complex orthopaedic complaints overtime. Foot pain contributes to 20% of hospital orthopaedic referrals and surgical procedures (Chuckpaiwong et al., 2009), with a larger proportion (24-41%) of the population suffering from foot pathology (Riskowski et al., 2011). Pathologies of the foot have been associated with many different causes including obesity, foot mechanics, foot posture, diabetes and reduced range of motion within the foot (Chuckpaiwong et al., 2009). Pain due to ankle osteoarthritis tends to be aggravated by weight-bearing activities, such as walking, climbing stairs and jogging (Valderrabano et al., 2007). Hill et al. (2008) examined 3,206 participants and found that obese cases had an increased prevalence of foot pain. Individuals with foot pain have been shown to have altered plantar pressure measures compared to those without foot pain (Mickle et al., 2010).

Overweight and obese people are experiencing more foot pathologies (such as plantar fasciitis, plantar heel pain) due to increased body weight than they did in the past (Riddle et al., 2003; Irving et al., 2007; Prichasuk, 1994). Some research studies have found that there is a clear association between foot pain and elevated plantar pressure for overweight and obese adults during gait. It has been demonstrated that load carriage can adversely affect a number of physiological parameters such as gait (Birrell & Haslam, 2009, 2010; LaFiandra et al., 2003; Smith et al., 2006). So far, some research studies in western countries have revealed that plantar pressure is increased in obese adults (Gravante et al., 2003; Hills et al., 2002; Birtane & Tuna, 2004). It is estimated that two million people around the world present with chronic plantar heel pain each year, of which the number one cause is plantar fasciitis (Thing et al., 2012). Also, the prevalence of arthritis in American normal weight adults over forty years of age is between 25.7 and 33%, while in obese adults the prevalence is between 37.0 and 44.4% (Ong et al., 2013). In addition, one of the most common presentations of foot pain in clinical practice is pain in the plantar heel. Pain related to the pathology of the plantar fascia is the most commonly diagnosed pain. Preventive examinations prove that insufficient quality or quantity of increased weight may initiate functional or structural modifications of the foot that may cause pain for the patient. Foot pain occurs relatively often. Approximately 3.6% South Australian suffer from heel pain because of increased body weight, according to the North West Adelaide Health Study, which shows a

similar prevalence to data for the United States (7.3%) and the United Kingdom (9.8%) (Garrow et al., 2004; Riddle & Schappert, 2004; Hill et al., 2008; Dufour et al., 2009).

2.2.1. Definition and classification for overweight and obese adults

Overweight is defined as a state of weighing more than healthy body weight, thus being at risk of developing obesity. This means that the weight exceeds the threshold of criterion standard or reference value BMI greater than 24.5 kg/m² (Fallon et al., 2005). Obesity is defined as having too much or abnormal or excessive fat accumulation of body fat adipose tissue and can be associated with serious medical complications that impair quality of life. Therefore, overweight and obesity are not synonymous terms (Kopelman, 2000; Fallon et al., 2005). Both terms thus imply that a person's weight is higher than the healthy level for his or her height. A human can be overweight, but not obese through having extra muscle or bone (Fallon et al., 2005). Both conditions are caused by the energy imbalance between calories consumed and calories expended (Sheehan & Gormley, 2013). Healthcare experts have commonly utilised the BMI in the adult population to classify and define the body state as underweight, healthy, overweight and obese (Rossouw et al., 2012). BMI (kg/m^2) is calculated by dividing the ratio of body weight (in kilograms) by the square of the height value (length) in metres of an individual (Jegede et al., 2017; Vijayakumar et al., 2016) as described in Equation 3.1. Chapter 3, Section 3.3.3.5.

According to this system, people with a BMI of less than 18.5 kg/m^2 are considered by the WHO to be underweight. A healthy or normal weight range is $18.5 \text{ to } 24.9 \text{ kg/m}^2$ while 25.0 to 29.9 kg/m² is considered to be overweight or pre-obese. Over 29.9 kg/m² is obese. The obese classification can be further split into three categories, Class I is from 30 to 35 kg/m², Class II is from 35 to 40 kg/m², Class III or morbidly obese is over 40 kg/m² (WHO, 2016). BMI is an indicator for assessing overweight and obesity in a population since it can be applied to all categories of people. BMI is a useful and simple index which has a clear connection to the amount of fat mass (WHO, 2016). In general, a person with the BMI greater than or equal to 30 is classified as obese (WHO, 2016).



Figure 2.2. Classification of BMIs human body requirements for each group (from CDC, 2014)

2.2.2. Causes of overweight and obesity

Reasons for increasing incidences from illnesses relating to overweight and obesity are not singular but include a wide range of influences. For instance, unhealthy food, quantity of food consumed; meaning an imbalance between energy intake and energy expenditure, insufficient physical activity, gender, environmental factors, illness, medication, psychological factors, and genetics play a role in this accumulation of excess body fat (Wearing et al., 2006; McAllister et al., 2009; WHO, 2011; Hageman et al., 2013; Zimmet et al., 2014).

2.2.3. Consequences of obesity and overweight on health

Both overweight and obesity have been recognised as crucial health problems (outcomes). Several health complications are associated with overweight and obesity. For example, it has been identified as the leading cause of Type II diabetes mellitus. Also, people who are obese are more likely to suffer from coronary heart disease, dyslipidemia (for example, high total cholesterol or high levels of triglycerides), hypertension (high blood pressure) and cancers (endometrial, breast, and colon).

Furthermore, people with obesity have higher rates of osteoarthritis (a degeneration of cartilage and its underlying bone within a joint for hip, knee, and ankle), poor physical functioning, and functional gait abnormalities (Birtane & Tuna, 2004; Kushner & Roth, 2003; Lyytinen et al., 2014; Russell & Hamill, 2011; Lee & Kean, 2012; WHO, 2016). In addition, obesity has a sturdy connection with lower extremity disorders, like osteoarthritis and pain in both ankle and foot (Frey & Zamora, 2007, Ozdemir et al., 2005). For instance, a recent systematic review investigating twenty-five studies involving 93,224 participants in the general population, have noted that there is a strong relation to non-specific foot pain and chronic plantar heel pain and obesity (Butterworth et al., 2012). Obesity modifies body geometry. It increases the mass of the different segments, and imposes functional limitations, primarily affecting the lower limbs (Kortt & Baldry, 2002), which can significantly influence the biomechanics of activities during daily living predisposing obese individuals to injury (Wearing et al., 2006).

An increasing number of deaths can be attributed to these conditions (Perju-Dumbrava et al., 2017). To summarise, various health conditions connected to excessive weight gain, relate to both overweight and obesity, such as Type II diabetes mellitus and poor physical functioning. Similarly, it is reported that obesity and an increase in weight in children and adults were related to musculoskeletal pain, especially involving the lower extremities (knee, feet, and hip), which leads to disability (Stovitz et al., 2008).

As mentioned above, obesity is associated with musculoskeletal problems, higher rates of hyperuricemia, and immobilisation due to pain in the lower back and knees, gout, and in the feet (Birtane & Tuna, 2004; Kushner & Roth, 2003; Lai et al., 2008; Lyytinen et al., 2014).

In addition, overweight or obesity causes several morphology physiological changes, including the adaptation and impairment of the musculoskeletal structure and function in adults because of excess body weight (Bankoff et al., 2003). Such as abnormalities of the foot associated with a different way of walking; and the abnormal gait that may eventually lead to falls (Pan et al., 2016). In addition, excess weight negatively affects the joints of the lower limbs like hips, knees, ankles and feet, which can cause their misalignment due to increased stress on joints (Bankoff et al., 2016).

al., 2003; CDC, 201; de Sá Pinto, 2006), and metatarsalgia related to pain and inflammation in the ball of the foot (Chang et al., 2012), limitations in muscle strength, decreased mobility, and altered positioning of the feet, all of which affects posture (Horak, 2006; Lyytinen et al., 2013).

Obesity also results in an increase in the ground contact area of the feet, altering the plantar pressure distribution and inducing pressure peaks in certain parts of the foot (Lee, 2009, Dowling et al., 2001). Peak plantar pressure is the highest pressure registered on each part of the foot, and its evaluation provides a strong clinical tool for understanding the structural and functional consequences of obesity (Filippin et al., 2007). The most negative consequences of being overweight or obese are that individuals often exhibit greater posture deficits (Gilleard & Smith, 2007; Greve et al., 2007) and gait alterations (Ko et al., 2010; Segal et al., 2009; Lai et al., 2008).

There is also likely to be an effect on balance control (Greve et al., 2007; Singh et al., 2009) and as a result, people suffering from obesity also have an increased risk of falling (Corbeil et al., 2001) as well as an increased likelihood of foot ulceration (Vela et al., 1998) and heel pain (Prichasuk, 1994). It has also been reported that impairments caused by overweight and obesity may include atrophied leg muscles, longitudinal arch and plantar fat pad, development of claw toes and hammer toes and limited joint mobility (Butterworth et al., 2012).

Numerous researchers have established that obesity affects walking patterns in negative ways because of the impact of carrying additional fat mass. This impact, together with reduced relative muscle strength, has been associated with a progressive worsening of health and increasing discomfort (Lai et al., 2008; Sheehan & Gormley, 2013). Therefore, to undertake human gait analysis, three major groups of gait parameters should be considered. The most common types of parameters researched in gait analysis are spatiotemporal, kinematic, and kinetic parameters (Winter, 2005). From these parameters, several types of knowledge about any person can be extracted (Collins et al., 2002). The following sections discuss different gait analysis components.

An increasing body of evidence has shown obesity to have a negative effect on the development, treatment, and outcome of lower extremity pathologic entities, including diabetic foot disease (Pirozzi et al., 2014). But, obesity, in particular, has been associated with numerous serious and chronic health conditions, which may have negative physiological and mechanical implications on the body. For example, Type II diabetes mellitus, coronary heart disease, metabolic syndrome, dyslipidemia (for example, high total cholesterol or high levels of triglycerides), hypertension (high blood pressure), inflammation, cancers (kidney, endometrial, colorectal and pancreatic prostate, breast and ovarian, and colon), stroke, sleep apnoea and cardiovascular diseases (Reeves et al., 2007; McClean et al., 2008; Guh et al., 2009; Bell et al., 2014; CDC, 2015; WHO, 2016; AIHW, 2017). Obesity also presents a greater risk for bone and joint problems, such as osteoarthritis (a degeneration of cartilage and its underlying bone within a joint for hip, knee, and ankle). Due to obesity, excessive and cumulative joint loads have been suggested as posing a high risk for the onset and progression of osteoarthritis (CDC, 2011; Jiang et al., 2012; Mezhov et al., 2014). Obesity has been reported as having a direct link to neuromusculoskeletal deficiencies (Dario et al., 2015; Singh et al., 2015), poor physical functioning, and functional gait abnormalities (Russell & Hamill, 2011; Lee & Kean, 2012; WHO, 2016). Overweight and obesity conditions can be attributed to increasing number of deaths (Perju-Dumbrava et al., 2017).

Limitations in muscle strength are associated with limited mobility, which can adversely affect the quality of life (Stenholm et al., 2009; Vincent et al., 2010) and decreased mobility. The risk of falls is twice as common among obese individuals compared to healthy weight individuals (27-15%) (Hahn et al., 2018). Postural instability goes hand in hand with an increased risk of falling and related injuries (HitaContreras et al., 2013) and altered positioning of the feet (Horak, 2006; Lyytinen et al., 2013). Overweight and obese individuals with foot pain show a higher level of heel-pad thickness and less heel-pad elasticity which may, lead to a decreased level of shock absorption and cushion (Prichasuk et al., 1994). The time to achieve peak pressure and peak pressures in the metatarsal heads were higher pressure rate or time to pressure is used to estimate the risk of injury in the plantar portion of the foot (Yan et al., 2013). Higher pressure indicates a greater impact on the foot, and a higher risk of injury (Yan et al., 2013).

The most negative consequences of being overweight or obese are that individuals often exhibit greater posture deficits (Gilleard & Smith, 2007; Greve et al., 2007) and gait alterations (Ko et al., 2010; Segal et al., 2009; Lai et al., 2008). There is likely to be an effect on balance control (Greve et al., 2007; Singh et al., 2009) and as a result, an increased risk of falling (Corbeil et al., 2001) as well as an increase in the likelihood of foot ulceration (Vela et al., 1998) and heel pain (Prichasuk, 1994). It is reported that the impairments often include atrophied leg muscles, longitudinal arch and plantar fat pad, development of claw toes and hammer toes and limited joint mobility (Butterworth et al., 2012). These structural and functional abnormalities of the foot are associated with a changed walk strategy, eventually leading to falls (Pan et al., 2016). Overweight or obese individuals often exhibit greater impairments in posture and gait and have an increased risk of falling.

Several researchers have established that obesity affects gait patterns in adverse ways because of the impact of carrying additional fat mass. This impact, together with reduced relative muscle strength, has been associated with a progressive worsening of health and increasing discomfort. One of the important causes of impaired physical function in obese individuals is believed to be reduced muscle strength (Stenholm et al., 2009) and lowered skeletal muscle mass (Zoico et al., 2004, Bosy-Westphal et al., 2015). Stenholm et al. (2009) reported that those older obese individuals with decreased lower limb muscle strength had a higher risk for the development of physical function disability compared with those of a healthy weight. Zoico et al. (2004) demonstrated that a low percentage of muscle mass significantly increased the probability of experiencing functional limitations. Sibella et al. (2003) determined that, in completing a sit-to-stand task, obese subjects adopted a different movement strategy from non-obese subjects. The obese subjects' forward trunk flexion was reduced, and their feet moved backwards from the initial position. Sibella et al. (2003) suggested that the reduction in trunk flexion represented an attempt to diminish loading of their lower back. The next section describes some fundamental gait concepts and introduces human gait.

2.3. Human foot anatomy

Foot and ankle movement are critical for human locomotion. They attenuate impact forces, provide support and stability for the body by interacting with the supporting surface and assist forward progression (Stefanyshyn & Nigg, 1997; Whittle, 1999). To enable walking, each of these functions relies on a complex series of coordinated joint motions (Lundgren et al., 2008). The foot is the link between the supporting surface and the lower extremity during walking. The foot, as the most distal part of the lower limb chain, is in contact with the ground and resists the forces that pass through it (Ledoux & Hillstrom, 2002). Abnormal pressure distribution on the plantar surface of the foot may cause pain and act as a source of pain (Becker et al., 1997). Such pain can be more prominent as a result of obesity where the centre of pressure constantly changes because of the persons increased body weight (Periyasamy et al., 2012; Yan et al., 2013). This illustrates that understanding foot biomechanics helps clinicians to more effectively evaluate normal and abnormal conditions (Kirby, 2000).

The foot is a major part of the skeleton that bears significant loads during locomotion (Periyasamy et al., 2012). In addition, it is considered one of our most distinctive morphological and functional features. The foot is one of the most complex musculoskeletal structures of the human body (Ganea et al., 2017). It consists of twenty-six bones and thirty-three joints (twenty of them active joints), layered with an intertwining web of more than hundred muscles, ligaments, cartilages, and tendons. It contains a network of blood vessels, nerves, skin, and soft tissue, such as the plantar fascia or plantar fat pad as shown in (**Figure 2.3**).



Figure 2.3. The foot connects with the lower leg at the talus, tibia, and fibula interface (Lochner, 2013)

In fact, the only part in contact with the ground through standing and movement is the foot. Generally various features of the foot structure is accompanied via changing plantar pressure distribution and situations of the lower extremity bones, muscles, tendons and ligaments, which in turn, influence lower limb biomechanical features and foot function (Cavanagh et al., 1997). Understanding both structure and function characteristics of the foot might assist recognition of the aetiology of a number of lower extremity defects. Excess body weight causes detrimental alterations of the foot structure and function, which is a potential exegesis to the association of obesity with ankle and foot defects (Butterworth et al., 2015).

Flexible foot structures work in unison to serve a variety of biomechanical functions. These structures allow the human foot to support weight, absorb the shock of impact force, serve as levers to propel the leg forward, and are responsible for maintaining body balance in statics and dynamics through symmetrical distribution of plantar pressure (Hicks, 1954; Hamilton, 1967; Elftman, 1969; Netter & Colacino, 1989; Putz & Pabst, 2006; Abdul Razak et al., 2012; Ganea et al., 2017).

Structurally, the main sections of the human foot can be divided into three main sections: the rearfoot, midfoot, and forefoot (**Figure 2.4**). This presents the way the human foot is connected to the body via the articulating surfaces provided by the tibia and fibula (Lochner, 2013).



Figure 2.4. View of the dorsal bones of the rearfoot, midfoot, and forefoot (adapted from (Oatis, 2009)

The foot plantar fat pad is organised into two layers of adipose and connective tissue that provides cushioning to the underlying foot structures (Hills et al., 2002). The function of the plantar fat pad is to absorb shock loading, particularly on the forefoot and heel regions (Ozdemir et al., 2004; Natali et al., 2010). The plantar fat pad is one of the main risk factors of developing foot pathologies like diabetic foot ulceration in both adults and children (Patry et al., 2013; Yan et al., 2013). Common in people suffering simultaneously with obesity and diabetes, is a foot ulcer; a state of skin erosion characterised by the inability to self-repair (Levin & O'Neal, 1988).

2.3.1. Foot type

Foot and ankle disorders have been reported in several studies, which show that they are positively associated with obesity (Butterworth et al., 2013; Butterworth et al., 2012). Moreover, research has shown that obese adults have a greater prevalence of foot and ankle disorders compared to both normal-weight adults and overweight adults (Krul et al., 2009). Explaining the potential exegesis for the connection of obesity to foot and ankle defects, is that increasing body weight causes mischievous changes to the foot structure and function (Butterworth et al., 2015). In addition, the foot structure and arch play key roles in optimal personal wellbeing (Fukano & Fukubayashi, 2009).

Because the medial longitudinal arch is the primary shock absorbing structure of the foot, this area of the foot is particularly important for foot function (Oatis, 2004). Moreover, the medial longitudinal arch not only absorbs, but also dissipates high ground reaction forces or pressures that are generated during activities of daily living. Although this arch comprises bony articulations as well as ligaments and muscles, it is primarily the ligaments that support and stabilise the longitudinal arch (Jahss, 1982; Ker et al., 1987). They also act as powerful stores of mechanical energy, thereby decreasing peak joint loads and increasing efficiency (Ker et al., 1987). Additionally, they prevent compression of the plantar regions of the foot that house muscles, nerves, and blood vessels (Jahss, 1982; Abboud, 2002). It has been reported that the impairments caused by overweight and obesity often include atrophied leg muscles, longitudinal arch and plantar fat pad, the development of claw toes and hammer toes, and limited joint movement (Butterworth et al., 2012). These

structural and functional abnormalities of the foot are associated with altered ways of walking (Pan et al., 2016). However, over time, excessive body weight is characterized by increased vertical forces which may lead to reduced muscle strength and may stretch ligaments beyond their elastic limit, damaging soft tissues and increasing the risk of foot discomfort and subsequent development of foot pathologies (Dowling et al., 2001).

As mentioned above, dissipating plantar pressure distribution is also affected by foot type. Typically, the foot structure is described and classified according to foot type. There are three various categories based on the height of the foot arch (medial longitudinal). These are high foot arch (pes cavus), normal foot arch and low foot arch or flatfoot (pes planus) as presented in (**Figure 2.5**). Arch type is typically discussed specifically in relation to the medial longitudinal arch.



Figure 2.5. Three different types of human foot arch and the contact area of each type (Kraushaar, 2005)

Load distribution on the foot is also affected by foot type. For example, previous literature has shown that the applied pressure underneath the plantar perfunctory of the flatfoot increases through foot movement, especially within the rearfoot, medial midfoot and lateral midfoot as compared with the normal foot (Queen et al., 2009). Contact area (CA) and peak pressure (PP) in midfoot noticeably decreases in individuals who have a high foot arch (cavus foot) (Queen et al., 2009).

Chuckpaiwong et al. (2008) studied the correlation between foot shape and the risk of plantar injuries and found that during foot movement, the stress and the PP concentrated on the fifth metatarsal of individuals with normal and high foot arch, whereas PP increased in the second and third metatarsal of the flat foot.

Plantar pressure measurements can be affected by both foot type and foot deformities. Using an Emed-SF® system, Burns et al. (2005) studied thirty subjects with idiopathic cavus foot; ten with neurogenic cavus foot and thirty with normal feet. The pressure time integrals in the rear foot and forefoot in both cavus foot groups were higher than in the normal group. Other studies have shown that subjects with hallux valgus, toe deformities, diabetes with neuropathy, leprosy and rheumatoid arthritis can display altered plantar pressure in the forefoot (Bus et al., 2005; Slim et al., 2012; Mickle et al., 2011; Gurney et al., 2008).

Accordingly, subjects with other forefoot complaints such as bunions, hammertoes, plantar plate pathology, systemic arthritis and diabetes have not been included in the current research project. Data were collected on both feet for ten healthy participants using the Emed®. Six measurements were demonstrated to be reliable with intra-class correlation coefficients (ICC) greater than 0.8 (Zammit et al., 2010).

2.4. Human gait analysis

The gait can be known as walking; a procedure of movement that uses both legs alternately and provides together propulsion and support (Whittle, 2007). In general, gait analysis assistance is used to study various kinds of movement patterns as well as forces involved in producing those movements and permits measurement and interpretation of kinematics and kinetics (Switaj & Connor, 2008). Human gait analysis is the examination of a person's style of gait, which is a term applied to the movement of human limbs through locomotion. As one researcher explains, human gait analysis is the methodical study of human locomotion (Chakravarthy et al., 2017). Gait is integral to the way of human movement and includes walking, running and crawling (Nutt et al., 1993). Based on the review of section 2.3 on the human

foot, human gait can provide a wide variety of information that can be utilised in the study of various aspects of health such as biometrics, disease diagnosis, load determination and injury rehabilitation (Boulgouris et al., 2005; Piecha, 2008; BenAbdelkader & Davis, 2002; Hailey & Tomie, 2000).

Examination of the human walk has been used in diverse areas, such as physiotherapy, the identification of people (Muro-De-La-Herran et al., 2014) and security (Nair & Kendricks, 2016). Physiotherapy analyses gait limitations caused by sports activities and musculoskeletal and neurological diseases, such as Parkinson's disease (Hannink et al., 2017). Physiotherapists analyse gait patterns to recognise normal and pathological gait movements (Vieira et al., 2017). Moreover, gait analysis, which is linked to human walking, has been undertaken by researchers. Gait analysis methods are used for observing negative deviations in the gait features, as well as determining their reason and influences (Johnson & Bobick 2001; Prakash et al., 2015). Moreover, gait analysis has been established as the clinical technique for determining gait disabilities and dynamic position and coordination during movement (Kavanagh & Menz, 2008). It can be a useful tool as it provides quantitative information about personal gait to help in the provision of optimal treatments (Kadaba et al., 1989; Simon, 2004). Gait analysis assists therapists, physicians, biomechanics and other experts as well as researchers. All information is used to assist the patient once they have undergone measurement of joint kinematics and kinetics, such as lower limb rotations, knee movement and foot placement (Agrawal et al., 2007; Yakimovich et al., 2006; Pollo et al., 2002). Gait analysis is an important tool in defining the biomechanical factors that may affect the progression of certain pathologic conditions, like shaking, rigidity, etc. and describes the internal and external forces acting on the stability of human movement (Duarte & Freitas, 2010; Chang et al., 2004; Miyazaki et al., 2002; Lynn et al., 2007). The goal of gait analysis is to identify gait phases, which may be seen in each normal gait cycle of human walking patterns and can thus be analysed (Miyazaki et al., 2002; Lynn et al., 2007). Gait analysis has been the most studied topic for assessing the locomotor characteristics of humans.

2.4.1. Human gait cycle and phases

The gait cycle consists of two periods of double support and two periods of single support. The double support phase is the period during which both feet are on the ground. The stance and swing phases or single support phase, typically make up the main phase sequence of this repetitious cycle by two limbs. The gait cycle starts with the first contact of one foot and ends at the next contact of the same foot. Each normal gait cycle of human walking patterns consists of two main phases, the stance phase and the swing phase (Miyazaki et al., 2002; Lynn et al., 2007). The term 'human gait' denotes locomotion or manner of moving on human feet (Perry & Davids, 1992). It involves two major phases, stance and swing (Figure 2.6) (Perttunen, 2002). The stance phase represents the foot-surface interaction with the supporting surface and can be divided into three stages: 1) an initial double limb stance/heel strike; 2) single limb stance/midstance, 3) terminal double limb stance/toe-off (Perry & Davids, 1992; Perttunen, 2002).

The double limb stance period occupies approximately 10% of the complete gait cycle. The single limb stance occurs from 10% to 30% of the cycle and represents the clinically important phase of the gait cycle. The terminal double limb stance or toe-off occurs from 40% to 60% of the period of the gait cycle (Cuccurullo, 2004).

The stance phase includes two periods of double support and one single support. The stance phase accounts for approximately 60% of a single gait cycle. The first phase is the stance, which begins at initial foot contact and refers to the entire time the foot is on the ground. This means that the stance phase starts at heel touchdown and ends at toe-off for the same foot, as shown in (**Figure 2.6**). The stance phase includes five sub-phases: initial contact, loading response, mid-stance, terminal stance, and pre-swing (Ayyappa, 1997). Pre-swing has been called double support and occupies approximately 10% of the cycle. Right after toe-off, there is a period of single support when just one foot is on the ground. The single gait cycle accounts for approximately 40% (Winter, 2009). It denotes the time that the foot is in the air advancing forward (Miyazaki et al., 2002; Lynn et al., 2007).



Figure 2.6. The normal gait cycle of an 8-year old boy (Perttunen, 2002)

Perttunen (2002) reported the period of the stance phase takes approximately 60% of the gait cycle, while the swing phase takes around 40% of the gait cycle. The motion of the foot, from heel contact through midstance to toe-off (propulsive phases), is of particular interest to clinicians because it provides information on the internal loading of the foot and the physical interaction between the foot and the surface. Thus, spatiotemporal, kinematic and kinetic parameters have been generally researched as the main parts of the human gait.

2.4.2. Gait measures

To study the human gait, some major groups of gait parameters should be considered. The most common types of parameters researched in gait analysis can be subdivided into spatiotemporal, physical dimensions of humans (anthropometric), kinematic and kinetic variables. These parameters are defined below. Spatiotemporal data provide a basic description of time and distance parameters during which movement takes place, namely stride and step length, step width, cadence, velocity, and phases (stance and swing) (Miyazaki et al., 2002; Lynn et al., 2007). Kinematic parameters include the acceleration of body segments, joint angles, angular motion as well as tracking coordinates (X, Y, and Z) and orientation. Kinetic refers to the study of force (F), such as ground reaction force (GRF) that causes human movement.

There is a useful set of information regarding GRF. For instance, plantar pressure measurements are associated with GRF (Muro-de-la Herran et al., 2014). Plantar pressure provides information about the function of the ankle and foot during gait. The most important clinically relevant variables are peak pressure (PP), pressure time integral (PTI), contact area (CA), contact time (CT) and centre of pressure (COP). Moreover, PP, PTI, CA, CT are plantar pressure measures commonly assessed during gait (Putti et al., 2008; Keijsers et al., 2010). PP is the single most generally reported measure and represents the maximum load in the plantar surface of the foot or given the area of the foot during gait (Bus & Waaijman, 2013; Keijsers et al., 2010; Melai et al., 2011). This is clinically important because the magnitude (the bulk) of PP can be localised to the plantar surface when walking (Melai et al., 2011). The PTI is defined as the area under the pressure-time curve within each foot mask and is expressed in kilopascals multiplied by seconds (Rao et al., 2011; Redmond et al., 2008). The PTI describes the total effect of pressure over time in a given area of the foot, providing a value for the total load exposure of the sole of the foot area with each step (Melai et al., 2011). Those with forefoot pain have been found to have significantly higher levels of PTI compared to healthy controls (Keijsers et al., 2010). The CT is also a clinically relevant measure as it indicates the time in milliseconds. Through the stance phase, each area of the foot is in contact with the pressure pad (Putti et al., 2008). Knowing contact time can aid in understanding loading patterns can identify compensations or fear-avoidance associated with loading the painful area (Rao et al., 2012).

All gait analysis information can be achieved using several types of sensors. For example, kinetic analysis involves the study of plantar pressure with pressure mats or insoles devices and GRFs. Moments and pressures are recorded through force plates. Kinematic analysis studies motion captured with a three-dimensional (3D) system, joint angles with goniometers, evaluating acceleration with SMAs and actions of muscles with electromyography (EMG) (Burnfield, 2010; Whittle, 2007; Mathie et al., 2004). These are presented in (**Figure 2.7**).



Figure 2.7. Systems in normal gait laboratory, utilizing photocells for synchronizing both camera system and force platforms. Further, accelerometers, electromyography, and pressure insoles can be used (Lyytinen et al., 2014)

2.5. Plantar pressure

All bodies load the ground due to the gravity force. Human bodies load the ground in the sitting, standing, lying and walking status. The counteracting force (F), ground reaction force (GRF), is transmitted through human bodies and is felt like pressure. Forces acting over small areas produce high pressure, however, a larger area (with equal force) reduces the pressure, as presented in the following equation for pressure:

P=F/A(2.1)

Where: P (pa) is pressure, F (N) is the force and A (m^2) is an area

A high PP damages skin, other soft tissues, and underlying structures. Plantar pressure is mainly measured in gait laboratories and clinical practices. The two most common techniques utilised are barefoot measurements and in-shoe pressure measurements. The design of barefoot measurement platforms varies - some are

large and can register several steps (**Figure 2.8**), while others such as walkway platforms, measure single steps.



Figure 2.8. The plantar pressure platform registers several steps while the patient walks barefoot a- foot plantar pressure system by Zebris Medical GmbH. b- plantar pressure sensor emed® by Novel (cited by Razak et al., 2012)

Plantar foot ulcers typically develop at points of the foot where there is high pressure. Sites of increased pressure typically occurs over bony prominent areas during walking (Ledoux et al., 2013). A number of studies (e.g. Waaijman et al., 2012) have reported that the most common locations of plantar foot ulcers are in toes and metatarsal heads (Waaijman et al., 2012). It is reported the most common position of foot sores occur at the first metaphalangeal joint region (27%), followed by the hallux position (18%) (Waaijman et al., 2012). Furthermore, it is clear that there is a correlation between the site of the ulcer and the magnitude of pressure. Consequently, it is important to gain a thorough understanding of the varied factors that may influence plantar pressure during walking (Hessert et al., 2005).

Levin and O'Neal (1988) explain that the foot skin acts as a mechanical protective layer which shields the foot from external stresses. Plantar soft tissue is constructed of three layers: the epidermis (the outer layer), dermis (second layer) and subcutaneous tissues (the layer between the skin and muscle), as shown in (**Figure 2.9**). In addition, these tissues comprise a number of layers of skin, fat, fascia and muscle, which work together to provide cushioning during walking (Chao et al., 2011). The process of increased pressure and foot ulceration originates from alterations to deeper layers of soft tissue (Chao et al., 2011). Murray et al. (1996)

point out that dry skin, in combination with high plantar pressures, and the repetitive nature of walking leads to the formation of a callus (toughened area of skin) under prominent bony areas, including the metatarsal heads. The process of keratinisation is stimulated by overactivity due to the repetitive compression applied during walking. This result in hypertrophy of the stratum corneum, is thought to increase the proliferation of epidermal cells (Murray et al., 1996). A callus increases the risk of developing an ulcer. This is due to the increased hardness and density of the skin which increases pressure during walking (Reiber et al., 1998). This research indicates that changes to tissue morphology can influence biomechanical properties, which have the potential to affect cushioning.



Figure 2.9. Layers of the plantar skin

2.5.1. Plantar load distribution: an overview

In both clinical practice and research studies, plantar load distribution has become a significant tool for assessing human gait (Orlin & McPoil, 2000; Monteiro et al., 2010). As mentioned in the prior section, plantar loading is a term used to refer to the

vertical ground reaction forces exerted on the plantar surface of the foot. The study of plantar load distribution is important for the understanding of human foot biomechanics, foot stress, foot injury mechanisms, and foot neuropathy diseases, such as diabetes (Maetzler et al., 2010; Perttunen, 2002; Castro et al., 2013; Mohd Said et al., 2016).

In fact, plantar load distribution studies are of particular interest as they provide direct and detailed information on the ground reaction forces or pressure acting on foot regions, and the physical interactions between the foot structures and the supporting surface (ground interaction) (Alexander et al., 1990; Orlin & McPoil, 2000; Titianova et al., 2004). Normally, plantar load distribution investigations emphasise the determination and analysis of GRF or pressures acting on the foot regions during static (standing) or dynamic (walking) conditions (Vereecke et al., 2003; Giacomozzi & Martelli, 2006; Mao et al., 2006; Menz & Morris, 2006). Usually, the interaction between the sole of the foot and the supporting surface is presented either by GRF graphs or by a pressure map, which depicts the values of localised and mean pressures (Hayafune et al., 1999; Perttunen, 2002).

Several conditions influence plantar load distribution as load distribution characteristics differ from person to person due to a range of factors, like variations in body weight (Menz et al., 2006; Martínez-Nova et al., 2008; Rodrigo et al., 2013; Yümin et al., 2016), walking speed (Taylor et al., 2004; Menz et al., 2006), gender (male and female) (Bennett & Duplock, 1993), foot shape (Chuckpaiwong et al., 2008), age (Morag & Cavanagh, 1999; Scott et al., 2007; Menz et al., 2006; Martínez-Nova et al., 2008; Yümin et al., 2016), foot geometry (Hayafune et al., 1999), race differences (Putti et al., 2010) and stride length (Menz et al., 2006; Martínez-Nova et al., 2008). Conversely, it has been hypothesised that body weight, pacing velocity and the structural variations in the foot arch have significant influences in changing foot load distribution (Taylor et al., 2004; Menz et al., 2006; Chuckpaiwong et al., 2008; Chang et al., 2012). Hence, a component of the current investigation focuses on identifying the influence of body weight, age and self-selected speed foot movement (gait) on changing plantar load distribution.

Plantar loading has been extensively observed in static (Mao et al., 2006) and dynamic situations (Vereecke et al., 2003; Giacomozzi & Martelli, 2006; Mao et al., 2006; Menz & Morris, 2006; Patry et al., 2013). It has been reported that each foot

carries about half (50%) the body weight during normal stance (Bennett & Duplock, 1993; Antunes et al., 2008; Cheng et al., 2008), with the highest plantar pressure distributed under the heel (H), forefoot and big toe (1stT), and the lowest plantar load located under the midfoot (MF) (Perttunen, 2002).

However, the plantar pressure distribution map alters noticeably during foot movement (Taylor et al., 2004; Chuckpaiwong et al., 2008). Previous literature has indicated that an increase in the speed of foot movement leads to an increase in PP in the H, medial and central metatarsals and 1stT region and excludes the medial foot and lesser toes (Perttunen, 2002; Mao et al., 2006; Morag & Cavanagh, 1999; Menz & Morris, 2006). Hennig et al. (1994) indicated that PP transfers sequentially from the H region to the metatarsal heads between the early and late stance. Hayafune et al. (1999) reported that during the late stance phase, the load acting on the first metatarsal head (1MH) and second metatarsal head (2MH) and the 1stT region is about 64% of the total forefoot load. Perttunen (2002) reviewed and summarised the values and the locations of PP during walking, relying on previous research (**Figure 2.10**). This data clearly shows that during foot movement, the PP is transformed rapidly from the rearfoot to the forefoot. The regions under the H, medial and central forefoot and 1stT, exhibit the highest PP value.



Figure 2.10. Mean and SD for the peak plantar pressure during walking (Perttunen, 2002)

2.5.2. Foot plantar pressure mask or anatomical regions

Foot plantar anatomical regions measure the foot plantar pressure, with various subareas of the sole (Walker & Fan, 1998; Bryant et al., 2000; MacWilliams et al., 2003). Analysing pressure data begins with the first stage, which is to divide the plantar area into different sub-areas of interest on the sole, to obtain clinically applicable data from a large number of sensors. The common, as well as a necessary method of investigating plantar pressure, is to 'mask', to define the regions of high risk (Cavanagh et al., 1992). A mask simply defines the borders of the foot into anatomical regions of interest to provide more clinically relevant information than observation of the foot as a whole (Razak et al., 2012). Specific plantar pressure variables can be computed for each person, region or mask. Such auto masks have been applied to analyse the plantar pressure or normal walking parameters (Bryant et al., 1999; Bryant et al., 2000) and various foot deformities (Bus et al., 2005, Orendurff et al., 2006). Analysis of load or pressure distribution applied to the plantar surface of the foot is undertaken by dividing the plantar surface into a number of regions. The number of regions within the foot varies among researchers from two to ten regions. The selection of the number, name and areas of foot regions may differ between researchers based on the purpose of the study. The numbers and names of the foot regions are displayed in detail below, according to the previous studies. For example,

a) Two regions: A foot is divided into two areas: the forefoot and hindfoot (Dowling et al., 2001).

b) Three regions: The foot is divided into three sections: rearfoot, midfoot, and forefoot (Oatis, 2009), otherwise known as forefoot, midfoot, and hindfoot (Bedi and Love 1998).

c) Four regions: For the purposes of analysis, there are a number of ways of categorising the foot: Vela et al. (1998) evaluated pressures located under the H, MF, 1MH, 2MH through 5MH regions. Pirozzi et al., (2014) considered areas under the H, MF, forefoot, and 1MH). Birtane and Tune (2004) included different anatomical foot locations (forefoot, rearfoot, total plantar, and forefoot plantar).

d) Five regions: The foot regions were divided into five regions, described in two different ways: H, MF, forefoot, 1stT and lesser toes (Walsh et al., 2017); or five anatomical foot areas called H, MF, MHs, 1stT and the lateral four toes (Elnaggar, 2016).

e) Six regions: Studies have divided the foot sole into six anatomical regions. Cousins et al. (2013), for example, divided the regions into six, namely lateral H, medial H, MF, 1MH, 2MH to 5MH, and 1stT. Mueller et al. (2016) considered six-foot regions: total foot, toes, forefoot, medial MF, lateral MF and hindfoot. Nyska et al. (1997) showed that the foot can be divided into six regions H, 1stT, 2nd to 5th toes, metatarsal region, medial MF, and lateral MF. Arnold et al. (2010) divided the foot into six regions, namely (M1 Hallux, M2 Lesser Digits, M3 First Metatarsal, M4 Second to Fifth Metatarsals, M5 MF, and M6 H). The masks mentioned above were developed to produce six anatomical regions; whole foot, H, MF, forefoot, 1stT and toes (Butterworth et al., 2015). And, described another way, six anatomical regions: H, MF, 1MH, 2-5MHs, 1stT, and lesser toes (Bus & de Lange, 2005).

f) Seven anatomical foot locations, called the posterior and anterior H, 1MH, 2MH, 4MH, 5MH, and 1stT (Zhu et al., 1991). According to Femery et al. (2003) seven areas were identified under the lateral H, medial H, the fifth (M5), the third (M3) and the first (M1) metatarsal heads and the 1stT. Speksnijder et al. (2005) called the areas H, MF, lateral forefoot, central forefoot, medial forefoot, 1stT and 2ndT to5thT.

g) Eight regions: Hills et al. (2001) evaluated eight anatomical foot locations (H, MF, I–VMHs, 1stT).

h) Nine areas under the foot: H, medial MF and lateral MF, 1stT, and each of the five metatarsals (Phethean & Nester, 2012). Maetzler et al. (2010) also divided the foot into nine regions: H, MF, 1MH to 5MHs, 1stT and 2ndT to5th T.

i) Ten regions, medial and lateral heel (HM and HL), metatarsal areas (M1–M5), MF, 1stT and 2nd-5thT were described by Monteiro et al., (2010). In addition, Mickle and Steele (2015) described ten anatomical landmarks: H, MF, 1MH, 2MH, 3MH, 4MH, 5MH,1stT, 2ndT and 3rd-5thT. The foot was divided into ten regions: H, MF, 1MH to 5MH, 1stT, 2ndT and 3rd-5thT by Putti et al. (2008); Putti et al. (2010). These regions are displayed in (**Figure 2.11**), and the symbols (a, b, c, d, e, f, g, h, and i) were given to indicate the number of regions of the foot.



Figure 2.11. Illustration of the variety of foot regions

This figure shows the foot divided into three regions, namely the forefoot, midfoot and hind foot. These zones can be reproduced by software provided by the various in-shoe or pressure platform systems. The number of regions varies among researchers from two to ten, foot regions (Putti et al., 2008; Putti et al., 2010; Birtane & Tuna, 2004; Chevalier et al., 2010; Bennett & Duplock, 1993; Hills et al., 2001; Ho et al., 2010; Aydos et al., 2012). However, most researchers who have investigated the plantar loading of children and adults agree that the plantar is divided into ten anatomical regions (**Figure 2.12**).



Figure 2.12. Ten anatomical regions of the foot defined for plantar pressure distribution analysis. M01: Heel, M02: Medial midfoot, M03: 1 MT Head, M04: 2 MT Head, M05: 3 MT Head, M06: 4 MT Head, M07: 5 MT Head, M08: Hallux, M09: M10: Toes (Putti et al., 2008)

2.5.3. Plantar pressure distribution

Plantar pressure assessment is commonly used in the clinical evaluation of the foot and provides insight into plantar loading characteristics during functional activities, such as walking and running. Currently, in-shoe pressure systems provide two techniques for presenting either through means of pressure or forces measured reaction of the sole to the ground. Many researchers have analysed plantar pressure for healthy participants (Bryant et al., 2000; Putti et al., 2008; Putti et al., 2010; Putti et al., 2010; Maetzler et al., 2010) and others have analysed overweight and obese populations (Vela et al., 1998; Putti et al., 2008; Hills et al., 2001; Dowling et al., 2001; Birtane & Tune, 2004; Fabris et al., 2006). These have been used in various studies to detect foot pathologies in different age groups (adults and children),

whether healthy or overweight or obese. These studies have identified pressure characteristics during gait, the contact area of the plantar aspect of the foot and the forces produced. There are a range of devices available to measure these characteristics during barefoot and in-shoe pressure measurements.

2.5.3.1. Normal plantar pressure measurements

In a previous study by Bryant et al. (2000) with respect to PP distribution on ten masked areas, the maximum mean values were found under the H, 2MH and 3MH, and 1stT regions. The least variation of plantar pressure (mean and peak pressures) was found under the H and the 2MH and 3MH regions. The greatest variations were found under the toes and MF. This may be due to variability in arch heights, as other studies have confirmed this variation (Goffar et al., 2013). Kanatli et al. (2008) studied plantar pressure distribution under the forefoot of 106 normal participants via an Emed-SF® method.

They divided the foot into the medial column (1MH), middle column (2MH and 3MH) and lateral column (4MH and 5MH). They found that during the midstance and the push-off phase of gait, the mean and PPs were the highest under the 2MH and 3MH regions. Their study questioned the existence of the transverse metatarsal arch during gait. The absence of the transverse metatarsal arch was supported in the previous study where they showed that the transverse arch collapses during the push-off phase (Kanatli et al., 2003). This questions the tripod theory proposed by Kapandji (1970) which states that the longitudinal and transverse arches of the foot result in weight-bearing like a tripod between H, 1MH to 5MHs. Other investigators have questioned the presence of the transverse arch using ultrasonography on hundred healthy subjects and measured plantar pressures of the forefoot using an Emed-SF® system (Daentzer et al., 1997). Using ultrasound measurements in the weight bearing position, they found 2MH to 4MHs to be in a more plantar position and the maximum plantar pressure measured by an Emed-SF® system to be present in the 3MH region. With respect to PTIs, Bryant et al. (2000) reported that the highest measurements were under the 2MH and 3MH and the 1stT, and the greatest variation was noted beneath the 5MH and toe.

Putti et al. (2008) completed an evaluation of foot pressure values in the healthy individual. The authors recruited 53 normal weight individuals (17 women and 36 men) aged between 19 and 52 years. An Emed ST4 system was utilised to gather the foot pressure data. The authors divided the foot plantar into ten regions. They selected six of the most clinically relevant parameters which were PP, CA, CT, PTI, FTI and IPP. They established that the highest areas of PP were found under the 2MH and 3MHs, with the mean equal to 361 kPa and 330 kPa, respectively, followed by the 1stT with 321 kPa and H with 313 kPa. CA was the highest under the H strike at 35 cm². The percentage of CT was in the range of 75-85% under the MHs, and 70% under the 1stT. Additionally, PTI was the highest under the 1MH to 3MH and 1stT at 87 kPa/s, 107 kPa/s, 104 kPa/s, and 87 kPa/s, respectively. FTI value was the greatest beneath the H region at 105 N/s. The ranges of parameters can be applied in orthopedic clinics as part of the assessment of pathological conditions.

Putti et al. (2010) investigated in-shoe pressure differences between two different racial groups of 33 participants comprised of 12 Caucasians aged between 19–52 years and 21 persons from India age between 21-46 years. The authors divided the foot into 10 regions. They were selective in their investigation and recorded eight of the most clinically significant parameters, which were IPP, maximum F (max F), PP, CA, CT, PTI, FTI, and mean F. They concluded that the Caucasian group was higher than the Indian group within two pattern parameters, those being PP and PTI in particular, the PP under the H was 293 kPa versus 251 kPa; (P < 0.001), under 1MH was 294 kPa versus 233 kPa; (P = 0.01), under 2MH was 266 kPa vs. 236 kPa; (P = 0.03), under 3MH was 254 kPa vs. 223 kPa; (P = 0.04), and 5MH was 168 kPa versus 133 kPa; (P = 0.04) and the PTI under 1MH was 79 kPa/s versus 62 kPa/s; (P = 0.03) and under 5MH was 58 kPa/s versus 44 kPa/s; (P = 0.03). There were no significant differences in the CA, CT, FTI, IPP, max F and mean F variables between the two groups.

Putti et al. (2010) investigated any potential foot pressure differences between males and females using an in-shoe measurement for ten different regions of the foot for all the important pressure measurement characteristics. Twenty-eight subjects (16 females aged between 19-52 years and 12 males aged between 21-52 years) were recruited. In their examination, the researchers selected eight of the clinical parameters, (CA, CT, PP, PTI, FTI, IPP, mean F and max F) which were recorded

and analysed. The authors observed that the contact area in males was significantly larger in all regions of the foot compared with females (P < 0.001). They found a higher value within the H region (44.1 cm² for male and 34.1 cm² for females). Furthermore, FTI was also significantly higher in males under the 1MH, 3MH, and 4 MH (58, 35, 26 N/s), while in females these were 41, 21, 13 N/s respectively. Also, the max F was also significantly higher in males under the H area compared to females (701 N and 525 N), 1MH (216 N and 168 N) and 3MH (115 N and 77 N). The mean F was greater in males under the 1MH, 3MH, and 4MH. The authors found that there were no significant differences between genders in CT, PP, PTI and IPP in the foot.

Maetzler et al. (2010) measured pressure value in the normal foot with the Emed1 ST2 system (barefoot). Twenty-three subjects (fourteen females and nine males) with mean age of 36.0 ± 11.6 years participated. The authors divided the foot into 10 regions to record seven-foot pressure variables: beginning of contact, end of contact, CT, PP, IPP, CA and PTI. The highest areas of PP were found under the 1stT and 2MH, with mean equal to 435 kPa and 407 kPa, respectively, followed by the 3MH of 345 kPa and the hindfoot of 332 kPa. The CT (range of pressure, %) was 74–85% under the MHs, and 71% under the 1stT. CA was the highest beneath the H region at 33.8 cm².

2.5.3.2. Overweight and obese plantar pressure

One of the first examples of investigating plantar foot pressure is presented in Vela et al. (1998), who used an in-shoe pressure measurement system to compare the effect of increased body mass on peak plantar foot pressure. In this test, the authors employed 19 adult participants (10 male and 9 female) under modified body weight. The authors used three weights to simulated body weight (0, 9.1 and 18.2 kg). They divided the foot sole into four regions: H, MF, 1MH, and two to five MHs. The authors found a significant increase in mean PP under the MHs, H, and MF for each incremental increase in weight (baseline vs.9.1 vs.18.2 kg, p < 0.05). The authors deduced that increases in body weight increased plantar foot pressure for the H, MF, 1MH and lesser metatarsal zones in both genders. The most common change in the plantar pressure came as a result of the change in body weight (mass). It was

demonstrated that there was increasing plantar foot pressure in the H, MF, and forefoot with increasing weight, and they concluded that increased weight could increase the risk of lower extremity ulceration in the setting of other pathophysiologic features (Vela et al., 1998).

Dowling et al. (2001) examined foot plantar pressures during static and dynamic conditions in 13 obese and 13 non-obese children (mean age 8.1 ± 1.2 years). Each dynamic pressure footprint was divided into two areas, the forefoot and hindfoot. The obese group demonstrated greater PF under both forefoot (341.0 ± 93.6 and 227.5 ± 33.9 N for obese and non-obese subjects respectively) and hindfoot (446.6 ± 83.9 and 311.0 ± 55.0 N for obese and non-obese subjects respectively) areas but this was distributed over a larger surface area (total foot area 97.1 ± 11.9 versus 74.3 ± 9.2 cm²). Therefore, hindfoot PP (force/area) was not significantly different between the obese and non-obese groups but forefoot PP was (39.3 ± 15.7 N/cm² versus 32.3 ± 9.2 N/cm²). The authors reported that increased forefoot plantar pressures in obese children might lead to discomfort and hinder participation in physical activity. A limitation of this study was the division of the foot in half to describe the forefoot and hindfoot eliminating important findings that may have been found under the MF region.

More than a decade ago, Hills et al. (2001) used a pressure platform device to collect the plantar pressure data to investigate plantar pressure differences between obese and non-obese adults during standing and walking. In the test obesity was defined as a BMI greater than 30 kg/m². The study was conducted on 35 males (age 42.4 ± 10.8 years; 67-179 kg) and 35 females (age 40.0 ± 12.6 years; 46-150 kg). The authors evaluated the pressure from eight anatomical foot locations: H, MF, first to fifth MHs and 1stT regions. They observed that obese subjects, both female and male showed an increased H, MF and two to four MHs plantar pressures. Also, foot width showed an increase at the forefoot during standing and walking, and higher plantar pressures of the MHs and longitudinal arch areas (Hills et al., 2001). They concluded that comparing the healthy group to the obese group showed increased forefoot width and higher plantar pressures during standing and walking. The greatest effect of body weight on higher PP in the obese was found under the longitudinal arch of the foot and under the metatarsal heads. The higher pressures for obese women compared to obese men during static weight bearing (standing), may be the result of reduced

strength of the ligaments of the foot. It was estimated that the ground reaction forces experienced by the lower limbs of a healthy weight individual could be as large as three to six times their own body weight. They discovered that obese adults produce greater plantar loading than non-obese adults, particularly in the H, MF and forefoot regions. In addition, obese adults had greater CA of the MF region compared to non-obese adults.

Birtane and Tune (2004) aimed to compare the plantar pressure distribution in obese and healthy adults during standing and walking using a mini-Emad pedobarograph device. Fifty participants; twenty twenty-five healthy aged mean \pm SD (48.0 \pm 12.2) years and twenty- five persons obese Class 1 aged mean \pm SD (53.0 ± 9.5) years. The authors evaluated both the static and dynamic states. They assessed eight parameters during static measurement, which were forefoot PP value (N/cm^2) , rearfoot PP value (N/cm²), total plantar F (N), forefoot plantar F percentage, rearfoot plantar F percentage, total CA (cm²), forefoot plantar CA percentage, and rearfoot plantar CA percentage. In addition, seven parameters were measured during the dynamic measurement. These were peak phalanx pressure (N/cm²), medial forefoot peak pressure (N/cm²), middle forefoot PP (N/cm²), lateral forefoot PP (N/cm²), middle foot PP (N/cm²), rearfoot PP (N/cm²), and plantar CA. Birtane and Tuna (2004) reported that, out of six plantar regions, only the MF area recorded a statistically significant increase in peak plantar pressure when obese subjects (BMI between 30.0 - 34.99 kg/m²) were compared to non-obese controls. For the feet of Class 1 obese subjects in the static condition, they found that there were significantly higher values in terms of forefoot PP, total plantar force and total contact area, while only middle foot PP was found to be higher in Class 1 obese subjects than controls, as a dynamic pedobarographic parameter. Moreover, Birtane and Tune (2004) compared the plantar pressure distribution variables that measured obese and control adults during standing and walking, based on the body mass index. They found the rvalue of the variables that were measured were between weak to moderate, which may be required consideration. They emphasised that the facts might be the result of mechanisms of adaptation to weight bearing vertical force acting in the plantar arch due to obesity (Birtane & Tuna, 2004).

2.6. Pressure measurement techniques

In research or clinical work, there is a diversity in approaches utilised to measure vertical components of the GRFs or pressure that is applied to the human foot plantar surface during static or dynamic situations (Abdul-Razak et al., 2012; Symiya et al., 1998; Perttunen, 2002; Vereecke et al., 2003; Bryant et al., 2005; Noce, 2005; Giacomozzi & Martelli, 2006; Kalamdani, 2006; de Souza, 2007; Rana, 2009; Abdul-Razak et al., 2012; Rodrigo et al., 2013; Muro-de-la-Herran et al., 2014). Researchers have described three approaches for measuring plantar pressure (Perttunen, 2002; Orlin & McPoil, 2000). Firstly, the assessment of plantar pressure or GRF between the plantar surface of the bare foot and ground support. Secondly, determining pressure or GRF interaction between the sole of the shoe and the ground support and thirdly, measuring pressure between the plantar surface of the foot and the insole of the shoe. The current investigation emphasises the third assessment approach because it can be adopted to study a cargo distribution on the foot plantar zones of the in-shoe, which is the central and major clinical purposes of this study.

2.6.1. Pressure and ground reaction forces measuring systems

Throughout the development of pressure and GRF measuring systems, valuable and specific information has been obtained about the ground reaction forces and the pressure load distribution, which results from the interaction between the static or moving foot and the supporting surface (Perttunen, 2002; Vereecke et al., 2003; Noce, 2005; Giacomozzi & Martelli, 2006; Kalamdani, 2006; de Souza, 2007; Rana, 2009; Abdul-Razak et al., 2012; Muro-de-la-Herran et al., 2014). Figure 2.13 (a, b) shows that the main difference between these techniques is the variety of devices or systems employed to record and capture the human foot load (pressure) during standing or moving conditions. Generally, the plantar load measuring systems have been classified into two main classes, a pressure measuring systems and a GRF measuring system (Shi, 2001; Perttunen, 2002; Bryant et al., 2005; Noce, 2005; Abdul-Razak et al., 2012). Currently, a number of measuring system products are commercially available, (Figure 2.13 (b)) such as a foot insoles, pressure pads, (Vereecke et al., 2003; Kalamdani, 2006; de Souza, 2007); Pedar (Abdul-Razak et al., 2003; Kalamdani, 2006; de Souza, 2007); Pedar (Abdul-Razak et al., 2003; Kalamdani, 2006; de Souza, 2007); Pedar (Abdul-Razak et al., 2003; Kalamdani, 2006; de Souza, 2007); Pedar (Abdul-Razak et al., 2003; Kalamdani, 2006; de Souza, 2007); Pedar (Abdul-Razak et al., 2003; Kalamdani, 2006; de Souza, 2007); Pedar (Abdul-Razak et al., 2003; Kalamdani, 2006; de Souza, 2007); Pedar (Abdul-Razak et al., 2003; Kalamdani, 2006; de Souza, 2007); Pedar (Abdul-Razak et al., 2003; Kalamdani, 2006; de Souza, 2007); Pedar (Abdul-Razak et al., 2003; Kalamdani, 2006; de Souza, 2007); Pedar (Abdul-Razak et al., 2003; Kalamdani, 2006; de Souza, 2007); Pedar (Abdul-Razak et al., 2003; Kalamdani, 2006; de Souza, 2007); Pedar (Abdul-Razak et al., 2003; Kalamdani, 2006; de Souza, 2007); Pedar (Abdul-Razak et al., 2003; Kalamdani, 2006; de Souza, 2007); Pedar (Abdul-Razak et al., 2003; Kalamdani, 2006; de Souza, 2007);

al., 2012; Muro-de-la-Herran et al., 2014; Noce, 2005); F-Scan (Tekscan), and pressure platform systems (Noce, 2005; Rana, 2009). In addition, a force plate (e.g., AMTI) and wearable force sensors are commonly utilised to measure a GRF of the moving foot (Perttunen, 2002; Giacomozzi & Martelli, 2006; Liu et.al., 2009; 2010; Rupérez et al., 2009; 2011) for instance, the flexi-tactile sensor system Nitta Corporation, (**Figure 2.13 (a**)). Technically, in this study each foot load measuring system used to measure plantar load or plantar pressure has several advantages and a number of limitations (Noce, 2005; Abdul-Razak et al., 2012).



Figure 2.13. (a) Ground reaction forces measuring system; (b) Pressure measuring systems (Chong et al., 2015; Al-Magsoosi & Chong, 2019)

2.6.2. Foot pressure measurement systems

Foot pressure measurement methods include electronic systems intended to assess the foot pressure that measures the consequence from the interaction within the foot and the support surface (Shi, 2001; Perttunen, 2002; Noce, 2005; Bryant et al., 2005; Martínez-Nova et al., 2008; Flórez & Velásquez, 2010). These systems can be utilised to determine the vertical forces (FZ) applied by the human foot by integrating individual pressure distributions (Shi, 2001). Generally, foot pressure measurement systems consist of several tiny pressure sensors, thus, these systems are perfect for assessing plantar pressure without disturbing the subject or patient's movement during testing because they are thin enough, which allows non-intrusive assessment, as seen in (**Figure 2.14**) (Noce, 2005; Bryant et al., 2005; Abdul-Razak

et al., 2012; Rodrigo et al., 2013). Moreover, pressure systems are useful devices in clinical walk test studies for (Abdul-Razak et al., 2012; Bryant et al., 2005; Rana, 2009) foot detection and rehabilitation (Abdul-Razak et al., 2012), footwear design, biomechanics (Rana, 2009) and sport enforcement (Orlin & McPoil, 2000). A number of scientific researchers have noticed that pressure systems can provide sufficient outcomes on testing human gait compared to the more precise information of the force plate (Symiya et al., 1998; Shi, 2001; Noce, 2005). Thus, they have become an indispensable tool for gait analysis in both clinical and research settings. The sensing technology is capable of measuring static and dynamic pressure, which has played a crucial role in understanding the effects of pressure on the foot (Wafai et al., 2015).



Figure 2.14. The matrix of force resistance sensors (Chong et al., 2015)

The lack of precision (ranging between \pm 5% to \pm 25%), uncertainty and differences in sensors readings, partially through the dynamic assessments, are the main limitations in methods that determine pressure (Symiya et al., 1998; Orlin & McPoil, 2000; Shi, 2001; Abdul-Razak et al., 2012). Moreover, one of the most important limitations associated with these systems is the reliance on force profile and that they do not relate to the force distribution with the movement of the foot (Orlin & McPoil, 2000). Finally, pressure measuring systems cannot determine the fore-aft or medial-lateral shear forces, considered necessary for studying the development of plantar ulcers in a diabetic's feet (Orlin & McPoil, 2000).

Numerous technologies are employed to manufacture pressure measurement systems and they can be grouped into two kinds, in-shoe and platform (Abdul-Razak et al., 2012). The F-Scan - Tekscan® (Boston, US) insole plantar sensor pad system, also referred to as the in-shoe plantar pressure assessment, is becoming increasingly used, and is popular worldwide as both a research and clinical tool for pressure
measurement and force data collection during dynamic movements. For instance, evaluating foot pressure in humans (Waaijman & Bus, 2012; Chong et al., 2017). It allows the collection of objectives, and quantifiable measures to study different types of human lower limb pathologies, and it ensures a high degree of portability in different clinical settings. It has been used for many applications, including the clinical walk test (Baan et al., 2012) and diagnosis of foot-related diseases, such as diabetic ulceration, plantar fasciitis, and arthritis (Badlissi et al., 2005). It is also a unique candidate as a pressure sensor due to its miniature size, light weight, and very good sensitivity (Baan et al., 2012). The advantages and drawbacks of these two system types are addressed in the following sections.

2.6.3. In-shoe pressure measurements

Recording and monitoring the local pressure values between the human plantar surface and footwear are explored by in-shoe pressure systems. This is becoming a standard in today's health assessment because of its ability to objectively characterise behaviour inside as well as outside the laboratory environment (Orlin & McPoil, 2000; Abdul-Razak et al., 2012; Muro-de-la-Herran et al., 2014). It has been well documented that plantar pressure is strongly associated with ulceration (Boulton, 2014). Therefore, in-shoe measurement is a valuable tool that can be used to provide evidence which may ultimately impact both shoe design and clinical practice (Abdul-Razak et al., 2012; Orlin & McPoil, 2000). The systems are fundamental in many aspects like medical and sports applications including human gait analysis (Abdul-Razak et al., 2012; Muro-de-la-Herran et al., 2014), orthotic design (Abdul-Razak et al., 2012; Muro-de-la-Herran et al., 2014), shoe research, and sports biomechanics (Orlin & McPoil, 2000). The in-shoe system consists of a pressure film put into the shoe, which is linked to a storage system by USB cable. Generally, a pressure film is a tiny sheet that gathers a large amount of data from such as a small pressure sensor (Figure 2.15). Sensor distribution in the pressure film is high resolution in the zones of the H area and forefoot area and tends to have a low resolution in the MF area. An in-shoe pressure system can be manufactured simply, while some studies have inserted various prototype models of in-shoe methods (Noce, 2005; Bamberg et al., 2008; Rana, 2009). These are shown in (Figure 2.15). Applying in-shoe pressure methods for assessing the plantar pressure of the foot comprises several of advantages and some constraints. One of the most important features of the in-shoe

system is the ability to measure and analyse sequential steps, as the foot typically remains aligned with the same sensors. In addition, in-shoe systems are appropriate for enrolment pressure and interaction amidst foot and shoe. Also, these techniques are movable and flexible and can be utilised to collect pressure data without any location restrictions (in-door or outdoor). Furthermore, in-shoe pressure devices are typically less expensive than pressure platform systems (Urry, 1999; Orlin & McPoil, 2000; Abdul-Razak et al., 2012). However, in-shoe systems encompass a limited number of pressure sensors with low resolution. In-shoe systems tend to provide low precision product compared to mat methods for the sensor in the in-shoe film in line up with a non-flat surface inner footwear (Orlin & McPoil, 2000).

Caution should be taken when interpreting the results of pressure measurements because these results are dependent on the systems and techniques that have been utilised (Giacomozzi et al., 2000). Moreover, accurate calibration and equilibration of the sensors and appropriate handling of the equipment are all required to obtain valid, reliable measurements (Luo et al., 1998). In-shoe pressure measurement determines the pressure acting on the sole of the foot while the patient uses his/her shoes. A thin sensor is adjusted to fit into the shoe. In-shoe measurements allow the patient to move over a larger area. If a wireless system is used, the patient can move outdoors while the pressure is registered. However, the sensors in the in-shoe system are prone to mechanical failure as the connecting cables connecting the sensors to each other can be bent or stretched through foot locomotion (Orlin & McPoil, 2000; Muro-de-la-Herran et al., 2014). Likewise, sensor attributes and behaviour can be impacted and spoiled due to humidity and hot environment within the footwear (Abdul-Razak et al., 2012; Muro-de-la-Herran et al., 2014).

Pressure insoles seem to provide the better trade-off for gait analysis. The inshoe measurement system is also capable of capturing multiple footsteps' data from both feet at the same time and of recording foot function patterns directly inside the shoe. During in-shoe recording, the participant is able to exhibit a natural walking pattern without the influence of targeting a force plate positioned on the floor (Tekscan, 2010). The features of the F-Scan system are similar to wireless capabilities and low-profile sensors (0.15 mm thick, pressure ranges from 345 to 862 kPa, and each F-Scan insole comprises approximately 960 sensing elements (3.9

sensels per cm²) (Coda & Santos, 2015). The F-Scan system has been used for adult podiatric management. It can collect accurate, repeatable gait recordings utilising very high spatial decisions that can reach one millimetre as well as automatically averaging data points across multiple steps within one trial (Lord, 1997 cited by Pataky 2012; Randolph et al., 2000; Coda & Derek, 2015). At present, no evidence is available with regards to the repeatability and reproducibility of the F-Scan system in weight-bearing or obese adults. The findings of this research may support the application of modern in-shoe technology for future gait analysis clinical trials to clearly identify preventable biomechanical issues related to the obese adult, who is at risk of ulceration due to an imbalance in pressure distribution between the forefoot and the rear foot during walking (Ko et al., 2012).



Figure 2.15. In-shoe pressure systems (Chong et al., 2015)

2.6.3.1. Barefoot plantar pressure measurement

Platforms or mat systems have been constructed for measuring dynamic or static pressure and vertical F used for the plantar of the foot (Orlin & McPoil, 2000; Abdul-Razak et al., 2012). There are various dimensions, shapes, and resolutions of pressure mats available to accommodate a vast range of uses (Orlin & McPoil, 2000; Abdul-Razak et al., 2012). Naturally, a platform tool made up of various small electronic pressure sensors arranged in a matrix form on a circuit layer, is put among the two heat stable polyester layers (Castano & Flatau, 2014; Chong et al., 2015), as shown in (**Figure 2.16**). Mat resolution depends on the density of sensors in the

pressure sensing mat (i.e. the resolution increases with an increase in the number of sensors).

Using platform devices for gait investigations have several notable benefits. As the pressure mat is thin, it is sufficient enough to allow a non-intrusive assessment and is ideal for assessment of force without disturbing the dynamics of a trial participant, thus the system is suitable for capturing barefoot movement (Orlin & McPoil, 2000; Abdul-Razak et al., 2012). Likewise, the high density of pressure sensors within a mat system leads to the precise capturing and transmitting pressure data (Orlin & McPoil, 2000).

The pressure mat device can report the vertical force as precisely as the force plate. The mat method is generally put on a flat surface parallel to the supporting ground like the force platform system (Orlin & McPoil, 2000), presented in (Figure 2.16). In addition, the sensor mat can be constructed in various resolutions and sizes with a variety of shapes that fit with specific clinical applications, such as assessing the pressure distribution between a patient's body and various supporting surfaces like seats, backrests mattresses, and cushions. These systems are often embedded into the floor or walkway and are commonly used in the analysis of barefoot pressure (Harada et al., 1999). However, there are a number of drawbacks that have been identified by previous studies regarding the employment of the platform system to capture foot pressure (Orlin & McPoil, 2000; Shi, 2001). The walkway pressure systems are better suited to reporting the pressure of bare feet in indoor environments, but they have limitations in the analysis of physical activity (Orlin & McPoil, 2000). In addition, to enhance the accuracy of collected pressure data, the subject is required to repeat each trial (crossing the walkway) between three and five times, which may be considered inappropriate and uncomfortable for patients who suffer from excessive ulcerations in their feet (Orlin & McPoil, 2000). Generally, a platform or mat system need a large environment with an appropriate setup, as a pressure mat is required to be put on a walkway to include that the trailed participant attains a steady-state walk (least three steps are gathering) prior to arrival at the mat (Shi, 2001).



Figure 2.16. Platform, walkway or mat systems (Orlin & McPoil, 2000; AL-Baghdadia et al., 2015)

2.6.4. Sensor calibration

The electric resistance of a sensor is technically changed when a force is applied to it (Noce, 2005; Rana, 2009; Flórez & Velásquez, 2010). The loaded sensor can give readings in (either ohms or volt) units (Noce, 2005; Rana, 2009). Thus, sensor calibration has the fundamental procedure of converting the sensor output data into pressure units (Noce, 2005; Rana, 2009). Calibration is important for determining the static and dynamic characteristics of the sensor, (such as, time drift, hysteresis, and repeatability) which are necessary for applications that require force measurement accuracy. The manual sensor calibration consists of two methods. In a number of commercial foot pressure measuring systems (for example, Tekscan system), these methods are sensor equilibration and sensor calibration. The sensor equilibration process is required to minimise or compensate for small variations (i.e., hysteresis, time drift, non-linearity, creep, and repeatability) between the sensing elements on any sensor. Technically, variations result from the unique characteristics of each sensor. Moreover, the sensor calibration process converts the raw digital output data of the pressure sensor to actual pressure units, such as KPa. Based on published literature (Symiya et al., 1998; Noce, 2005; Rana, 2009), the current inquiry considers both sensor calibration procedures (sensor equilibration and sensor calibration) as a single process which is referred to as a sensor calibration process. Selecting appropriate calibration procedures can reduce the mean error in the reading

of loaded sensors from 31 to 11% (Symiya et al., 1998; Shi, 2001). Sensor calibration can be achieved by using dynamic and static calibration methods. Applying both methods is considered necessary for improving sensor performance over time and the accuracy of the sensor outputs (Noce, 2005; Rana, 2009). Therefore, these calibration procedures are recommended for calibrating all foot pressure measurement systems (Noce, 2005).

2.7. Discussion

This review has provided a detailed discussion of some of the main causes and impacts of weight gain in humans. Furthermore, the major factors influencing the distribution of plantar loads are considered in the development of the research methodology and the selection of participants in the investigation. A review of works that have been done used normal, obese and simulated participants. It also identified a number of scientific arguments with respect to the selection of an appropriate type of loading measurement system. Therefore, the significant key points of this chapter can be summarised as follows:

- The distribution of load on the shoe varies between individuals due to a wide range of factors such as variations in walking speed, foot shape, body weight, human age, and foot abnormality. However, it has been decided that the variation in weight has the greatest influence on changing foot load distribution;
- 2. In measuring plantar loading during gait, it is necessary to apply both static and dynamic calibration procedures for calibrating pressure measuring systems before using them;
- 3. In-shoe and mat pressure systems have suitable measuring systems and they are commercially available for recording the load acting between the plantar surface and a support surface. However, while each system has many advantages and limitations, the in-shoe is considered ideal for this research study.

2.8. Chapter summary

This chapter presented a literature review of 1) the effects of weight on plantar load distribution, 2) techniques recently utilised to record the loading data applied to both barefoot and in-shoe systems; and 3) the merits and drawbacks of each technique. Various theories associated with foot load distribution were reported. Body weight seems to have major influences on the distribution of the plantar load. However, the impact of these factors on gait have not been clarified in the literature. An intensive investigation of the previous literature was done in this study to find the research gap. The author found that there are limited studies that have investigated the plantar loading differences for the same weight category during gait. In addition, the review examined plantar loading measuring systems. The main emphasis focused on pressure insole methods used for measuring plantar load. A comparison between these measurement practices was performed to illustrate the feasibility of each for gait studies. The comparison found that the pressure insole is recommended for indoor as well as outdoor gait research.

The next chapter addresses the methodology that has been chosen during this research, including the instrumentation, software and participants.

3. METHODOLOGY

3.1 Introduction

This chapter focuses on describing the methodology chosen for this work and outlines the protocol used in subsequent experimental work in terms of foot plantar pressure patterns. As has become evident from the preceding chapters, foot loading has been extensively studied, especially in normal gait. In order to understand the true meaning of foot loading its measurement needs to be integrated with other measures of increased body weight. Additionally, general information on the participant's recruitment is included. Lastly, data analysis is described on which the experiments are based.

3.2 Instrumentation and software

The detailed instrumentation and software of the present study can be categorised as follows:

3.2.1. In-shoe sensor

The in-shoe pressure assessment device (Tekscan, Boston, MA, with overall mass of 1.3 kg) was utilised to collect and evaluate the dynamic distribution of plantar foot pressure data (pressures on the sole of the foot). Recently, they have been used in a variety of studies measuring plantar pressure (e.g., Armstrong et al., 1998; Chesnin et al. 2000; Chong et al., 2017; Pirozzi et al., 2014; Waaijman & Bus, 2012). It is a system of sensors, cuffs, ankle bands, cuff cables, waist belt, USB hub, power supply and a USB cable.

Participant setup with the F-Scan® is required and undertaken by:

- 1. Managing the cables which are five metres in length and linked to the subject's waist.
- 2. Trimming the sensors to fit any shoe presented.
- 3. Preparing the shoes by attaching (pasting the sensors into the shoes by medical adhesive) an insole system to the original insole of the shoe, as presented in (Figure 3.1 A and B).

This system comprises a wireless transmitter and two sensor cuffs. The inshoe sensor, F-ScanR 3000E standard was utilised. The sensors are approximately (2 mm) thick and comprised of 960 sensing elements. Plantar pressure data was recorded in kilopascals (kPa) at a frequency of 100 Hz. The frequency (100 Hz) was utilised for all trials in this work. According to Castro et al. (2011); Debbi et al. (2012); Munoz-Organero et al. (2016), 100 Hz frequency is sufficient to record the true peak plantar pressure values during gait. Insole calibration was carried out at the participant's own body mass in the initial data gathering, based on the manufacturer's instructions. This frequency was sufficient to record the true peak plantar pressure values during gait (Ramanathan et al., 2010; Healy et al., 2012). Sensor positions were determined for each participant according to their foot size. Five pairs of insoles were utilised to accommodate varying foot sizes. Each insole consists of 960 sensing elements which can then be used to determine the average of the plantar pressure parameters (force, contact area, contact pressure, pressure time integral and peak pressure) over the ten regions of the foot, namely heel (H), midfoot (MF), first metatarsal head (1MH), second metatarsal head (2MH), third metatarsal head (3MH), fourth metatarsal head (4MH), fifth metatarsal head (5MH), hallux (1stT), second toe $(2^{nd}T)$, and third to fifth toes $(3^{rd}-5^{th}T)$.

The in-shoe system captures plantar pressure data during foot movement and seems to be very sensitive to changes in pressure during walking. The in-shoe pressure data-acquisition system was utilised to measure the distribution of plantar pressure in all the experiments in this study. The in-shoe measurement system is preferred, and assessment of pressure at the foot-shoe interface is demanded (Cavanagh et al., 1992). The insole pressure system is shown by the presentation to be a better trade-off to do a gait analysis. The in-shoe measurement system is also capable of capturing multiple footstep data from both feet at the same time and record foot function patterns directly inside the shoe. During in-shoe recording, the participant is able to exhibit a natural walking pattern without the influence of a force plate positioned on the floor (Tekscan, 2010). The measurement system was calibrated up to 100 kPa using the F-Scan calibration device according to the manufacturer's manual. The insoles measurement system was assessed on a daily basis, before and after data collection, to ensure the accuracy of the individual sensors. The features of the F-Scan system are similar to wireless capabilities and

low-profile sensors, 0.15 mm thick, pressure ranges from 345 to 862 kPa, and each F-Scan insole comprises approximately 960 sensing components (3.9 per square centimeter). The unique sensor design allows the sensor to be trimmed to fit almost any shoe, as presented in (**Figure 3.1 C and D**) (Coda & Santos, 2015; Patrick & Donovan, 2018).

This study examined the repeatability of the insole system. Ranges of pressure distribution and contact times are important parameters that can be referred to by clinicians when investigating whether the obese may develop foot pathologies. Pathological foot defects in the foot, e.g. ulcer in the diabetic foot, may cause excessive weight (loading) on certain parts of the plantar surface, and can have serious consequences for children's and adults' feet. Various foot problems and injuries are associated with these pathological feet discoveries (Dananberg, 2000; Bus et al., 2005). These foot problems and injuries are common across all sectors and ages of the community (Bus et al., 2005).

The in-shoe pressure-measuring system has previously demonstrated reliability, as shown in a study which reported a between-trial intraclass correlation coefficient (ICC) of 0.99 (McPoil et al., 2001; Vela et al., 1998). Step calibration was performed immediately prior to each participant's analysis (Coda & Santos, 2015). Each participant was calibrated using the Tekscan software while wearing the F-Scan belt, the connecting wires, the battery and finally the ankle cuff (**Figure 3.1 C and D**).

The study initially attempted to calibrate the pressure measurements to kilopascals, an accepted unit for plantar pressure measurement (Vela et al., 1998; Patry et al., 2013). The F-Scan software accompanying the pressure measurement system was used to calibrate the sensors according to each participant's body weight prior to data collection, and record approximately six to seven steps (stances) per foot at 100 Hz for each participant. Once all walking trials were completed, plantar pressure data was saved on the computer in the laboratory then transferred to an exchangeable memory card for processing and analysis. All information was gathered and treated using the F-Scan (version 6.70) research software.



Figure 3.1. In-shoe pressure system (pressure sensor). (A) Flexible, thin in-shoe sensors. (B) Inserting the insole system inside the shoe (C) Represents how each participant was set up with the F-Scan system prior to each calibration and recording sensors. (D) Represents a typical plantar pressure recording obtained using the insole system, where high pressure areas can be easily identified.

3.2.2. Shoes

Five pairs of standard Australian men's size shoes were used. The shoe sizes of the male participants were between size 9 and size 11. The standard shoes were provided to participants when they undertook the test and were prepared by attaching an insole system to the original insole of the shoe with tape and non-slip socks (Burnfield et al., 2004) as presented in **Figure 3.1 (B)**. As mentioned earlier, footwear also has a considerable effect on plantar pressure distribution (e.g., Schaff & Cavanagh 1990; Sarnow et al., 1994; Nyska et al., 1995; Perry et al., 1995). Therefore, to minimise any effects of shoes on performance, every subject wore the same type of shoe in every experiment. In addition, the 'best-fit' insoles were carefully selected for every subject.

3.2.3. F-Scan software

F-Scan Research ® software is a gait analysis package that comes with the Matscan system (Tekscan). F-Scan software (Tekscan, Inc System; Version 6.70-03) was used to analyse data and evaluate the value of the pressure data for specific locations under the foot sole (Pirozzi et al., 2014). Pressure data was obtained at 100 Hz and processed as described above, as presented in (**Figure 3.2**).



Figure 3.2. FScan software

3.2.4. Wall stadiometer

A portable 200 cm wall mounted stadiometer, shown in (Figure 3.3) was utilised to measure the participant's height in centimeters accurate to the nearest 0.1 cm (+/-1mm).





3.2.5. Body composition scanner scale

The scale is a model SC-512 health station with 160 kg /350 lb. capacity electronic scale and was used to measure body weight. This scale is easy to use and setup, as shown in (**Figure 3.4**).



Figure 3.4. Body composition scanner scale

3.2.6. IBM SPSS statistics

The statistics software package IBM SPSS statistics (version 25) was utilised for interactive or batched statistical analysis, as indicated in (Figure 3.5).



Figure 3.5. IBM SPSS statistics version 25

3.2.7. Iron mass

To modify participants' weight for simulation, nine pieces of metal weighing 3 kg with the dimensions of length, width, and thickness (26.2, 7.4 and 2 cm), respectively were used. An image of the metal is presented in (**Figure 3.6**).



Figure 3.6. Show the piece of metal

3.2.8. Army vest.

An army vest was used to simulate body weight. It is flexible in size, so it can be fitted to every participant during testing, as shown in (**Figure 3.7**).

Back



Figure 3.7. Army vest jacket

3.3 University approval and experiments

3.3.1 Ethical approval

The experiments were carried out at the University of Southern Queensland located in Toowoomba (Queensland, Australia) during 2017 and 2018. Prior to participating in the study, each subject read and signed an informed consent form approved by the University. All data collected in this study was stored in accordance with the protocols approved through the National Statement on Ethical Conduct in Human Research out of the Research Ethics Committee (proposal number: H18REA065), and all participants freely signed an informed consent form based on the National Statement on Ethical Conduct in Human Research (2007). The informed consent form explained the purpose and the procedures of the study and that full ethics approval had been granted. The authors reported to all individuals, the steps and the process for gathering data; likewise the privacy guarantees. The subjects wore standardised footwear, and the test was organised during university hours (9:00 am-4:00 pm) and took place within a five-week period.

3.3.2 Ethical issues

Informed consent - All participants were aged between 24-50 years and enrolled at the university. They were sent a letter to complete and were asked to sign an assent form prior to entering the project questionnaire. The information letter included my direct phone number and subjects were asked to contact me if they had any inquiries.

Participants who consented to enrol in this study were contacted by telephone and were interviewed at a mutually convenient location. Before the test (data collection) commenced, the purpose of the research was explained, and interviewees were encouraged to ask questions if they were unsure about the test process. The interviews were recorded with the participants' consent on the basis that only I would listen to the interview and that the full written transcript of the interview would be seen only by my supervisor and myself.

Right to withdraw - Individuals who did not wish to take part in the study were asked to opt out of the study by completing and sending a message to ensure that they were not contacted again. Interview participants were informed that they did not have to answer an interview question if they did not wish and that they had the right to withdraw from the interview at any point.

Anonymity - As the main investigator, I was the only person to have access to participants' names, age, and addresses. In order to maintain participant anonymity, all questionnaires were assigned an identity number.

A copy of the **Ethical approval** and **consent forms** given to the respondents can be found in **Appendix A and Appendix B**.

3.3.3 Participants

All adults participating in these studies were students and staff at the University of Southern Queensland. Forty-six subjects participated in the various experiments chosen for this study. Participants were excluded from participation in the study if they disclosed a history of orthopaedic problems, suffered from pain in their feet or injury of the lower limbs, had any surgery on their feet at any time in their life, including the orthopaedic surgery patients, if they are classified as being underweight, leg length discrepancies, if they have foot abnormalities (e.g. missing toes), abnormal walking gait,

any problems of balance, musculoskeletal dysfunction or any difficulties in independent locomotion neurological and/or musculoskeletal problems likely to affect their gait. Tests were initiated between June 2017 and May 2018. **Table 3.1** shows the physical characteristics of the subjects. Experiments in this study were based on previous research. The first experiment comprised fifteen adult subjects, as is reported in **Section 3.4.1**. The second experiment consisted of thirty-one participants, and the same these participants were utilised for collecting data in the third experiment, as described in **Section 3.4.2 and 3.4.3**.

| Experiment | Subjects | Age (y | ears) | Weigh | t (Kg) | Height (m) | | BMI (Kg/ m ²) | |
|------------|---------------------|--------|-------|-------|--------|------------|------|------------------------------|------|
| | | Mean | SD | Mean | SD | Mean | SD | Mean | SD |
| Exp. 1 | Fifteen subjects | 35.7 | 5.83 | 81.40 | 14.09 | 1.73 | 0.08 | 27.19 | 4.87 |
| Exp. 2 | Thirty-one subjects | 37.5 | 6.92 | 89.82 | 17.92 | 1.74 | 0.08 | 29.7 | 5.1 |

Table 3.1. Physical characteristics of the subjects mean and SD

3.3.3.1 Measurements to describe the participants

The following information was collected from all participants and used to describe the universal characteristics of the research sample:

3.3.3.2 Demographic characteristics for data analysis

A questionnaire that included questions about age, race, level of education, and past medical history was utilised. Properties such as age, gender, height, weight, BMI, and foot size were assessed and recorded, as well as all the participant demographics. The study participants were gathered into three groups according to their BMI values. All measuring instruments were calibrated.

3.3.3.3 Height

Participants removed their shoes and stood up straight with their heels together, took a deep breath, and held it while looking straight ahead in order to ensure reliability and accuracy of their height measurement. Height was measured to the nearest 0.1 cm, utilising a portable 200 cm wall mounted height measuring tape. Height values were expressed as centimeters (WHO, 2008).

3.3.3.4 Weight

Weight was measured with a digital weighing scale. Weight: body weight was measured with a body composition scanner scale (model SC-512 health station with the 160 kg capacity). The scale was kept completely still, and then the weight was reset to zero, the scale was calibrated before each participant was weighed. Participants were asked to wear light clothing only: shorts and a T-shirt, of the same type, standardised to ensure consistency of measurements for all participants. Measurements were taken prior to the test to ensure consistency of measurement. In addition, vest weight was recorded

3.3.3.5 Body mass index

Participants were classified according to their body mass. Both height and weight were measured in accordance with standardised procedures (WHO, 2016). This information was used to calculate the body mass index (BMI) for each subject, to demonstrate their weight class based on the WHO (2016). BMI will be calculated by **Equation (3.1)**:

$$BMI = \frac{Weight (Kg)}{Height^2 (m^2)}$$
(3.1)

3.3.3.6 Dynamic plantar pressure assessment

All participants in all experiments were given time to familiarise themselves with the process of walking with the insole system to ensure they were comfortable with the procedure. Throughout data collection, all participants were encouraged to adopt a natural gait pattern and to walk at a self-selected speed. Three complete trials of the foot were recorded for each participant (Butterworth et al., 2015).

Laboratory tests (experiments **1**, **2**, **3**) were completed with the independent variable being body mass index. To analyse the plantar pressure distribution, the foot was partitioned in ten regions, as adapted from other studies (Xu et al., 2017; Putti et al., 2008). Within each region of interest, force, contact area, contact pressure, pressure time integral and peak pressure were obtained. The dependent variables

were measures of plantar pressure (force, contact area, contact pressure, pressure time integral and peak pressure) over the ten regions of the foot, namely heel (H), midfoot (MF), first metatarsal head (1MH), second metatarsal head (2MH), third metatarsal head (3MH), fourth metatarsal head (4MH), fifth metatarsal head (5MH), hallux (1stT), second toe (2ndT), and third to fifth toes (3rd-5thT) (**Figure 3.8**), using F-Scan software (Version 6.70-03, Tekscan, Boston, MA).



| Mask | Region |
|----------------------|------------------------|
| Н | Heel |
| MF | Midfoot |
| 1MH | First metatarsal head |
| 2MH | Second metatarsal head |
| ЗМН | Third metatarsal head |
| 4MH | Fourth metatarsal head |
| 5MH | Fifth metatarsal head |
| 1 st T | Hallux |
| 2 nd T | Second toe |
| 3^{rd} - $5^{th}T$ | 3–5 Toes |

Figure 3.8. Plantar foot divided into ten regions

3.4 Experiment design

3.4.1 Experiment 1: foot loading pattern variations between healthy, overweight, and obese adults

Fifteen adult male participants were recruited. They were aged between 25-46 years. Adults were excluded from participation in the study if they disclosed a history of orthopaedic problems, any disorder of balance, musculoskeletal dysfunction or any discomforts in the independent movement, neurological and/or musculoskeletal problems likely to affect their gait. The participants were divided into three categories of healthy weight (BMI <25), overweight (25< BMI <30), and obese (BMI >30) as displayed in (**Figure 3.9**). Participants were grouped according to BMI, and by following similar protocols to those used with children and adults, as described by Filippin et al. (2007); Hills et al. (2001); Pirozzi et al. (2014); three groups were obtained for comparison. Participant demographics are presented in **Table 3.2**.

Testing protocols for the first experiment were as follows. The protocol for the assessment of plantar pressure in this experiment was the three-step method with three trials. The plantar pressure insole was calibrated for each subject. In each category, three valid trials were collected for each foot, as this number of trials has previously been found to be sufficient to ensure adequate reliability of force and pressure data (van der Leeden et al., 2004; Hughes et al., 1991). The average of the three measurements was used to describe the range of motion of the ankle, subtalar, and first metatarsophalangeal joints for each subject (Arndt et al., 2004; Leardini et al., 1999). Participants were admitted for ten minutes where they could practice gait at a self-selected velocity with an insole system. Walking speed was not controlled because, although a prescribed walking speed might help to compare the pressure manners of various participants, it would stop the generation of a normal walking manner. In addition, using a metronome might cause an atypical step (Rosenbaum and Becker, 1997; Monteiro et al., 2010). All participants were asked to walk along a five-metre walkway with an insole system that was attached to the participants' shoes as shown in (Figure 3.1(C).

| Table | 3.2. | Participant | information | for | each | of | the | three | groups | of | participants |
|--------|------|-------------|-------------|-----|------|----|-----|-------|--------|----|--------------|
| (Exper | imen | t 1) | | | | | | | | | |

| Group | Ν | Age (years) | | Heigh | t (m) | Weigh | t (kg) | BMI (Kg/m ²) | | |
|------------|---|-------------|------|-------|-------|-------|--------|--------------------------|------|--|
| | | Mean | SD | Mean | SD | Mean | SD | Mean | SD | |
| Healthy | 5 | 35.40 | 6.35 | 1.78 | 0.09 | 69.42 | 7.48 | 21.88 | 1.26 | |
| Overweight | 5 | 36.20 | 4.76 | 1.68 | 0.05 | 76.12 | 6.61 | 26.82 | 1.60 | |
| Obese | 5 | 35.4 | 7.4 | 1.7 | 0.1 | 98.7 | 2.9 | 32.9 | 1.7 | |



Figure 3.9. Participants were grouped within three categories based on their BMI: (A) healthy, (B) overweight, and (C) obese

3.4.2 Experiment 2: simulation body weight protocol and determination of plantar pressure

An experimental design using subjects and repeated measures were employed for this study. This research was in a laboratory environment and exploratory in nature. The

participant was chosen among the suitability of students and staff. The participants of this study were limited to 24 to 50 year old adults and comprised thirty-one adult males. The mean of all following features were gathered: age 37.55 ± 6.92 years, height 1.74 ± 0.08 m, the body mass 89.82 ± 17.92 kg and BMI 29.7 ± 5.1 kg/m². Healthy was defined as free of cardiac and/or neurological problems, such as arrhythmia, heart attack, or stroke, and free of any musculoskeletal or pathological problems (osteoarthritis, bone fractures or shin splints). Participants did not have a history of major lower extremity injury and/or surgery within the last year. These conditions were screened using a medical-history form.

In this experiment, the author replicated the basic structure of a previously published study design investigating the effects of increasing weight on peak plantar pressures (Vela et al., 1998) but with modification in terms of the changes of added mass (simulation body weight protocol); all the subjects wore jackets. That study had added (9.1 and 18.1) kg to asymptomatic participants wearing normal shoes, without reference to the particular BMI or weight.

Testing protocol for the second experiment adhered to protocols for the assessment of plantar pressure in adults previously used in a laboratory (Ramachandra et al., 2012) because the three-step gait protocol showed enhanced reliability to measure the walking and plantar pressure pattern compared with the one step gait protocol and the midstance protocol. It was decided that the best protocol to use for this investigation was the three-step method. The participants continued to walk ten metres while looking straight ahead until three valid trials were captured bilaterally. In each condition, three valid trials were collected for each foot, as this number of trials has previously been found to be sufficient to ensure adequate reliability (AL-Baghdadi et al., 2011; Barisch-Fritz et al., 2014; Ramachandra et al., 2012; van der Leeden et al., 2004; Yan et al., 2013; Yan et al., 2014; Yan et al., 2016). Once all walking trials were completed, plantar pressure data was saved on the computer in the laboratory then transferred to an exchangeable memory card for processing and analysis. Mean values of the foot measures were calculated across the three trials for each foot (Arndt et al., 2004; Leardini et al., 1999); each trial consisting of an average of the three-step method over ten meters. The plantar pressure insole was calibrated for each subject.

Participants were provided with standard shoes when they undertook the test, prepared by attaching an insole system to the original insole of the shoe with tape and non-slip socks (Burnfield et al., 2004). Each participant was calibrated using the Tekscan software while wearing the FScan belt, the connecting wires, the battery and finally the ankle cuff. All information was gathered and treated using the F-Scan (version 6.70) research software. Plantar pressure data was obtained at 100 Hz and processed as described. The frequency (100 Hz) was utilised for all trials in this work. 100 Hz frequency is satisfactory to record the true peak plantar pressure values during gait (Ramanathan et al., 2010; Healy et al., 2012).

The pressure data collection protocol involved pressure measurements with an additional four separate loading conditions (0, 10, 20, and 30 kg of iron mass respectively). To ensure that participants felt as comfortable as normal with the increase in body weight, each subject acted as their own control. Loading conditions to simulate an increase in body mass were achieved with the application of a weighted vest. Subjects completed each walking trial on a 10-meter walkway. As we added weight, we tried as much as possible to distribute the weights equally between the front area (chest) and the back area (back) in order to simulate and obtain a weight tolerance for normal weight gain as shown in (**Figure 3.10**).



Figure 3.10. Modification of volunteer body weight by adding weight evenly distributed anteriorly and posteriorly using vests. State one walk without any additional weight; state two (10 kg); State three (20 kg); State four (30 kg): (A) Front view, (B) Back

For each participant, the weight was raised from a range of variance BMI (20-46.4) kg/m² with the vest included, and then filled with iron bars and fixed in the central area of each subject. The mass placed inside the vest ranged from 10 to 30 kg (mean mass = 20 ± 10 kg). Normal walking was performed for 20-minute sessions with subjects wearing the vest type mass of a 10 kg, 20 kg and 30 kg external load (loaded walking) (Lee et al., 2008). The participants wore the jacket then attached shoulder straps and two side straps so that it did not obstruct any upper or lower body moves. Three trials were repeated in each simulation condition. However, the first trials were considered as practical sets to attain a physiological steady state, and only the rest of the three trials were analysed. Later, the subjects wore the jacket and walked the same way as mentioned earlier.

3.4.3 Experiment 3: profiling the centroid of the area of contact captured

The same thirty-one adult male subjects who were used in experiment two, were used for collection of centroid data in experiment three. Only adults who possessed a neutral foot position were involved, although subjects who had similar foot arch features were recruited to participate in the experiment. The participants had what could be considered a normal (AHI = 0.328 ± 0.022 mm) foot arch as determined by a musculoskeletal physiotherapist (Roposh, 2010). All the participant demographics are presented in **Table 3.3**.

| | Ν | Age (ye | Age (years) | | (m) | Weigh | t (kg) | BMI (Kg/m ²) | | |
|---------|----|---------|-------------|------|------|-------|--------|--------------------------|-----|--|
| | | Mean | SD | Mean | SD | Mean | SD | Mean | SD | |
| Subject | 31 | 37.55 | 6.92 | 1.74 | 0.08 | 89.82 | 17.92 | 29.7 | 5.1 | |

Table 3.3. Participant characteristics of the study population

Prior to the experiment, the F-Scan sensor sheets were tailored to make sure they had the same shape and size as the instrumented insoles. They were then pasted underneath the instrumented insoles; it was then noted that the instrumented insole was placed in between the foot and F-Scan sensor sheet, as shown in (Figure 3.1, A & B). The procedure was similar to experiment two. The participants were instructed to practice a three-step gait. When the subjects were confident, their gait was recorded. And the recording of centroid data in each three-foot position (heel strike, midstance, and toe-off) were divided according to the weight added. Next, each participant performed three sets of three steps, which were repeated with each weight that was attached to the participant's body. Normal walking was performed for 20minute sessions.

An innovative technique for the calculation of the centroid of the area of contact captured by the system was developed, with a reference line on every plantar pressure pedograph (Pataky et al., 2008; Chong & Milburn, 2017), which is also introduced in this section. The centroid of the area of contact with the surface is located by calculating the geometric centre of a set of cloud points having the lowest z coordinate value as shown in (**Figure 3.11**). In this case, the z value is the surface height of the insole measurement system and is equal to 0.0 mm. A modified version of the algorithms developed by Yu et al. (2007) is used in the centroid calculation. Generally, a centroid position can be calculated for every computed between foot strike and toe-off. The method may be used to determine the centroid of the plantar surface at any z value or a range of z values.

$$X_{C} = 0.0 = \frac{\sum_{i=1}^{N} X_{i}}{N}$$
 (3.2)

$$Y_{C} = \mathbf{0}. \mathbf{0} = \frac{\sum_{i=1}^{N} Y_{i}}{N}$$
 (3.3)

In this case, the z value is the surface height of the insole measurement system and is equal to 0.0 mm. xc and yc are the centroid coordinates, and xi and yi are the coordinates of the individual points in a cluster of point clouds during the three-step method within three trials. In this part, a foot pressure sensing insole has been used to calculate the moment of heel strike, midstance and the toe-off phases by using F-Scan software (Version 6.70-03, Tekscan, Boston, MA), as shown in (**Figure 3.11**).



Figure 3.11. Determining the centroid of the heel strike during gait. The one-step sequence of gait cycle phases which start from heel strike, midstance and toe-off phases.

3.5 The method used in data analysis

All data was analysed using statistical software (SPSS version 25.0). Statistical significance was accepted at the P < 0.001 level of confidence.

3.5.1 The coefficients of variation:

The repeatability coefficient (CR) is a precision measure which represents the value below, which is the absolute difference between two repeated test results that may be expected to lie with a probability of 95% (Hopkins, 2000; Chong, 2016; Putti et al., 2008). The analysed variables for gait are measured in each selected area during stance; the CA is determined by the sum of the area of all overloaded sensors within an area (Putti et al., 2008; Chong, 2016). The CR is calculated using SD/mean multiplied by 100, as in the equation below:

$$CV = SD/mean \times 100\%$$
 (3.4)

3.5.2 Analysis of variance

Analysis of variance (ANOVA) is a statistical method utilised to verify the significance of the differences between the averages of three or more groups in a single dependent variable. It is intended to compare averages or to arrive at a decision about whether there are differences between the performance averages of

groups that have been subjected to various processes to arrive at agents that make some means different from other means.

3.5.3 Intra-class correlation coefficients

Intraclass correlation coefficient (ICC) methods have been widely utilised for estimating or evaluating the quantitative measurements obtained by different raters or assessors. According to Donner and Koval (1980), reliability is the extent of the consistency or stability of measurement that exposes the steadiness in the repeated measurement to the same subject under identical conditions. Therefore, a reliable test can be obtained if the random error or unsystematic variation in measurement in the repeated measurements is minimal. A number of statistical approaches are employed to describe the reliability of the measured data. ICC is normally utilised in assessing the quantitative measurements obtained by different rates or assessors. Typically, the magnitude of ICC ranges between zero and one. A perfect agreement between assessors happens if the value of the ICC approaches one, whereas an imperfect agreement between the assessors occurs if the ICC value approaches zero. Currently, ICC has been commonly used in conservative care medicine to evaluate inter-rater, test-retest, and intra-rater reliability (Cramer et al., 2010; Owens et al., 2004; Koo et al. 2011; Houweling et al., 2014; Clare et al., 2003; Battaglia et al., 2014; Leach et al., 2003; Russell et al., 2012; Koo & Li, 2016). See below for definitions of each estimate:

- Interrater reliability considers the differences amidst two or more raters measuring the same set of participants.
- Test-retest reliability must invert the differences in assessments taken via an instrument on the same participant under the same situations. It is normally indicative of reliability in conditions when raters are not involved, or rater impact is negligible, like a self-report survey instrument.
- Intrarater reliability considers the different data measured by one rater across two or more trials.

There are various types of ICC published literature has classified ICC into three different classes (field, 2005) depending on model and form of ICC, which are presented in **Table 3.4**.

Table 3.4. Different cases for calculation ICC depending on model and form of ICC

| ICC type | Description |
|----------|--|
| ICC(1,1) | Each subject is assessed by a different set of randomly selected raters, and the reliability is calculated from a single measurement. Uncommonly used in clinical reliability studies. |
| ICC(1,k) | As above, but reliability is calculated by taking an average of the k raters' measurements. |
| ICC(2,1) | Each subject is measured by each rater, and raters are considered representative of a larger population of similar raters. Reliability is calculated from a single measurement. |
| ICC(2,k) | As above, but reliability is calculated by taking an average of the k raters' measurements. |
| ICC(3,1) | Each subject is assessed by each rater, but the raters are the only raters of interest. Reliability is calculated from a single measurement. |
| ICC(3,k) | As above, but reliability is calculated by taking an average of the k raters' measurements. |

ICC estimates and their 95% confidence intervals were calculated using SPSS statistical package version 25. The following boundary conditions were used with respect to the ICC: 0.75 and 0.90 limits for good to excellent or a high or good reliability (Youdas et al., 1991; Portney & Watkins, 2015), 0.5 and 0.75 moderate reliability, and less than 0.5 poor reliability (Watkins & Portney, 2009).

These evaluations are essential to clinical assessments because without them, we would have no confidence in their measurements, nor could they draw any rational conclusions from their measurements (Koo & Li, 2016). ICC statistical techniques are used to evaluate the reliability among several measurements. Different cases for calculating ICC depending on models vary depending on types of trials (judges) and targets (subjects). Based on Shrout and Fleiss (1979), the cases of ICC, the study used a full factorial design with two independent variables. Namely, tenfoot regions, and simulated body weight. Each participant received the same four conditions, walk without weight and simulated with three different weights (10, 20

and 30 kg). Additionally, the ICC for each measure in each condition were calculated to investigate its reliability using SPSS. The ICC formula is shown below:

1. Model 1: each subject is assessed by a different set of randomly selected raters, one-way random effects model and form:

ICC (1, K) =
$$\frac{BM_S - WM_S}{BM_S}$$
 (3.5)

Where, BMS is the between-subjects mean square from the analysis of variance, and WMS is the within-groups (error) mean square. K is the number of ratings for each subject.

2. Model 2: each subject is assessed by each rater and raters are randomly selected, two-way random effects model and form:

ICC (2, k) =
$$\frac{BM_{S} - EM_{S}}{BM_{S} + \frac{RM_{S} - EM_{S}}{n}}$$
(3.6)

Where: EMS is the error mean square, RMS is the between-raters mean square, k is the number of raters, and n is the number of subjects tested.

3. Model 3: each subject is assessed by each rater, but raters are the only items of interest, two-way mixed model and form:

ICC (3, k) =
$$\frac{BM_S - EM_S}{BM_S}$$
 (3.7)

For this dissertation, **Model 2** is used to calculate the reliability of gait variables since it is a repeated measures method. In addition, the sample size requirements are computed based on the ICC values. Detailed results will be introduced **in chapter 5**, sections 5.2.

3.6 Analysis of experiments

3.6.1 Analysis of experiment 1

The statistical procedures were done using the SPSS statistical package (version25; SPSS Inc, Chicago, IL, USA). Area analyses involved 10 anatomical regions which were calculated by the F-Scan software (**Figure 3.2**). The mean of the three repetitions for each subject was computed, and all the statistical procedures were performed with these mean values. Mean and standard deviations (SD) were calculated for each variable. A one-way within subject repeated measures analysis-of-variance (ANOVA) model was utilised to consider whether there were significant differences in the parameter for the groups with obese, overweight and normal weight conditions, with and without variation in BMI. This was used to test for significant differences in plantar loading. These were done on each of the dependent variables with categories of healthy and obese, as the independent variable (Kunter et al., 2005). Where differences existed between groups, Tukey post-hoc tests were used to compare variables and to ascertain the location of the difference between the three groups for both feet. The level of significance was set at P <0.01.

3.6.2 Analysis of experiment 2

The data was collected and analysed. The results were expressed as mean \pm SD values, and presented in the format of mean (SD). The SD of the between data differences identified in the ANOVA were used to determine the coefficient of repeatability (CR) of each parameter (Bland & Altman, 1986). This was expressed as a percentage of the mean by using the formula [(CR)/mean] - 100, i.e., the lower the CR, the stronger the repeatability (Putti et al., 2008). Repeatability of the three trials for each left and right foot, was calculated separately, and these values were used to calculate the mean CR for all measures. Data was tested for outliers and summarised using descriptive statistics. The methods of Bland and Altman (1986) were used to assess repeatability. As suggested by Hopkins (2000), CR was used to calculate the repeatability. Then the data was exported into Microsoft Excel 2016. Descriptive statistics were reported in terms of the mean, SD, range, and percentage of increases and decreases. The level of statistical significance was set at the (P \leq 0.05) level.

Variability in the data will be assessed via the calculation of coefficients of variation; this analysis of absolute reliability provides information regarding within-trial variability expressed as a percentage (with the associated P-values), based on the General Linear Model. CR expresses the typical error as a percentage of the mean and is defined as (SD/mean)×100. The higher the CR value, the weaker the repeatability. The differences between the measurements over two occasions and the averages from the trials were calculated to test repeatability. Using the F-Scan software, the foot was divided into ten areas and the five most clinically relevant parameters for both feet were assessed. Measurements from both feet were analysed separately for repeatability. These accounted for 100 evaluated variables.

Data was not normalised for foot size and weight. The standard deviation reflects within subject and between subject variations as well as trial-to-trial differences and variation in the in-shoe system (Gurney et al., 2008). For all participants left foot and right foot data of three trials were combined, which was assessed using the reliability of ICC. The size of the variations within these trials was reported as a CR. The intra-individual repeatability for the variables and duration of stance phase was verified by means of ICC, since the difference in results between test weights was random (Cavanagh & Ulbrecht, 1994; Wearing et al., 1999). The mean of the three trials of each subject was computed and all the statistical procedures were performed with these mean values.

3.6.3 Analysis of experiment 3

The data was collected and analysed. The results were expressed as mean \pm SD values, and presented in the format 'mean (SD)' for this experiment. The x-axis and y-axis coordinates can be shown by other methods, which include mean and standard deviation by recording three walks with 31 participants at the normal walking speed. The information was arranged in mean and standard deviation formats, considering the values of three steps each day, for each foot.

3.7 Assumptions

The following assumptions were made in this study:

1. Participants understood and followed the instructions closely;

2. Walking speed was not tested during this investigation because it was assumed that everybody's walk is normalised, therefore, all walk patterns should be the same between trials;

- 3. Participants were comfortable wearing the footwear provided;
- 4. The setting, a typical gait lab, did not influence gait pattern significantly.

3.8 Chapter summary

This chapter focused on the equipment, participants, procedures and inputs used in the search to collect data and provide access to the results related to the subject of research. The main conclusions derived from **chapter 3** are summarised below:

• Participants' properties were measured (height, weight, and BMI).

• Plantar pressure of the foot was collected, and data recorded by using the insole system.

• All data were identified and analysed.

The following chapters (chapter 4, chapter 5 and chapter 6) present the finding from the first, second and third studies for each of the three research objectives. The next chapter (chapter 4) begins with the findings of objective 1.

4. RESULT OF EXPERIMENT 1

4.1 Introduction

This chapter exhibits the outcomes that answer objective one, which studied the effect of differences in body mass (healthy, overweight and obese) in adults aged between 25 and 46 years and the associated effects on plantar pressure distribution characteristics by calculating the foot force, contact area, contact pressure, peak pressure and pressure time integral to different foot regions during gait, using the inshoe system. There are six main sections in this chapter: experiment details, force and contact area results, contact pressure and pressure time integral results, peak pressure outcomes, discussion of results and the chapter summary.

The three groups (five people in each group) of adult male participants were recruited into the study. Division of these groups was based on participants' body mass and BMI: normal weight, overweight and obese. The subjects were between 25 and 46 years of age, and the BMI rate obtained for obese adults (32.9 kg/m^2) was significantly higher (P <0.001) than that of the healthy group (21.88 kg/m^2). To study the effect of differences in the body mass of adults on plantar pressure distribution characteristics (calculating foot force, contact area, contact pressure, peak pressure and pressure time integral to different foot regions during gait) an in-shoe system was used. The characteristics are shown in **Table 3.2**.

4.1.1. Force and contact area measurements

Descriptive data for force (F, N) and contact area (CA, cm²) is shown in **Table 4.1**. There was a statistically significant variation (P <0.001) into the F measure. The lower differences in mean F measured value was located at the $3^{rd}-5^{th}T$ and $2^{nd}T$ between normal weight and overweight. The mean value for F within these groups at the $3^{rd}-5^{th}T$ were 6.21 N and 14.24 N, respectively. Moreover, the mean value for F under the $2^{nd}T$ were 7.53 N and 10.82 N within healthy, and overweight, respectively. Moreover, the biggest mean values of the F as appears in **Table 4.1**, was under the obese category followed by the overweight group compared with the healthy group, which were 91.70 N, 69.87 N to 55.31 N.

CHAPTER 4: RESULTS OF EXPERIMENT 1

| | | | Force | (N) | | | Contact area (cm ²) | | | | | | | |
|------------------------------------|---------|-------|------------|-------|--------|--------|---------------------------------|------|------------|------|-------|------|--|--|
| Foot region | Healthy | | Overweight | | Obese | | Heal | thy | Overweight | | Obese | | | |
| Foot region | Mean | SD | Mean | SD | Mean | SD | Mean | SD | Mean | SD | Mean | SD | | |
| 3 rd -5 th T | 6.21 | 6.74 | 14.24 | 8.90 | 14.99 | 8.38 | 0.74 | 0.26 | 1.17 | 1.01 | 1.37 | 0.15 | | |
| 2 nd T | 7.53 | 5.53 | 10.82 | 7.18 | 12.32 | 7.84 | 0.81 | 0.35 | 0.83 | 0.70 | 0.84 | 0.45 | | |
| 1 st T | 45.29 | 36.54 | 58.39 | 28.01 | 103.37 | 41.64 | 3.21 | 1.43 | 2.99 | 1.33 | 4.77 | 1.82 | | |
| 5МН | 26.06 | 15.91 | 31.02 | 15.10 | 37.69 | 19.46 | 2.57 | 1.44 | 2.87 | 1.37 | 3.16 | 1.38 | | |
| 4MH | 34.79 | 18.27 | 39.63 | 14.91 | 63.65 | 26.17 | 3.08 | 1.50 | 3.42 | 1.07 | 4.57 | 1.31 | | |
| ЗМН | 42.99 | 20.51 | 56.13 | 28.95 | 81.33 | 23.20 | 3.56 | 1.63 | 4.06 | 0.91 | 5.27 | 1.23 | | |
| 2MH | 49.50 | 20.62 | 69.05 | 29.29 | 98.01 | 25.30 | 3.44 | 1.64 | 4.23 | 0.93 | 5.81 | 1.37 | | |
| 1MH | 70.41 | 35.11 | 102.62 | 40.12 | 142.49 | 35.14 | 6.15 | 2.80 | 7.20 | 1.78 | 9.41 | 2.54 | | |
| MF | 40.31 | 20.40 | 45.87 | 21.69 | 62.61 | 25.60 | 3.68 | 1.14 | 4.90 | 2.64 | 7.88 | 2.49 | | |
| Н | 230.00 | 69.82 | 270.90 | 93.08 | 300.52 | 121.63 | 15.55 | 4.47 | 16.53 | 5.72 | 17.88 | 6.21 | | |
| Mean | 55.31 | | 69.87 | | 91.70 | | 4.28 | | 4.82 | | 6.10 | | | |

Table 4.1. Summary of the dynamic force (N) and contact area (cm^2) data for the normal weight (n=5), overweight (n=5) and obese (n=5) [Mean (SD) of three trials]
Statistically significant differences were found between the obese adults group and normal weight adults group in the following regions with mean values for F measure (P-values <0.001) at the H was 300.52 N, 1MH was 142.49 N, followed by the 1stT and 2MH with mean value 103.37 N and 98.01 N, respectively. In contrast, the mean values of the F variable in the obese group were approximately twice that under the 1MH with 142.49 N, 1stT with 103.37 N and 2MH with 98.01 N in contrast with the healthy group with values of 70.41, 45.29 and 49.50 N, at the same foot area. It was observed that the average values of the F measure under the whole foot and BMI for these three groups, shown in (**Figure 4.1**).



Figure 4.1. The relationship between force (N) measure and BMI (kg/m²)

Illustrative data for CA (cm²) measures in the adults who were obese, overweight and normal weight is seen in **Table 4.1**. In general, there were statistically significant differences (P-values <0.001) for the CA. The lower differences for the CA measure was located at the $3^{rd}-5^{th}T$ and $2^{nd}T$ between the normal weight, overweight and obese adults. The mean values for CA under $3^{rd}-5^{th}T$ were 0.74 cm², 1.17 cm² and 1.37 cm², respectively. Whereas the mean CA measure under the $2^{nd}T$ were 0.81 cm², 0.83 cm² and 0.84 cm², respectively. The higher CA value was found to be within the obese then overweight category compared to the normal category.

The overweight category presented greater CA than the normal weight adults of the same age, these differences were located beneath the H with 16.53 cm², 1MH with 7.20 cm², MF with 4.90 cm² followed by 2MH with 4.23 cm² and 3MH with 4.06 cm², regions. Whereas in the healthy group CA was 15.55, 6.15, 3.68, 3.44, and 3.56 cm². The obese group had higher foot CA values in comparison with the control group, under the H, 1MH and MF regions with the values 17.88, 9.41 and 7.88 cm², respectively. Thus, all values were higher within the obese group. **Figure 4.2** presents the relationships detected between the CA measurements within the whole foot regions within these three groups (according to their BMI), displaying a higher mean value beneath the obese category.



Figure 4.2. The relationship between the contact area (cm²) measure and BMI (kg/m²)

Post-hoc analysis using the Tukey's honestly significant difference test, confirmed that the obese group had a significantly higher mean dynamic force of 117.52 N in comparison with the overweight group of 92.30 N and the normal weight group of 75.82 N. Furthermore, the obese subjects displayed a notably higher mean contact area of 7.78 cm² than the normal subjects of 6.14 cm² and the overweight group of 6.51 cm².

4.1.2. Contact pressure and pressure-time integrals

A consistent increase in overall foot loading data for contact pressure (CP, kPa) and temporal characteristics of foot loading pressure–time integrals (PTI, kPa/s) for the groups are summarised in **Table 4.2**. There were notable variances between all three categories within CP and PTI measures which were observed at (P <0.001).

The greater CP values were measured under the 2MH, H, 3MH, 1stT and 1MH regions in the obese group compared with the normal group. These regions had mean CP values in the obese group of 118.96, 115.42, 111.27, 103.84 and 100.81 kPa, respectively. Whereas, these regions had mean CP values in the healthy group of 69.98, 90.20, 68.14, 55.38, and 61.53 kPa, respectively. The highest CP was under the MF in the obese group and the mean 85.58 kPa, which was approximately twice that under the same region in the normal group had a mean of 42.86 kPa. In addition, the obese group presented roughly double the value under the 1stT zone with a mean value of 103.84 kPa compared to the healthy category of 55.38 kPa.

There was a small dissimilarity amidst the overweight and normal weight groups for CP located at H with 98.98 kPa, 2MH with 84.67 kPa, 1stT with 83.82 kPa and 3MH with 80.68 kPa. In the normal group the CP values were 90.20, 69.98, 55.38 and 68.14 kPa respectively under the same foot area. On the other hand, the lowest CP value was observed within 3rd-5thT and 2ndT regions with the mean value for overweight of 37.65 and 42.11 kPa and healthy groups of 26.32 and 30.89 kPa. It was observed that the greater mean values of the CP measure under the whole foot and body mass index within these three groups, are shown in (**Figure 4.3**).

| | | | Contact p | ressure (kl | Pa) | | Pressure time integrals (KPa s) | | | | | |
|------------------------------------|-------|-------|-----------|-------------|--------|-------|---------------------------------|-------|--------|--------|--------|-------|
| | Hea | althy | Overv | weight | Ob | ese | Hea | lthy | Overv | weight | Obe | ese |
| Foot region | Mean | SD | Mean | SD | Mean | SD | Mean | SD | Mean | SD | Mean | SD |
| 3 rd -5 th T | 26.32 | 11.49 | 37.65 | 15.75 | 34.01 | 28.50 | 28.34 | 15.92 | 37.04 | 11.42 | 58.76 | 14.96 |
| 2 nd T | 30.89 | 14.32 | 42.11 | 15.22 | 52.67 | 10.41 | 34.58 | 12.20 | 44.62 | 10.97 | 82.59 | 19.65 |
| 1 st T | 55.38 | 15.32 | 83.82 | 19.30 | 103.84 | 56.92 | 57.31 | 23.96 | 90.81 | 26.60 | 116.63 | 17.38 |
| 5MH | 52.80 | 12.86 | 62.97 | 24.58 | 75.91 | 43.70 | 61.14 | 25.33 | 65.97 | 18.04 | 74.10 | 38.73 |
| 4MH | 63.86 | 17.76 | 73.77 | 32.13 | 97.32 | 45.81 | 70.77 | 23.60 | 78.33 | 27.72 | 96.25 | 42.16 |
| ЗМН | 68.14 | 16.52 | 80.68 | 27.11 | 111.27 | 34.20 | 77.95 | 30.41 | 83.80 | 21.38 | 100.02 | 41.21 |
| 2MH | 69.98 | 17.54 | 84.67 | 25.44 | 118.96 | 31.34 | 84.57 | 27.32 | 93.64 | 20.27 | 109.65 | 53.41 |
| 1MH | 61.53 | 19.52 | 70.90 | 26.04 | 100.81 | 27.33 | 70.82 | 28.14 | 86.56 | 24.56 | 101.12 | 40.76 |
| MF | 42.86 | 16.44 | 50.93 | 15.20 | 85.58 | 38.21 | 35.21 | 14.65 | 49.94 | 11.80 | 58.21 | 14.71 |
| Н | 90.20 | 22.75 | 98.98 | 25.95 | 115.42 | 61.73 | 93.54 | 28.91 | 116.00 | 71.96 | 123.24 | 60.36 |
| Mean | 56.20 | | 68.65 | | 89.58 | | 61.42 | | 74.67 | | 92.06 | |

Table 4.2. Summary of the contact pressure (kPa) and pressure-time integral (kPa/s) data for the healthy weight (n=5), overweight (n=5) and obese (n=5) [Mean (SD) of three trials]



Figure 4.3. The relationship between contact pressure (KPa) measure and BMI (kg/m^2)

Elevated PTI were attributed to the smallest level of probability the Tukey's honestly significant difference test, for the PTI for these groups. The normal weight group was with 61.42 kPa/s, followed by the overweight with 74.67 kPa/s and the obese group with 92.06 kPa/s. The results show higher PTI value underneath the H, 1^{st} T, 2MH and 3MH in the obese group (P < 0.001) compared to the healthy category. The obese category had mean values under the H, 1^{st} T, 2MH and 3MH regions with 123.24, 116.63, 109.65, 100.02 kPa/s, respectively.

The overweight group PTI also was higher compared with the healthy group beneath the H, 2MH, 1stT and 1MH regions, with mean values of 116.00, 93.64, and 90.81, 86.56 kPa/s. While, the healthy group had mean values of 93.54, 84.57, 57.31 and 70.82 kPa/s, within these regions respectively. On the other hand, the lowest region within this PTI measure was absorbed under $3^{rd}-5^{th}T$ with 58.76 and 37.04 kPa/s for both groups (P < 0.001) compared with the normal weight group with 28.34 kPa/s, is shown in **Table 4.2**. It was observed that the a higher mean values of the PTI measure under the whole foot and body mass index within these three groups, as shown in (**Figure 4.4**).



Figure 4.4. The relationship between pressure time integral (kPa/s) measure and BMI (kg/m²)

4.1.3. Peak pressure

The peak pressure (PP) was generally higher in the obese group compared with the other two groups. The highest PP obtained in the obese group under the H with 204.38 kPa followed by the 1stT with 189.00 kPa. Descriptive data for the maximum and minimum PP in healthy, overweight and obese groups are represented in (**Figure 4.5**) while the obese category was likened to both the overweight and healthy groups. The PP value had a maximum variance within the obese group at both regions of H with 204.38 kPa and the 1stT with 189.99 kPa, which approximately double those of the healthy group with 101.71 and 88.37 kPa respectively. The lowest value, in contrast, was under the 3rd-5thT and 2ndT, which was the lowest change between these three groups (obese, overweight and healthy) under these regions. The mean PP values for these region were 55.36, 42.62 and 32.20 kPa; and 72.13, 56.73 and 38.89 kPa, respectively within the same area. It was found to be greater than both groups (overweight and healthy) as displayed in (**Figure 4.5**), which shows the comparisons between the healthy, overweight and obese groups.



Figure 4.5. Bar chart for healthy weight, overweight and obese, representing the variation in the ten foot regions

4.2 Discussion

The experiment was designed to examine the effect of differences in body mass in adults aged between 25 and 46 years in the three groups (overweight, obese and healthy weight), on plantar pressure distribution characteristics by calculating clinical foot measurements (F, CA, CP, PP and PTI) in ten different foot regions during gait. The findings identified that those who were overweight displayed marked differences in foot loading when compared with adults of healthy weight. Overweight adults generated significantly greater F and CA MF and second to fifth MH regions of the foot, a trend that was similar for the obese adults. Overweight adults also generated an increased CP and PTI under the MF and 2 to 5MHs. A similar trend was found for the obese adults with significant differences also found at the H in comparison with normal weight adults.

Plantar foot pressure parameters such as PP and CP presented statistically remarkable rises with increasing weight. This was noticed with the rising volunteer BMI class. The results provide important objective information regarding the functional limitations specific to foot mechanics in dynamic (walking) situations. Observing the marked dissimilarity in plantar pressures as a function of increased adiposity (BMI), it is interesting to speculate on the structural consequences of repetitive loading on the feet and other parts of the lower extremity.

The physiological manifestations of loading may be reflected in self-reported pain, soreness or discomfort in the lower extremity as a consequence of the increased pressures and forces acting on the musculoskeletal structure of the human foot. For instance, plantar heel pain is commonly characterised by pain and tenderness on the calcaneal tuberosity at the point of attachment of the plantar fascia (Hill & Cutting, 1989). Although the specific aetiology of plantar heel pain is unknown, a number of studies (Cousins et al., 2013; Dowling et al., 2001; Dowling et al., 2004; Mickle et al., 2006; Yan et al., 2013; Hill & Cutting, 1989) have reported that increased body weight is an associated factor.

There is a suggestion that musculoskeletal pain in the lower extremity may cause people to alter their gait pattern in an attempt to avoid or minimise discomfort (Messier et al., 1995). Additionally, McGoey et al. (1990) stated that chronic musculoskeletal pain is common in morbidly obese individuals and that weight loss can lead to significant relief in self-reported pain. Other researchers have reported a decreased sensitivity in the feet of diabetics, skin breakdown and a predisposition to foot ulceration associated with neuropathy in individuals with higher static pressures under the feet (Vela et al., 1998; Cavanagh et al., 1985). Ease of walking, as determined via discomfort experienced in the feet, may be a major limiting factor in the predisposition of the obese (including diabetics who are overweight and obese) to participate in habitual physical activity such as walking. Nyska et al. (1997) pointed out that the foot adapts itself to excessive weight bearing by maintaining the medial longitudinal arc. Excessive weight bearing due to increased body weight in subjects with obesity is believed to cause structural foot dysfunction, such as the collapse of the longitudinal arc, which especially leads to an increased middle foot contact area (Riddiford-Harland et al., 2000).

4.2.1. Force and contact area

The F values of the whole foot were the higher mean value in obese and overweight adults, while the lowest F mean value with healthy weight adults. Changes in PF data was largely detected owing to body mass. However, the findings at the H and

forefoot (specifically 2MH, 3MH and 1stT) for adults who are obese and overweight, demonstrated increased loading at these sites. These results are consistent with previous research (Dowling et al., 2001; Yan et al., 2013). These findings have identified elevated levels of loading, in children who are obese at the plantar H, MF and 2MH to 5MH when compared with non-obese children.

The outcomes for F, found by Mickle et al. (2006), correspond with those of this study. They found that H, MF and forefoot had the higher F value. The objective of the study undertaken by Mickle et al. (2006) was to determine the effects of overweight and obesity on plantar pressures generated by pre-school children during gait. They emphasised that higher stress made the feet of overweight/obese children vulnerable to bone fatigue and soft tissue damage due to higher F over larger CA.

A noteworthy main effect of obesity on dynamic CA was also found as shown in (Figure 4.2). The higher foot CA in overweight and obese adults was compared with their normal-weight counterparts. The higher CA was measured in both groups under the H, MF, 1MH, 2MH, and 3MH anatomical regions. In this study, the higher CA within MF was confirmed by previous research (e.g., Hills et al., 2001; Prabhu et al., 2001; Periyasamy et al., 2012). These studies stressed that the foot pressure distribution parameter power ratio value is greater for diabetic foot than the normal foot (Prabhu et al., 2001). This power ratio value decreases as the BMI increases because the total contact area increases due to exposure to higher loads and does not depend on larger feet. Hence this study found significant variations in power ratio value in MF and the plantar ground contact area of overweight subjects compared with healthy subjects while standing, because plantar pressure increase begins from the MF in low grade obese subjects (i.e. overweight) due to an increase CA in MF.

Mickle et al. (2006) outcomes for CA correspond with those of this study. They found H, MF and forefoot CA were higher in overweight and obese compare to healthy children. The objective of this study undertaken by Mickle et al., (2006), determines the effects of overweight and obesity on plantar pressures generated by pre-school children through walk. They underlined that increased CA was not found to be sufficient to compensate for the high F generated during walking, thus causing higher plantar pressure with increased stress (Mickle et al., 2006).

Considering that the obese group had the greatest CA and that the greatest difference between the groups obese and non-obese in terms of the current results disagree with those of Filippin et al., (2007). The objective of the research undertaken by Filippin et al., (2007) was to determine if there were differences in static and dynamic plantar pressure distribution between obese and eutrophic children and specifically, where the pressure is located on the foot and in what proportion it correlates to body mass. It is acceptable to assume that the obese children displayed flattening of the medial longitudinal arch. In contrast, it is not yet clear why this flattening and consequently, the rise in CA in the MF region, occurs.

4.2.2. Contact pressure and pressure-time integrals

As described in the results section, overweight and obese adults have a higher value of CP at the H, 2MH, 3MH, 1MH, 1stT and MF regions. These increased the CP consequences are compatible with Hills et al. (2001). Hills et al. (2001) investigated plantar pressure differences between obese and non-obese adults during standing and walking protocols using a pressure distribution platform. They highlighted the cause for subjects to have a higher body weight.

In this study, the CP was statistically significantly higher in the H, 2MH, 3MH, 1MH, 1stT and MF regions. This finding was dissimilar to the finding of Yoon et al. (2016), who found that no notable change was observed in the PP in forefoot and H areas. Yoon et al. (2016) determined foot stability according to the results of plantar pressure and spontaneity balance in the normal group and in the obesity, group based on BMI, using a pressure distribution platform. They accentuated the cause for participants to a reduction in the spontaneity balance index in the H region of both the healthy and obese BMI categories. This is thought to be a mechanism of compensating for the decreased hindfoot balance in the forefoot area, with forefoot pressure increasing to compensate for balance.

PTI also referred to as impulse, was found to be statistically higher beneath the H, 1stT, 2ndT, 3MH, 1MH and MF. PTI measured value, gives details regarding weight distribution through time. The PTI outcomes for Mickle et al. (2006) correspond with those of this study. Mickle et al. (2006) found that the MF displayed significantly higher values. The objective of the study undertaken by Mickle et al. (2006), determine the effects of overweight and obesity on plantar pressures

generated by pre-school children during gait. They confirm that the higher stress and, in turn, vulnerability to bone fatigue and soft tissue damage to higher forces over larger contact areas, occurs in the overweight/obese children

In this study, the PTI was statistically significantly higher in 1MH to 3MH. This finding is similar to the finding of Putti et al. (2010). Putti et al. (2010) investigated the in-shoe pressure differences in Caucasians and Indians. This variable is thought to be important in the pathogenesis of skin lesions (Fuller, 1996).

4.2.3. Peak pressure

As for peak pressure (PP), the obese adults generally displayed greater values than the healthy group during gait, however, the H, 1stT and metatarsal head regions in particular (2MH, 3MH and 1MH) suffered greater pressure, and the 3rd-5thT area had the lowest peak pressure, as presented in (**Figure 4.5**). The results are in accordance with (Hills et al., 2001; Dowling et al., 2001; Filippin et al., 2007). The objective of the study by Filippin et al. (2007) was to determine if there are differences in static and dynamic plantar pressure distribution between obese and eutrophic children and, specifically, where the pressure is located on the foot and in what proportion it correlates to body mass. Butterworth et al. (2015) also found weight to be a significant independent predictor of PP under the H, MF and forefoot. The purpose of the study by Vela et al., (1998) was to determine if increased weight contributes to increased mean PP foot pressures when foot function, deformity and structure are controlled.

The PP outcomes from Mickle et al. (2006) correspond with those of this study. Mickle et al. (2006) found that the MF displayed a significantly higher value. The objective of this study undertaken by Mickle et al., (2006) determines the effects of overweight and obesity on plantar pressures generated by pre-school children during gait. They confirm that the higher stress and, in turn, vulnerable to bone fatigue and soft tissue damage due to higher F over larger CA, the overweight/obese children.

In the present study, it was established that the PP mean values were statistically notably higher under the H, 1stT and metatarsal head regions in particular (2MH, 3MH and 1MH). These findings were in absolute agreement with those of Yan et al. (2013), who found that no significant change was observed in the forefoot

and hind-foot PP. Yan et al. (2013) study was designed to examine the effects of obesity on dynamic plantar pressure distribution during walking for prepubescent children, using a plantar pressure plate system. They emphasised that higher plantar pressure in obese children will increase loading on the developing foot and may result in foot discomfort and possibly deformity, which may increase the risk of injury.

In this study, the highest PP was found under the H region. These findings are supported by studies investigating foot function in children (Bosch et al., 2007; Müller et al., 2012; Phethean et al., 2014; Bosch et al., 2009) and adults (Putti et al., 2008; Bosch et al., 2009; Scott et al., 2007).

4.3 Chapter summary

The present study has demonstrated differences between plantar foot loading characteristics of obese, overweight and healthy weight adults, aged 25–46 years, during level walking. Obese and overweight adults showed increases in heel and forefoot foot regions. The highest foot pressure measure increases in the obese were found under the heel of the foot and the metatarsal heads. Compared with the non-obese groups, there were increases in pressure under the H, MF and the middle of the forefoot (2MH and 3MH). This difference is the result of reduced strength of the ligaments of the foot in obese individuals. These findings have implications for pain and discomfort in the lower extremity in the obese, the choice of footwear and predisposition to participate in activities of daily living such as walking. Further study in this area is warranted.

The next chapter aims to answer objective two, which involves simulation body weight to determine plantar pressure data variables during the loading phase of the gait.

5. RESULT OF EXPERIMENT 2

5.1. Introduction

This chapter focuses on covering the main results of the simulated bodyweight experiment. The first section of the chapter considers experiment two's objective, which was to estimate the effect of simulated body mass on foot plantar measurers force (F), contact area (CA), contact pressure (CP), pressure time integral (PTI), and peak pressure (PP). The second section of the chapter measures the repeatability F-Scan system for the result through foot plantar measures (F, CA, CP, PTI, and PP). The chapter discussion is presented in the next section and the last section focuses on the chapter summary.

The experiment was conducted to investigate the influence of simulated body weight on the plantar pressure parameters during gait. Both feet (62 feet) in a total of thirty-one participants were examined. The characteristics of the participants were as (mean \pm SD) follows: age 37.55 \pm 6.92 years, body weight 89.82 \pm 17.92 kg, height 1.74 \pm 0.08 cm, and BMI 46.00 \pm 20.1 kg/m², respectively. Statistically significant correlation coefficients (P < 0.001) have been found between simulated body weight (SBW) (an increase in load) with the main effects on all foot variables (F, CA, CP, PTI and PP). Parameters were statistically observed to increase their values gradually under SBW, which may present problems for the longer-term compared with the normal group (NBW, 0 kg). This effect was consistent at any given simulated body weight as displayed in (**Figure 5.1 A, B, C, D, E**).



(A) Simulated body weight









Figure 5.1. The average effect of simulated body weight on foot plantar pressure (A-force, B- contact area, C- contact pressure, D- pressure time integral, and E- peak pressure)

5.1.1. Force

Force (F) measure was studied with SBW and resulted in a notable rise in mean F variables between 30, 20, and 10 kg load simulations compared to the NBW (0 kg) group in all foot regions. The F measure values with increased body weight are presented in **Table 5.1**. The F variable displayed a mean increase under the H, 1MH, 2MH, 3MH and MF regions, showing significant agreements (P < 0.001) with SBW.

In the 30, 20, 10, and 0 kg simulations, mean F value was significantly greater in the H value with 370.81, 354.89, 321.53 and 307.42 N, followed by 1MH zone with 142.69, 133.35, 117.44 and 105.74 N; 2MH region with 113.68, 104.39, 95.52 and 80.56 N, in the MF region with mean F values at 78.69, 66.75, 60.44 and 51.45N, respectively.

In the 30 kg SBW load, F was significantly larger in all regions of the foot compared with other load conditions, but there was a small change in all SBWs (10, 20, 30 kg) under the 3rd-5thT with mean value 17.51, 17.30 and 17.08 N, 2ndT with mean value 12.75, 10.36 and 9.98 N, and 1stT with mean value 66.96, 66.81 and 62.28 N regions, compared with NBW group (0 kg) which had a mean value of 15.46, 9.97 and 58.38 N.

Individual means of all the three repeated trials for each foot were calculated, and these values were used to calculate the between-weights coefficient of variance (CV) for all parameters, since this was shown to be accepted methods of measuring reliability between testing sessions. Measurements with a higher CV were recorded mainly under the toes. CV values for the regional F pressures ranged between SBW and NBW (0 kg), as shown in **Table 5.1**.

| Force (N) | | | | | | | | | | | | | | |
|------------------------------------|--------|---------|------|--------|--------|------|--------|--------|------|--------|--------|------|--|--|
| Foot region | | Zero kg | | | 10 kg | | | 20 kg | | | 30 kg | | | |
| root region | Mean | SD | %CV | Mean | SD | %CV | Mean | SD | %CV | Mean | SD | %CV | | |
| 3 rd -5 th T | 15.46 | 5.83 | 1.29 | 17.08 | 7.85 | 1.12 | 17.30 | 7.30 | 0.92 | 17.51 | 7.39 | 1.03 | | |
| 2 nd T | 9.97 | 3.53 | 1.18 | 9.98 | 3.96 | 0.86 | 10.36 | 4.71 | 0.90 | 12.75 | 5.67 | 0.97 | | |
| 1 st T | 58.38 | 12.45 | 0.79 | 62.28 | 11.20 | 1.03 | 66.81 | 18.33 | 0.79 | 66.96 | 18.14 | 0.80 | | |
| 5MH | 35.13 | 18.36 | 0.59 | 39.62 | 19.02 | 0.73 | 42.37 | 23.61 | 0.52 | 46.11 | 24.78 | 0.45 | | |
| 4MH | 55.30 | 20.57 | 0.47 | 62.55 | 19.99 | 0.50 | 70.64 | 20.28 | 0.36 | 75.59 | 29.83 | 0.33 | | |
| 3MH | 71.78 | 34.37 | 0.41 | 83.07 | 43.53 | 0.43 | 93.19 | 36.92 | 0.34 | 101.77 | 37.92 | 0.31 | | |
| 2MH | 80.56 | 39.32 | 0.43 | 95.52 | 42.36 | 0.37 | 104.39 | 44.25 | 0.37 | 113.68 | 44.58 | 0.33 | | |
| 1MH | 105.74 | 56.21 | 0.97 | 117.44 | 56.37 | 0.41 | 133.35 | 68.03 | 0.45 | 142.69 | 70.01 | 0.41 | | |
| MF | 51.45 | 26.57 | 0.84 | 60.44 | 28.25 | 0.87 | 66.75 | 36.23 | 0.78 | 78.69 | 39.14 | 0.71 | | |
| H | 307.42 | 133.63 | 0.38 | 321.53 | 134.80 | 0.40 | 354.89 | 142.87 | 0.35 | 370.81 | 140.40 | 0.33 | | |

5.1.2. Contact area

The CA measure was also investigated with SBW. There was a significant difference in (P < 0.001) in CA between SBW and NBW (0 kg), as shown in **Table 5.2.** The three main regions that made the largest CA were the insole system starting with the H region with 19.11, 18.19, 17.29 and 16.26 cm², followed by 1MH with 8.97, 8.15, 7.74, 6.02 cm², and the MF with measurements of 8.87, 7.96, 6.28, 4.82 cm².

However, in the simulated body weights (30, 20 and 10 kg), the CA demonstrated a slight increase within the $2^{nd}T$ with a mean value were 0.99, 0.81, and 0.70 cm², compared to the unweighted control group which had a CA value of 0.67 cm² in the same region. The CA value located at the $3^{rd}-5^{th}T$ in SBW (30, 20, and 10 kg) were modest changes compared to the NBW control group. The values under the $3^{rd}-5^{th}T$ area were 1.57, 1.55, 1.46 cm² while the CA value in the NBW group was 1.42 cm².

| Area (cm ²) | | | | | | | | | | | | | | |
|------------------------------------|-------|---------|------|-------|-------|------|-------|-------|------|-------|-------|------|--|--|
| Fact marian | | Zero kg | | | 10 kg | | | 20 kg | | | 30 kg | | | |
| root region | Mean | SD | %CV | Mean | SD | %CV | Mean | SD | %CV | Mean | SD | %CV | | |
| 3 rd -5 th T | 1.42 | 2.69 | 1.29 | 1.46 | 1.03 | 0.61 | 1.55 | 1.06 | 0.66 | 1.57 | 1.24 | 0.71 | | |
| 2 nd T | 0.67 | 0.6 | 1.18 | 0.70 | 1.35 | 1.12 | 0.81 | 0.51 | 0.60 | 0.99 | 0.55 | 0.62 | | |
| 1 st T | 3.11 | 1.73 | 0.79 | 3.30 | 2.1 | 0.49 | 3.54 | 1.54 | 0.43 | 3.86 | 1.59 | 0.39 | | |
| 5MH | 3.00 | 1.73 | 0.59 | 3.16 | 1.66 | 0.45 | 3.44 | 1.76 | 0.46 | 4.96 | 1.96 | 0.44 | | |
| 4MH | 3.98 | 1.72 | 0.47 | 4.24 | 1.64 | 0.32 | 4.58 | 1.58 | 0.30 | 4.79 | 1.45 | 0.25 | | |
| 3MH | 4.42 | 1.53 | 0.41 | 4.86 | 1.56 | 0.28 | 5.18 | 1.45 | 0.24 | 5.40 | 1.33 | 0.22 | | |
| 2MH | 4.60 | 1.62 | 0.43 | 5.20 | 1.51 | 0.26 | 5.26 | 1.48 | 0.25 | 5.88 | 1.54 | 0.24 | | |
| 1MH | 6.02 | 3.18 | 0.97 | 7.74 | 2.98 | 0.35 | 8.15 | 3.61 | 0.36 | 8.97 | 3.29 | 0.34 | | |
| MF | 4.82 | 5.75 | 0.84 | 6.28 | 5.77 | 0.88 | 7.96 | 5.99 | 0.81 | 8.87 | 5.63 | 0.69 | | |
| H | 16.26 | 6.8 | 0.38 | 17.29 | 5.69 | 0.29 | 18.19 | 5.7 | 0.28 | 19.11 | 7.15 | 0.31 | | |

Table 5.2. Contact area measure with increased body weight for ten regions of the foot

5.1.3. Contact pressure

The highest contact pressure (CP) was found under the 2MH, H, and 3MH regions with SBW states, which are displayed in **Table 5.3**. The difference between the averages of the four states was significant (P < 0.001). The mean CP values under 30, 20 and 10 kg weight were beneath the 2MH with 149.11, 135.55 and 116.99 kPa, while the NBW (0 kg) had the mean value for same the zone with 110.48 kPa. In addition, the H area was presented as statistically significant in all SBW loads and was greatest with 30 kg followed by 20, and 10 kg weight compared with the NSBW (0 kg). The values for both loads were 146.24, 139.10, 122.07 kPa and 111.15 kPa. Furthermore, the CP value with SBW at 30, 20 and 10 kg compared with NBW group (0 kg) within 3MH region showed 130.22, 122.70, 111.86 kPa and 103.85 kPa.

However, in relation to the smaller changes in CP value with the increased SBW in mean contact pressure under the $3^{rd}-5^{th}T$ region, the values for this region with all control and SBW were 32.13, 40.23, 43.35, 49.70 kPa, respectively. The CP value was approximately double that under the MF within 30, 20 and 10 kg loads with a mean value 89.16, 60.78, 55.48 kPa compared with the NBW group, which had a mean value of 43.25 kPa.

| Contact pressure (kPa) | | | | | | | | | | | | | | |
|------------------------------------|--------|---------|------|--------|-------|------|--------|-------|------|--------|-------|------|--|--|
| | | Zero kg | | | 10 kg | | | 20 kg | | | 30 kg | | | |
| Foot region | Mean | SD | %CV | Mean | SD | %CV | Mean | SD | %CV | Mean | SD | %CV | | |
| 3 rd -5 th T | 32.13 | 60.4 | 0.89 | 40.23 | 29.74 | 0.50 | 43.35 | 30.79 | 0.57 | 49.70 | 28.51 | 0.57 | | |
| 2 nd T | 37.87 | 38.61 | 0.72 | 39.96 | 31.71 | 0.68 | 41.53 | 29.8 | 0.69 | 46.05 | 39.42 | 0.78 | | |
| 1 st T | 78.54 | 72.29 | 0.73 | 80.97 | 61.47 | 0.60 | 82.97 | 58.88 | 0.64 | 88.02 | 60.87 | 0.65 | | |
| 5MH | 54.78 | 32.12 | 0.42 | 69.30 | 30.29 | 0.37 | 74.01 | 29.92 | 0.33 | 81.46 | 37.79 | 0.37 | | |
| 4MH | 89.58 | 37.14 | 0.35 | 94.96 | 31.8 | 0.29 | 103.64 | 31.96 | 0.26 | 110.51 | 34.71 | 0.27 | | |
| 3MH | 103.85 | 44.3 | 0.35 | 111.86 | 38.55 | 0.30 | 122.70 | 40.46 | 0.28 | 130.22 | 39.02 | 0.26 | | |
| 2MH | 110.48 | 55.87 | 0.39 | 116.99 | 41.53 | 0.32 | 135.55 | 90.38 | 0.53 | 149.11 | 58.27 | 0.34 | | |
| 1MH | 73.42 | 35.87 | 0.37 | 88.09 | 32.79 | 0.33 | 101.23 | 83.89 | 0.60 | 111.96 | 35.03 | 0.30 | | |
| MF | 43.25 | 36.17 | 0.51 | 55.48 | 32.22 | 0.47 | 60.78 | 44.9 | 0.51 | 89.16 | 46.21 | 0.45 | | |
| Н | 111.15 | 78.14 | 0.71 | 122.07 | 68.92 | 0.40 | 139.10 | 83.01 | 0.44 | 146.24 | 64.95 | 0.37 | | |

 Table 5.3. Contact pressure measure with increased body weight for ten regions of the foot

5.1.4. Pressure time integral measurement

The pressure time integral (PTI, kPa/s) measure was also investigated with SBW. There was a significant difference in (P < 0.001) in PTI between SBW and NBW groups. There were notably increases between the 30 kg, 20 kg and 10 kg load compared with the NBW group (0 kg) status within the H, 2MH, 3MH, 4MH, 1MH, 1^{st} T, 5MH and MF which are shown in **Table 5.4**.

The three main regions that made the largest PTI were the insole system starting with the H region with 169.58, 151.92, 140.71 and 120.84 kPa/s; 2MH region with 154.04, 142.94, 126.41 and 119.79 kPa/s; and 3MH area with 146.15, 133.78, 123.37 and 118.18 kPa/s; 4MH region with 124.44, 114.1, 103.35 and 92.46 kPa/s, followed by the 1MH region 121.71, 108.17, 96.31 and 83.25 kPa/s; and the 1stT region with 116.8, 97.01, 96.05 and 84.43 kPa/s, respectively.

However, in the SBW (30, 20 and 10 kg), the PTI demonstrated a slight increase within the 2ndT with a mean value were 62.26, 51.58, and 48.66 kPa/s, compared to the NBW control group which had a PTI value of 41.94 kPa/s in the same region. The PTI value located at the 3rd-5thT in simulated body weight groups (30, 20, and 10 kg) was modest changes than the unweighted control group. And the value under the 3rd-5thT area were 73.3, 61.52 and 59.46 kPa/s while the PTI value in NBW group was 43.68 kPa/s.

| Pressure time integral (kPa/s) | | | | | | | | | | | | | | |
|------------------------------------|--------|---------|------|--------|--------|------|--------|--------|------|--------|--------|------|--|--|
| | | Zero kg | | | 10 kg | | | 20 kg | | | 30 kg | | | |
| root region | Mean | SD | %CV | Mean | SD | %CV | Mean | SD | %CV | Mean | SD | %CV | | |
| 3 rd -5 th T | 43.68 | 99.61 | 1.24 | 59.46 | 45.63 | 0.71 | 61.52 | 48.26 | 0.79 | 73.3 | 60.6 | 0.89 | | |
| 2 nd T | 41.94 | 76.3 | 1.07 | 48.66 | 47.54 | 0.86 | 51.58 | 46.76 | 0.89 | 62.26 | 52.36 | 0.88 | | |
| 1 st T | 84.43 | 140.61 | 1.08 | 96.05 | 83.26 | 0.73 | 97.01 | 83.86 | 0.79 | 116.8 | 86.18 | 0.78 | | |
| 5MH | 69.28 | 45.05 | 0.53 | 75.61 | 31.13 | 0.34 | 87.18 | 160.72 | 1.23 | 99.38 | 44.16 | 0.40 | | |
| 4MH | 92.46 | 59.52 | 0.52 | 103.35 | 37.62 | 0.32 | 114.1 | 41.09 | 0.30 | 124.44 | 49.13 | 0.35 | | |
| 3MH | 118.18 | 65.91 | 0.50 | 123.37 | 54.24 | 0.36 | 133.78 | 45.92 | 0.30 | 146.15 | 54.45 | 0.34 | | |
| 2MH | 119.79 | 65.75 | 0.47 | 126.41 | 45.67 | 0.33 | 142.94 | 53.98 | 0.35 | 154.04 | 61.62 | 0.38 | | |
| 1MH | 83.25 | 51.18 | 0.46 | 96.31 | 39.7 | 0.38 | 108.17 | 48.88 | 0.41 | 121.71 | 42.59 | 0.35 | | |
| MF | 50.06 | 40.51 | 0.56 | 67.43 | 36.58 | 0.47 | 71.39 | 41.02 | 0.48 | 89.28 | 64.69 | 0.59 | | |
| Н | 120.84 | 148.94 | 0.78 | 140.71 | 103.59 | 0.58 | 151.92 | 115.01 | 0.58 | 169.58 | 116.37 | 0.63 | | |

 Table 5.4. Pressure time integral measure with increased body weight for ten regions of the foot

5.1.5. Peak pressure

Observed values for the PP depicted maximum values in this test, with SBW; 30, 20 and 10 kg groups compared with NBW group beneath fifth regions of the ten regions. These regions were the H, 2MH, 3MH, 1MH and 1st T. Furthermore, the main values for these regions were (256.49, 207.68, 192.68, 188.35 and 167.49 kPa); (242.44, 195.42, 180.18, 167.36 and 152.66 kPa); followed by (227.14, 173.09, 160.60, 148.95 and 145.07 kPa), and (211.57, 147.26, 142.28, 136.52 and 132.96 kPa), respectively. The PP showed a fluctuating higher value in specific simulations compared with control simulations within the 3rd-5thT and 2ndT, which were 7.29, 9.39, 42.64 and 36.91 kPa and 63.44, 54.83, 53.08 and 31.91 kPa, respectively. There were statistically significant increases between the 20 and 30 kg weights and NBW group (0 kg) within the H, 2MH, 3MH, 1MH, 1stT, and 4MH as displayed in (**Figure 5.2**).



Figure 5.2. Comparison of peak pressure parameter among simulated body weight under four weighted simulations for ten regions of the foot. The next section focuses on the coefficients of repeatability for the insole system

5.2. Repeatability of the F-Scan system

The results of the ICC for analysis shows a generally good level of reliability, the quality of which is dependent on the region of the foot and the variable investigated with SBW loads. This research displayed that high between-day reliability could be obtained between testing days in healthy, symptom-free participants, using SPSS software to calculate ICC values by applying model two. Lower reliability was found in the typically less-loaded areas, such as the 3rd-5thT. For this dissertation, Model 2 is used to calculate the reliability of gait variables since it is a repeated measures method. In addition, the sample size requirements are computed based on the ICC values. Detailed calculating via a two-way mixed effects model has been applied and is highlighted in **chapter 3, section 3.5.3**.

It must be realised however, that the results apply only for the specific instrument that was used here and is not generally transferable to other available pressure distribution measurement systems. These may be established on various sensor technologies and may not achieve comparable reliability. In addition, the F-Scan masking method may limit our ability to generalise the conclusions from this study to a wider population. The F-Scan mask method, selected for its strong reputation as a valid method of dividing the foot into anatomical regions of interest, still makes assumptions regarding the boundaries of particular regions. In addition, areas such as the MF may become under-represented by such masking algorithms therefore, when investigating afflictions such as Pes planovalgus, an investigator may choose a mask which is more appropriate for this examination. However, since the examination being attempted in this study was of a non-specific nature, the more generic mask was selected.

The study showed that all measures are repeatable. From the outcome data highlighted in this study (F, CA, CP, PTI and PP), it can be seen that the findings support the application of an in-shoe pressure capturing system as a dependable clinical investigation modality. These applications include evaluating the pressure distribution under the feet in various common foot pathologies, diabetic foot disorders, pre-operative assessment of foot deformity, objectively assessing the success of corrective foot surgery and assessment of rheumatoid arthritis.

5.2.1. Force variable

Positive outcomes were attained at the baseline, as shown in **Table 5.5**. All anatomical regions of the foot was researched. It allowed the potential to notice 'good to excellent' ICC values for the three-positions: left, right and both feet. ICC values for both feet column is obtained by calculating the average left foot and right foot as presented in **Table 5.5**. As indicated in the calculation, the reliability ICC value ranges between (0.75 - 0.98) foot area locations, respectively which scored 'good and excellent' ICC values. Therefore, the results display an acceptable repeatability trend when the F variable was measured at the baseline with unweighted and additional weights. Detailed calculating two-way mixed effects model which has been applied, is shown in **chapter 3**, **section 3.5.3**.

| Force (N) | | | | | | | | | | | | | | |
|------------------------------------|-----------------|---------|------|------|-------|------|------|-------|------|------|-------|------|--|--|
| Foot region | | Zero kg | | | 10 kg | | | 20 kg | | | 30 kg | | | |
| root region | Left Right Both | | | Left | Right | Both | Left | Right | Both | Left | Right | Both | | |
| 3 rd -5 th T | 0.97 | 0.80 | 0.89 | 0.92 | 0.89 | 0.90 | 0.86 | 0.82 | 0.84 | 0.93 | 0.96 | 0.94 | | |
| 2 nd T | 0.93 | 0.76 | 0.85 | 0.93 | 0.77 | 0.85 | 0.88 | 0.86 | 0.87 | 0.93 | 0.93 | 0.93 | | |
| 1 st T | 0.90 | 0.86 | 0.88 | 0.88 | 0.75 | 0.82 | 0.94 | 0.96 | 0.95 | 0.98 | 0.97 | 0.98 | | |
| 5MH | 0.78 | 0.88 | 0.83 | 0.81 | 0.77 | 0.79 | 0.84 | 0.94 | 0.89 | 0.87 | 0.87 | 0.87 | | |
| 4MH | 0.80 | 0.87 | 0.84 | 0.91 | 0.76 | 0.84 | 0.85 | 0.93 | 0.89 | 0.89 | 0.84 | 0.87 | | |
| 3MH | 0.91 | 0.93 | 0.92 | 0.93 | 0.76 | 0.85 | 0.86 | 0.92 | 0.89 | 0.76 | 0.90 | 0.83 | | |
| 2MH | 0.95 | 0.96 | 0.95 | 0.94 | 0.78 | 0.86 | 0.82 | 0.94 | 0.88 | 0.94 | 0.95 | 0.95 | | |
| 1MH | 0.86 | 0.91 | 0.88 | 0.88 | 0.92 | 0.90 | 0.80 | 0.92 | 0.86 | 0.92 | 0.91 | 0.91 | | |
| MF | 0.89 | 0.89 | 0.89 | 0.92 | 0.84 | 0.88 | 0.96 | 0.94 | 0.95 | 0.95 | 0.94 | 0.94 | | |
| Н | 0.96 | 0.92 | 0.94 | 0.89 | 0.90 | 0.89 | 0.93 | 0.96 | 0.93 | 0.92 | 0.95 | 0.93 | | |

 Table 5.5. Intra-class correlation coefficients result for F-Scan system regarding force variable

5.2.2. Contact area variable

As shown below in **Table 5.6**, the data analysis of the contact area (CA) measure, obtained positive results at the baseline. In all ten anatomical zones considered, it was observed to have 'good and excellent' ICC values for the states: left, right and both feet. ICC value ranges between (0.78 - 0.99) foot region locations, respectively which scored 'good and excellent' ICC values. Detailed calculating two-way mixed effects model which has been applied, is shown in **chapter 3, section 3.5.3**.

| Area (cm ²) | | | | | | | | | | | | | | |
|------------------------------------|-----------------|---------|------|------|-------|------|------|-------|------|------|-------|------|--|--|
| East resise | | Zero kg | | | 10 kg | | | 20 kg | | | 30 kg | | | |
| Foot region | Left Right Both | | | Left | Right | Both | Left | Right | Both | Left | Right | Both | | |
| 3 rd -5 th T | 0.82 | 0.85 | 0.84 | 0.94 | 0.80 | 0.87 | 0.92 | 0.85 | 0.88 | 0.94 | 0.90 | 0.92 | | |
| 2 nd T | 0.92 | 0.78 | 0.85 | 0.78 | 0.87 | 0.79 | 0.91 | 0.87 | 0.89 | 0.91 | 0.93 | 0.92 | | |
| 1 st T | 0.89 | 0.91 | 0.90 | 0.99 | 0.92 | 0.96 | 0.92 | 0.91 | 0.91 | 0.93 | 0.93 | 0.93 | | |
| 5MH | 0.93 | 0.94 | 0.94 | 0.93 | 0.97 | 0.95 | 0.93 | 0.95 | 0.94 | 0.83 | 0.94 | 0.89 | | |
| 4MH | 0.94 | 0.90 | 0.92 | 0.86 | 0.93 | 0.90 | 0.94 | 0.94 | 0.94 | 0.94 | 0.88 | 0.91 | | |
| 3MH | 0.96 | 0.95 | 0.95 | 0.94 | 0.96 | 0.95 | 0.91 | 0.94 | 0.92 | 0.92 | 0.91 | 0.92 | | |
| 2MH | 0.95 | 0.95 | 0.95 | 0.94 | 0.92 | 0.93 | 0.93 | 0.94 | 0.93 | 0.93 | 0.94 | 0.94 | | |
| 1MH | 0.92 | 0.95 | 0.94 | 0.94 | 0.95 | 0.94 | 0.92 | 0.81 | 0.86 | 0.94 | 0.93 | 0.93 | | |
| MF | 0.97 | 0.95 | 0.96 | 0.98 | 0.97 | 0.97 | 0.99 | 0.98 | 0.98 | 0.98 | 0.98 | 0.98 | | |
| Н | 0.96 | 0.86 | 0.91 | 0.91 | 0.94 | 0.93 | 0.94 | 0.96 | 0.95 | 0.82 | 0.96 | 0.89 | | |

 Table 5.6. Intra-class correlation coefficients result for F-Scan system regarding contact area at baseline

5.2.3. Contact pressure variable

The CP repeatability results obtained at baseline with the in-shoe pressure system highlighted 'good to excellent' ICC results in all of the anatomical areas investigated for the left, right and both feet together, as presented in **Table 5.7.** The ICC value ranges between (0.76 - 0.98) foot region locations.

| Contact pressure (kPa) | | | | | | | | | | | | | | |
|------------------------------------|-----------------|---------|------|------|-------|------|------|-------|------|------|-------|------|--|--|
| East region | | Zero kg | | | 10 kg | | | 20 kg | | | 30 kg | | | |
| root region | Left Right Both | | | Left | Right | Both | Left | Right | Both | Left | Right | Both | | |
| 3 rd -5 th T | 0.97 | 0.84 | 0.91 | 0.76 | 0.77 | 0.77 | 0.93 | 0.85 | 0.89 | 0.93 | 0.91 | 0.92 | | |
| 2 nd T | 0.96 | 0.83 | 0.89 | 0.92 | 0.86 | 0.89 | 0.91 | 0.89 | 0.90 | 0.94 | 0.97 | 0.95 | | |
| 1 st T | 0.93 | 0.90 | 0.91 | 0.93 | 0.96 | 0.95 | 0.95 | 0.96 | 0.96 | 0.98 | 0.92 | 0.95 | | |
| 5MH | 0.85 | 0.91 | 0.88 | 0.90 | 0.93 | 0.91 | 0.79 | 0.94 | 0.86 | 0.85 | 0.92 | 0.89 | | |
| 4MH | 0.88 | 0.90 | 0.89 | 0.90 | 0.90 | 0.90 | 0.79 | 0.91 | 0.85 | 0.86 | 0.87 | 0.86 | | |
| 3MH | 0.92 | 0.95 | 0.94 | 0.93 | 0.93 | 0.93 | 0.88 | 0.95 | 0.92 | 0.92 | 0.93 | 0.92 | | |
| 2MH | 0.94 | 0.77 | 0.86 | 0.92 | 0.91 | 0.91 | 0.85 | 0.95 | 0.90 | 0.77 | 0.95 | 0.86 | | |
| 1MH | 0.89 | 0.94 | 0.91 | 0.86 | 0.92 | 0.89 | 0.84 | 0.83 | 0.83 | 0.90 | 0.95 | 0.92 | | |
| MF | 0.89 | 0.98 | 0.93 | 0.97 | 0.86 | 0.92 | 0.93 | 0.78 | 0.86 | 0.88 | 0.96 | 0.92 | | |
| Н | 0.87 | 0.94 | 0.91 | 0.94 | 0.94 | 0.94 | 0.81 | 0.96 | 0.88 | 0.91 | 0.92 | 0.92 | | |

 Table 5.7. Intra-class correlation coefficients result for F-Scan system regarding contact pressure at baseline

5.2.4. Pressure time integral variable

The pressure time integral (PTI) values showed that 'good to excellent' ICC scores were found for all anatomical areas at the baseline for the left, right, and both feet, as can be seen in **Table 5.8.** When PTI analysis was carried out, the ICC indicated a highly repeatable trend for most of the anatomical areas investigated, ICC value ranges between (0.77 - 0.97) foot region locations. In all ten anatomical areas investigated, it was possible to observe 'good' ICC values for left, right and both feet. The values and results of the ICC for analysis show a generally good level of reliability, the quality of which is dependent on the region of the foot and the variable investigated with SBW loads.

| Pressure time integral (kPa/s) | | | | | | | | | | | | | | |
|------------------------------------|-----------------|---------|------|------|-------|------|------|-------|------|------|-------|------|--|--|
| | | Zero kg | | | 10 kg | | | 20 kg | | | 30 kg | | | |
| root region | Left Right Both | | | Left | Right | Both | Left | Right | Both | Left | Right | Both | | |
| 3 rd -5 th T | 0.97 | 0.91 | 0.94 | 0.90 | 0.85 | 0.87 | 0.94 | 0.88 | 0.91 | 0.81 | 0.94 | 0.87 | | |
| 2 nd T | 0.97 | 0.87 | 0.92 | 0.95 | 0.87 | 0.91 | 0.94 | 0.88 | 0.91 | 0.95 | 0.96 | 0.95 | | |
| 1 st T | 0.94 | 0.92 | 0.93 | 0.97 | 0.97 | 0.97 | 0.96 | 0.95 | 0.96 | 0.97 | 0.87 | 0.92 | | |
| 5MH | 0.95 | 0.87 | 0.91 | 0.80 | 0.83 | 0.82 | 0.85 | 0.91 | 0.88 | 0.85 | 0.91 | 0.88 | | |
| 4MH | 0.96 | 0.88 | 0.92 | 0.87 | 0.82 | 0.84 | 0.83 | 0.89 | 0.86 | 0.88 | 0.92 | 0.90 | | |
| 3MH | 0.95 | 0.91 | 0.93 | 0.88 | 0.77 | 0.82 | 0.89 | 0.92 | 0.90 | 0.90 | 0.94 | 0.92 | | |
| 2MH | 0.91 | 0.90 | 0.91 | 0.92 | 0.89 | 0.90 | 0.94 | 0.90 | 0.92 | 0.81 | 0.95 | 0.88 | | |
| 1MH | 0.85 | 0.85 | 0.85 | 0.91 | 0.88 | 0.89 | 0.85 | 0.92 | 0.88 | 0.90 | 0.95 | 0.93 | | |
| MF | 0.89 | 0.89 | 0.89 | 0.93 | 0.83 | 0.88 | 0.89 | 0.87 | 0.88 | 0.87 | 0.96 | 0.91 | | |
| Н | 0.89 | 0.96 | 0.92 | 0.93 | 0.94 | 0.93 | 0.89 | 0.95 | 0.92 | 0.96 | 0.91 | 0.94 | | |

Table 5.8. Intra-class correlation coefficients result for F-Scan system regarding pressure time integral at baseline

5.2.5. Peak pressure

The peak pressure (PP) values displayed that 'good' ICC scores were presented for all anatomical areas at baseline for the left, right and both feet, as shown in **Table 5.9.** In all ten anatomical areas for the four simulations tested, there was potential to note 'excellent' and 'good to excellent' ICC values for left, right and both feet. The ICC value ranges between (0.78 - 0.99) foot region locations. Thus, the findings of this examination display a commonly better level of accuracy, with quality based on the foot region and measures researched.

| | Peak pressure (kPa) | | | | | | | | | | | | |
|------------------------------------|---------------------|---------|------|------|-------|------|------|-------|------|------|-------|------|--|
| East magian | | Zero kg | | | 10 kg | | | 20 kg | | | 30 kg | | |
| root region | Left Right Both | | | Left | Right | Both | Left | Right | Both | Left | Right | Both | |
| 3 rd -5 th T | 0.97 | 0.86 | 0.92 | 0.85 | 0.78 | 0.82 | 0.87 | 0.85 | 0.86 | 0.92 | 0.91 | 0.92 | |
| 2 nd T | 0.95 | 0.84 | 0.90 | 0.94 | 0.84 | 0.89 | 0.90 | 0.81 | 0.85 | 0.94 | 0.96 | 0.95 | |
| 1 st T | 0.91 | 0.90 | 0.90 | 0.94 | 0.94 | 0.94 | 0.96 | 0.97 | 0.97 | 0.99 | 0.96 | 0.97 | |
| 5MH | 0.83 | 0.88 | 0.85 | 0.89 | 0.93 | 0.91 | 0.79 | 0.94 | 0.86 | 0.85 | 0.90 | 0.87 | |
| 4MH | 0.87 | 0.88 | 0.88 | 0.91 | 0.89 | 0.90 | 0.81 | 0.92 | 0.86 | 0.89 | 0.84 | 0.86 | |
| 3MH | 0.93 | 0.95 | 0.94 | 0.93 | 0.93 | 0.93 | 0.90 | 0.95 | 0.93 | 0.91 | 0.94 | 0.93 | |
| 2MH | 0.94 | 0.96 | 0.95 | 0.94 | 0.91 | 0.92 | 0.95 | 0.95 | 0.95 | 0.96 | 0.94 | 0.95 | |
| 1MH | 0.89 | 0.88 | 0.89 | 0.86 | 0.93 | 0.90 | 0.88 | 0.95 | 0.92 | 0.93 | 0.95 | 0.94 | |
| MF | 0.79 | 0.84 | 0.82 | 0.98 | 0.82 | 0.90 | 0.87 | 0.84 | 0.85 | 0.88 | 0.88 | 0.88 | |
| Н | 0.94 | 0.96 | 0.95 | 0.91 | 0.94 | 0.93 | 0.85 | 0.94 | 0.89 | 0.90 | 0.95 | 0.93 | |

 Table 5.9. Intra-class correlation coefficients result for F-Scan regarding peak pressure at baseline
5.3. Discussion

The result of this experiment suggests that the increased weight by SBW, can notably rise pressures on the sole of the foot when other complicating agents, like the existence of disfigurements and restricted joint mobility but are controlled. The simulated body weight was achieved with the addition of 10, 20 and 30 kg of weight which increased foot pressures variables These findings were consistent for all regions of the foot. The result of the ANOVA displayed a noteworthy variation in foot measures, for the 10, 20 and 30 kg simulated body weight conditions compared with the normal group (0 kg). In general, an increase in the plantar measures was noticed, and alterations in gait pattern during simulated body weight gait was also observed when compared to the health category. The impacts of simulated body weight (four loads) were consistent at any foot measures given the simulations as shown in (**Figure 5.1 a, b, c, d, e**).

5.3.1. Force variable

The present study observed statistically significant increases in PF which occurred in the H, MF, 1MH, 2MH, and 3MH regions, for 10, 20 and 30 kg simulated body weights groups versus the NBW group (0 kg), as presented in **Table 5.1**. The findings of this study are clinically significant since areas of very high F or pressure under the foot are good indicators of potential damage being caused to the underlying tissue, particularly in symptomatic diabetic feet (Cavanagh et al., 1993).

This trend was particularly true in regions of the foot where relatively high loading typically occurs through gait, such as the H and forefoot (1MH, 2MH, and 3MH). Generally, the lower reliability was found in the typically less-loaded areas, such as the toes region. This study established a range of values of CV% for PTI beneath the toes region. Our findings were in absolute agreement with those of Gurney et al. (2008). Both studies found that the toes also showed the highest CV% across F parameter within all un-simulated and SBW.

5.3.2. Contact area variable

The higher foot contact area (CA) was shown within SBW groups (30, 20 and 10 kg) compared control counterparts (0 kg), as shown in **Table 5.2**. The largest CA with the insole system starting with the H region, 1MH and MF whereas, CA variables were slightly changed within the second toe zone. It is found that CA measure within the 3rd-5thT region in SBW groups, was a smaller value than in the control group.

This study established a range of values of CV% for CA beneath the toes region. Our findings were in absolute agreement with those of Gurney et al. (2008). Both studies found that the toes region had lower reliability, which was found in the typically less-loaded areas. The toes also showed the highest CV% across CA variable within all un-simulated and SBW.

5.3.3. Contact pressure variable

The 2MH, H, and 3MH regions respectively showed greater changes with simulated body weight, as displayed in **Table 5.3**. The variation between the averages of simulated body weights was significant (P < 0.001). The H region displayed differences in contact pressure (CP) values for the 10, 20 and 30 kg SBW group compared with the NBW group and thus appeared sensitive to smaller changes in body weight. The present experiment established no differences in mean CP within the $2^{nd}-5^{th}T$ and $1^{st}T$ regions. This result is consistent with observations reported by Martinez-Nova et al. (2008), who did not identify body mass as a statistically significant predictor of mean pressure within the lesser digits and $1^{st}T$ regions.

This study established a range of values of CV% for CP beneath the toes region. Our findings were in absolute agreement with those of Gurney et al. (2008). Both studies found that the lower reliability was found in the typically less-loaded areas, such as the toes region. The toes also showed the highest CV% across F parameter within all NBW and SBW.

5.3.4. Pressure time integral variable

The pressure time integral (PTI, kPa s) measure notably increased between the SBW groups compared with the NBW group. PTI variable was significantly different for

CHAPTER 5: RESULTS OF EXPERIMENT 2

all foot regions under SBW with the 10, 20 and 30 kg compared with the control groups (0 kg), as seen in **Table 5.4.** The H zone displayed differences in PTI for the 10, 20 and 30 kg simulated body weight states compared with the NBW (0 kg), and thus appeared sensitive to smaller changes in body weight. This may be due to a function of the increased mass alone. It would seem more likely for a difference in PTI to occur due to the larger SBW groups. This may also explain the absence of statistically significant differences in PTI for all simulated weights under the H, MF, 1MH, 2MH, 3MH, 4MH and 5MH, between the unweighted and weighted 30 kg. Also, these findings were in absolute agreement with those of McKay et al. (2017). Both studies found that the forefoot (MHs), H and MF had the biggest values of all the mentioned PTI measures.

This study established a range of values of CV% for PTI beneath the toes region. Our findings were in absolute agreement with those of Gurney et al. (2008). Both studies found that the toes region also showed the highest CV% across the F parameter within all un-simulated and SBW. Whereas, this region had lower reliability, which was found in the typically less-loaded areas, such as the toes region

5.3.5. Peak pressure variable

The measurement of peak pressure showed that statistically significant increases in peak pressure only occurred in the H, 2MH to 5MH, MF and 1stT regions for 10, 20 and 30 kg simulated versus control group, and the MF, 2MH and 1stT regions for 30 kg versus control group. These findings concurred with the study conducted by Vela et al. (1998). They found that statistically significant increases ($p \le 0.05$) in PP under the H, MF, 1M, and lesser metatarsals (lesser toes) when 9.1 and 18.2kg of additional load were added to subjects.

In the present study, no statistically significant increases were found in PP during walking for the MF region with simulated body weight. However, this contrasts with Birtane and Tuna (2004), who reported that, out of six plantar regions, only the MF area recorded a statistically significant increase in peak plantar pressure when obese subjects (BMI 30.0-34.99 kg/m²) were compared to non-obese controls. In contrast, previous studies by Hills et al. (2001) and Clarke (1981) found that healthy weight adults showed low correlation between body weight and PP under the

125

feet. These studies did not consider PP across a range of anatomical locations and attributed the lack of relationship between weight and PP to either increased foot CA through the stance phase of gait or the distribution of high loads to larger anatomical areas of the foot. Contrary to the findings in adults, body weight was the main factor in the magnitude of pressure under the feet of school children aged between six and ten years (Hennig et al., 1994).

5.3.6. Repeatability of the F-Scan system

A capacitive pressure distribution insole system sampling at 100 Hz was used to collect plantar pressure patterns during in-shoe walking at a self-selected speed. Five parameters were investigated (Keijsers et al., 2010; Coda & Santos, 2015): force, contact area, contact pressure, pressure time integral and peak pressure, and these were investigated in ten areas of the foot using the FScan mask method of subdividing the foot into ten anatomical areas of interest. Individual means for all three repeated trials for each foot were calculated, and these values were used to calculate ICC for all parameters. The accuracy of foot plantar pressure was assessed with the mean value of three repeated trials within simulated body weight. These findings indicate that the insole system has displayed 'good' to 'excellent' reliability for the five analysis variables of F, CA, CP, PTI and PP for all ten evaluated areas, with the excluding of the mean pressure value for the second toe.

All insole systems produced the highest ICC values for pressure above 100 kPa. To summarise, an appropriate pressure measurement device is selected depending on the duration of loading, the magnitude of loading and the outcome variables sought. Medilogic and Tekscan are most effective between 200-300 kPa; Pedar performed well across all pressures (Price et al., 2014).

The findings of this study showed a commonly higher level of accuracy. Regions with typically high simulation characteristics, like the MF, demonstrated a greater level of precision in the ICC's (>0.9) than lower simulated zones, such the $3^{rd}-5^{th}T$ (<0.8). The conclusion of this research is that plantar pressure measurements can be applied in comparative assessments as the measures of repeatability are acceptable for the measurers and foot zones usually used in the study of clinical

people like abnormal plantar pressure distribution with overweight and the obese foot or neuropathic diabetics.

According to Kernozek et al. (1996) and Putti et al. (2007) dynamic pedobarography is a constantly expanding field with new devices being designed and employed for various clinical applications. The repeatability of the earlier Pedar inshoe systems was evaluated by Kernozek et al. (1996) and Putti et al. (2007). During the normal gait process, the heel initiates ground contact and the load is progressively spread across the various areas of the foot, of which the metatarsal heads region is one of the most important. The insole (Pedar-X ®) system showed a similar repeatability pattern for most foot areas considering the various parameters evaluated. When comparing the values of both feet (L and R), the majority of parameters showed similar repeatability for various foot regions. The results of the current study are consistent with that of Putti et al. (2007), and Ramanathan et al. (2010). Therefore, rise accuracy amidst testing periods within these foot zones as has been displayed by the result is very acceptable for clinical checking purposes. The influence of different foot orthotics on initial loading through gait were investigated by Cornwall and McPoil (1997). They pointed out that the between-trial accuracy ICC was markedly high for the forefoot areas but less so for other zones, which is in conformity with the outcomes for various measures. In addition, Gurney et al. (2008) determined the reliability of repeated plantar pressure distribution measurements during normal gait across multiple testing sessions. They established that the between-trial accuracy ICC was notably high for the centre metatarsal head zone, which is in accordance with the current outcomes for various measures.

5.4. Chapter summary

Dependant on the results offered in this chapter, the simulated body weight has achieved its intended purpose when using insole sets. The experiment investigated the effect of simulated body weight added at known intervals on foot variables (force, contact area, contact pressure, pressure time integral and peak pressure) in adult subjects during walking. The H, 2MH, 3MH and 1stT regions were considered the main foot regions affected by SBW within all important pressure variable characteristics. The determined measures for variance (CV) may serve for power estimation in clinical intervention experiments. The 'good to excellent' ICC scores

CHAPTER 5: RESULTS OF EXPERIMENT 2

were gathered in most of the ten plantar pressure areas when investigating on foot variables. The findings of this experiment pointed out that an insole pressure system is a reliable tool for evaluating foot plantar forces and pressures through the walk. The plantar pressure measures can be used in relative assessments, as the measures of repeatability are favourable for the measures and foot zones generally utilised in the study of clinical people like neuropathic diabetics.

The next chapter of this study involves developing an innovative technique for the calculation of the centroid of the area of contact data during the moment of heel strike, midstance and toe-off phase of gait.

6. RESULT OF EXPERIMENT 3

6.1. Introduction

There are four main parts to this chapter. The first section (6.2) discusses the third experiment which studied the effect of simulated body weight on the centroid of the area of contact with heel strike, midstance and toe-off phases in adults. The section (6.2.1) introduces the centroid of the foot plantar in three phases (heel strike, midstance, and toe-off) during gait. The third section presents a discussion of the findings. Finally, the last section focuses on a summary of the chapter.

The enthusiasm for studying the human foot and gait loading to determine and manage impairments associated with various musculoskeletal, integumentary, and neurological disorders has intensified considerably in recent years. This type of research is particularly important in identifying the causes of diabetic foot ulceration and its treatment. In this chapter, the dynamic ground reaction force is captured with an insole pressure system. This chapter presents the results that answer objective three, which presents the method used to capture plantar pressure; an innovative technique for the calculation of the centroid of the area of contact captured by the developed system. The results and analysis are limited to three significant stages of the load phase: 1) Heel strike, where the heel is in initial contact with the substrate, 2) Midstance or the instant in the gait cycle where the swing leg passes the support leg, and 3) Toe-off or the instant before the plantar surface leaves the substrate. Thirty-one adults with a neutral foot type and a mean age of 37.55 ± 6.92 years, mean body height of 1.74±0.08 m, body weight of 89.82±17.92 kg, and BMI of 29.7±5.1 kg/m^2 participated in the study. The chapter discusses the results of the study at centroid of the foot plantar during the stance phase of walk. This feature is related to the habits expressed by the participants' feet while carrying the simulated body weights. Two coordinate values (x-axis and y-axis) during the three phases are expressed as a percent of the total three trials studied.

6.2. The centroid of heel strike contact surface with the simulated body weight

The objective was to examine the impact of simulated body weight (SBW) within the participants' centre of the heel zone, by adding (10, 20, 30 kg). A sample of thirty-one adult males was tested. These subjects were grouped according to SBW and NBW to four groups: 0, 10, 20, and 30 kg. Participant characteristics are presented in **Table 3.3**.

6.2.1. X-axis coordinate of heel strike

The results show that, for the 31 subjects, the x-axis coordinate value increased at heel strike when moving with the simulated body weight. In all 4th groups (SBW and NBW), left foot and right foot values were almost the same values for each condition (0, 10, 20, and 30 kg). But overall, left and right foot values showed differences between the control (0 kg) and SBW (10, 20, and 30 kg) groups. The results for the x-axis coordinate values for both feet are given in **Table 6.1**.

The mean x-axis coordinate values under simulated body weight (30 kg) loads were higher in both feet, compared with the other weights of 20 kg and 10 kg and the unsimulated body weight. The x-axis coordinate mean values for NBW (0 kg) and SBW of 10, 20, 30 kg for the left foot were 5.54, 7.80, 9.67, and 11.77 mm, respectively. Moreover, the x-axis coordinate mean values for the right foot of 5.41, 7.73, 9.69, and 11.76 mm, respectively were obtained on the NBW (0 kg) and SBW with 10, 20, 30 kg.

CHAPTER 6: RESULTS OF EXPERIMENT 3

| Table 6.1. The mean value and SD for left and right foot in Heel strike phase (x-axis coordinate, min) | Table 6.1. | . The mean | value and | SD for l | eft and | right foot | in Heel | l strike | phase | (x-axis | coordinate, | mm) |
|---|------------|------------|-----------|----------|---------|------------|---------|----------|-------|---------|-------------|-----|
|---|------------|------------|-----------|----------|---------|------------|---------|----------|-------|---------|-------------|-----|

| | Heel strike-Left and right foot X-axis | | | | | | | | | | | | | | | |
|--------------|--|------|---------|------|---------|------|---------|------|---------|------|---------|------|---------|------|---------|------|
| Participants | | 01 | cg | | | 10 | kg | | | 20 | kg | | | 30 | kg | |
| Id | Mean Lf | SD | Mean Rf | SD | Mean Lf | SD | Mean Rf | SD | Mean Lf | SD | Mean Rf | SD | Mean Lf | SD | Mean Rf | SD |
| 1 | 5.97 | 0.29 | 5.10 | 0.46 | 7.47 | 0.47 | 7.87 | 0.57 | 10.33 | 0.21 | 8.85 | 0.26 | 11.90 | 0.75 | 11.90 | 1.14 |
| 2 | 5.77 | 0.49 | 5.30 | 0.78 | 8.20 | 0.36 | 7.97 | 0.25 | 9.37 | 0.58 | 9.93 | 0.06 | 11.75 | 0.15 | 11.75 | 0.23 |
| 3 | 5.77 | 0.49 | 5.30 | 0.78 | 8.20 | 0.36 | 7.97 | 0.25 | 9.84 | 0.58 | 9.93 | 0.06 | 11.47 | 0.15 | 11.47 | 0.23 |
| 4 | 6.30 | 0.35 | 5.40 | 0.17 | 8.50 | 0.10 | 8.00 | 0.10 | 9.50 | 0.10 | 10.60 | 0.60 | 12.14 | 0.32 | 12.14 | 0.17 |
| 5 | 5.50 | 0.87 | 6.10 | 0.40 | 7.53 | 0.06 | 7.77 | 0.67 | 9.27 | 0.31 | 9.07 | 0.55 | 11.50 | 0.26 | 11.50 | 0.64 |
| 6 | 5.83 | 0.21 | 6.03 | 1.45 | 7.63 | 1.17 | 7.87 | 0.15 | 9.47 | 0.06 | 10.23 | 0.80 | 11.85 | 0.26 | 11.85 | 0.80 |
| 7 | 6.03 | 0.06 | 3.33 | 0.06 | 8.53 | 0.55 | 7.53 | 0.55 | 9.38 | 0.17 | 9.30 | 0.17 | 11.67 | 0.35 | 11.67 | 0.35 |
| 8 | 5.87 | 0.25 | 5.50 | 0.10 | 8.03 | 0.60 | 7.70 | 0.17 | 9.87 | 0.32 | 7.83 | 0.06 | 11.23 | 0.40 | 11.23 | 0.40 |
| 9 | 5.33 | 0.06 | 3.33 | 0.06 | 7.60 | 0.53 | 7.60 | 0.53 | 10.43 | 0.83 | 9.43 | 0.83 | 11.53 | 0.06 | 11.53 | 0.06 |
| 10 | 5.13 | 0.57 | 5.90 | 0.26 | 8.03 | 0.29 | 7.93 | 0.21 | 9.57 | 0.06 | 10.23 | 0.15 | 11.71 | 0.15 | 11.71 | 0.84 |
| 11 | 5.33 | 0.06 | 5.33 | 0.06 | 7.60 | 0.53 | 7.60 | 0.53 | 9.43 | 0.83 | 10.43 | 0.83 | 11.65 | 0.06 | 11.65 | 0.06 |
| 12 | 5.90 | 0.20 | 5.93 | 0.38 | 8.00 | 0.10 | 7.50 | 0.26 | 9.43 | 1.10 | 9.93 | 0.86 | 11.73 | 0.51 | 11.73 | 0.44 |
| 13 | 6.33 | 0.06 | 5.33 | 0.06 | 7.60 | 0.53 | 7.60 | 0.53 | 9.84 | 0.83 | 10.43 | 0.83 | 11.85 | 0.06 | 11.85 | 0.06 |
| 14 | 5.33 | 0.06 | 5.63 | 0.21 | 7.60 | 0.53 | 8.90 | 0.10 | 9.43 | 0.83 | 9.97 | 0.38 | 11.65 | 0.06 | 11.65 | 0.21 |
| 15 | 5.83 | 0.46 | 5.33 | 0.00 | 8.10 | 0.20 | 7.33 | 1.70 | 9.49 | 0.26 | 10.07 | 0.74 | 12.00 | 0.95 | 12.00 | 0.50 |
| 16 | 5.33 | 0.06 | 5.33 | 0.06 | 7.60 | 0.53 | 7.60 | 0.53 | 9.43 | 0.83 | 10.43 | 0.83 | 11.75 | 0.06 | 11.75 | 0.06 |
| 17 | 5.33 | 0.25 | 6.73 | 0.49 | 7.83 | 0.58 | 7.67 | 0.61 | 10.23 | 0.47 | 10.10 | 0.30 | 12.03 | 0.06 | 12.03 | 0.17 |
| 18 | 5.20 | 0.26 | 5.37 | 0.59 | 8.03 | 0.12 | 8.90 | 0.48 | 10.03 | 0.15 | 9.93 | 0.15 | 11.92 | 0.03 | 11.92 | 0.10 |
| 19 | 5.07 | 0.70 | 6.93 | 0.15 | 8.00 | 0.61 | 7.43 | 0.29 | 9.54 | 0.35 | 9.43 | 0.25 | 11.57 | 0.29 | 11.57 | 0.32 |
| 20 | 5.33 | 0.06 | 5.33 | 0.06 | 8.53 | 0.55 | 7.53 | 0.55 | 9.53 | 0.17 | 9.30 | 0.17 | 11.47 | 0.35 | 11.37 | 0.35 |
| 21 | 5.33 | 0.06 | 5.33 | 0.06 | 7.60 | 0.53 | 7.68 | 0.53 | 9.43 | 0.83 | 9.94 | 0.83 | 11.53 | 0.06 | 11.53 | 0.06 |
| 22 | 5.00 | 0.10 | 6.33 | 0.03 | 8.30 | 1.41 | 7.90 | 0.10 | 10.20 | 0.53 | 10.07 | 0.55 | 12.03 | 0.45 | 12.03 | 0.32 |
| 23 | 5.03 | 0.45 | 5.40 | 0.40 | 7.63 | 0.47 | 7.33 | 0.45 | 9.90 | 0.85 | 10.43 | 0.12 | 12.59 | 0.62 | 12.59 | 0.70 |
| 24 | 5.37 | 0.59 | 5.67 | 0.12 | 7.87 | 0.51 | 8.03 | 0.58 | 9.29 | 0.06 | 10.11 | 0.87 | 12.30 | 0.36 | 12.30 | 0.42 |
| 25 | 5.40 | 0.20 | 6.03 | 0.20 | 7.53 | 0.12 | 7.23 | 0.32 | 9.97 | 0.65 | 9.23 | 1.15 | 12.03 | 0.32 | 12.03 | 0.65 |
| 26 | 5.33 | 0.06 | 3.33 | 0.06 | 7.60 | 0.53 | 7.53 | 0.55 | 9.43 | 0.83 | 9.30 | 0.17 | 11.53 | 0.06 | 11.53 | 0.35 |
| 27 | 5.50 | 0.10 | 5.40 | 0.35 | 8.23 | 0.23 | 7.77 | 0.23 | 9.70 | 0.36 | 9.57 | 0.29 | 11.57 | 0.12 | 11.47 | 0.10 |
| 28 | 5.48 | 0.26 | 5.63 | 0.31 | 7.53 | 0.06 | 7.23 | 0.06 | 9.60 | 0.26 | 7.86 | 0.21 | 11.73 | 0.38 | 11.73 | 1.59 |
| 29 | 5.43 | 0.64 | 5.03 | 0.49 | 5.47 | 0.32 | 7.43 | 0.12 | 9.83 | 0.31 | 9.80 | 0.26 | 12.00 | 0.17 | 12.10 | 0.17 |
| 30 | 6.13 | 0.32 | 5.97 | 0.12 | 7.87 | 0.23 | 7.60 | 0.35 | 9.17 | 0.25 | 9.98 | 0.10 | 11.83 | 0.25 | 11.80 | 0.00 |
| 31 | 5.13 | 0.47 | 5.10 | 0.42 | 7.47 | 0.06 | 7.63 | 0.15 | 9.97 | 0.15 | 8.65 | 0.15 | 11.20 | 0.82 | 11.20 | 0.36 |
| mean | 5.54 | | 5.41 | | 7.80 | | 7.73 | | 9.67 | | 9.69 | | 11.77 | | 11.76 | |

6.2.2. Y-axis coordinate of heel-strike

The results show that the y-axis coordinate values increased at heel strike for the 31 subjects moving with simulated body weight. In all four groups (0, 10, 20, and 30 kg) left foot and right foot values were almost the same values for each condition. However, left foot and right values did show differences between the normal state (0 kg) and simulated (10, 20, and 30 kg) states. The results for y-axis coordinate values for both feet are given in **Table 6.2**.

The NBW group (0 kg) had a lower y-axis coordinate mean values than the SBW group (10, 20, 30 kg) for both feet. The y-axis coordinates mean values for the NBW (0 kg) and SBW with 10, 20, 30 kg were 6.17, 8.18, 9.67, and 11.97 mm, respectively for the left foot. In addition, the y-axis coordinates mean values based on the NBW (0 kg) and SBW with 10, 20, 30kg for the right foot were 6.20, 8.17, 10.10, and 11.96 mm, respectively.

CHAPTER 6: RESULTS OF EXPERIMENT 3

Table 6.2. The mean value and SD for left and right foot in Heel strike phase (y-axis coordinate, mm)

| | Heel strike- Left and right foot Y-axis | | | | | | | | | | | | | | | |
|--------------|---|------|---------|------|---------|------|---------|------|---------|------|---------|------|---------|------|---------|------|
| Participants | | 01 | ĸg | | | 10 | lkg | | | 20 | kg | | | 30 | lkg | |
| Id | Mean Lf | SD | Mean Rf | SD | Mean Lf | SD | Mean Rf | SD | Mean Lf | SD | Mean Rf | SD | Mean Lf | SD | Mean Rf | SD |
| 1 | 6.50 | 0.69 | 6.07 | 0.40 | 8.97 | 0.75 | 8.77 | 0.18 | 10.23 | 0.21 | 10.47 | 8.87 | 11.43 | 0.76 | 11.45 | 0.55 |
| 2 | 6.23 | 1.08 | 6.33 | 0.55 | 8.60 | 0.20 | 7.95 | 0.17 | 9.83 | 0.31 | 9.77 | 0.26 | 11.53 | 0.49 | 11.52 | 0.12 |
| 3 | 6.23 | 1.08 | 6.33 | 0.55 | 8.60 | 0.20 | 8.50 | 0.17 | 9.83 | 0.31 | 10.70 | 0.26 | 11.53 | 0.49 | 11.53 | 0.12 |
| 4 | 6.20 | 0.00 | 5.67 | 0.67 | 8.10 | 0.10 | 8.13 | 0.52 | 10.22 | 0.25 | 10.20 | 0.30 | 11.73 | 0.24 | 11.69 | 0.17 |
| 5 | 5.73 | 0.59 | 6.67 | 0.49 | 7.77 | 0.42 | 8.60 | 0.92 | 9.40 | 0.26 | 9.85 | 0.96 | 11.40 | 0.20 | 11.42 | 0.06 |
| 6 | 6.30 | 0.61 | 5.83 | 1.84 | 8.50 | 0.95 | 8.07 | 0.32 | 9.77 | 0.31 | 10.92 | 1.14 | 12.03 | 0.37 | 11.99 | 0.90 |
| 7 | 6.73 | 0.23 | 6.73 | 0.23 | 8.07 | 0.81 | 8.07 | 0.81 | 9.13 | 0.71 | 10.01 | 0.71 | 12.15 | 0.78 | 12.05 | 0.78 |
| 8 | 5.93 | 0.12 | 6.53 | 0.06 | 8.13 | 0.67 | 8.27 | 0.32 | 9.70 | 0.53 | 9.73 | 0.59 | 11.67 | 0.23 | 11.57 | 0.26 |
| 9 | 5.73 | 0.23 | 5.78 | 0.23 | 8.22 | 0.10 | 8.00 | 0.10 | 9.57 | 0.15 | 10.57 | 0.15 | 12.80 | 0.48 | 12.79 | 0.48 |
| 10 | 6.30 | 0.50 | 6.70 | 0.36 | 8.60 | 0.10 | 8.07 | 0.21 | 9.27 | 0.21 | 10.37 | 0.21 | 11.97 | 0.12 | 11.97 | 0.40 |
| 11 | 5.73 | 0.23 | 5.73 | 0.23 | 8.11 | 0.10 | 8.11 | 0.10 | 9.86 | 0.15 | 10.36 | 0.15 | 12.80 | 0.48 | 12.80 | 0.48 |
| 12 | 5.93 | 0.50 | 6.87 | 0.35 | 7.83 | 0.15 | 8.43 | 0.15 | 9.13 | 0.91 | 9.77 | 0.61 | 11.86 | 0.29 | 11.82 | 0.38 |
| 13 | 5.73 | 0.23 | 5.73 | 0.23 | 8.00 | 0.10 | 7.89 | 0.10 | 9.57 | 0.15 | 9.76 | 0.15 | 12.33 | 0.48 | 12.33 | 0.48 |
| 14 | 5.73 | 0.23 | 6.17 | 0.06 | 8.12 | 0.10 | 7.87 | 0.25 | 9.87 | 0.15 | 10.27 | 0.23 | 12.50 | 0.48 | 12.50 | 0.58 |
| 15 | 5.83 | 0.29 | 6.73 | 1.00 | 8.03 | 0.15 | 7.83 | 1.22 | 9.53 | 0.25 | 10.10 | 9.27 | 11.86 | 0.06 | 11.83 | 0.33 |
| 16 | 5.73 | 0.23 | 5.73 | 0.23 | 8.20 | 0.10 | 8.10 | 0.10 | 9.57 | 0.15 | 9.76 | 0.15 | 12.80 | 0.48 | 12.81 | 0.48 |
| 17 | 6.07 | 0.56 | 6.70 | 1.31 | 8.90 | 1.59 | 7.85 | 0.62 | 9.67 | 0.12 | 9.82 | 0.82 | 12.03 | 0.35 | 12.01 | 0.17 |
| 18 | 6.73 | 0.23 | 5.57 | 0.06 | 7.65 | 0.42 | 7.89 | 0.35 | 10.30 | 0.44 | 10.03 | 0.15 | 11.53 | 0.06 | 11.53 | 0.25 |
| 19 | 5.99 | 0.10 | 6.27 | 0.31 | 8.03 | 0.55 | 8.63 | 0.91 | 9.87 | 0.48 | 10.10 | 0.20 | 11.87 | 0.25 | 11.87 | 0.12 |
| 20 | 6.73 | 0.23 | 5.73 | 0.23 | 8.07 | 0.81 | 8.07 | 0.81 | 9.13 | 0.71 | 10.13 | 0.71 | 12.50 | 0.78 | 12.60 | 0.78 |
| 21 | 6.73 | 0.23 | 6.73 | 0.23 | 8.12 | 0.10 | 8.02 | 0.10 | 9.67 | 0.15 | 10.76 | 0.15 | 12.10 | 0.48 | 12.20 | 0.48 |
| 22 | 5.70 | 0.40 | 5.73 | 0.47 | 7.53 | 0.81 | 8.85 | 0.36 | 10.00 | 0.62 | 10.17 | 0.92 | 11.82 | 0.83 | 11.83 | 0.45 |
| 23 | 6.47 | 0.18 | 6.53 | 0.12 | 8.03 | 0.21 | 8.27 | 0.50 | 9.57 | 0.38 | 9.37 | 0.51 | 11.70 | 0.38 | 11.60 | 0.45 |
| 24 | 6.20 | 0.79 | 6.80 | 0.70 | 8.90 | 0.53 | 7.60 | 0.81 | 9.17 | 1.22 | 10.37 | 0.68 | 11.86 | 0.66 | 11.86 | 0.46 |
| 25 | 6.12 | 0.35 | 5.13 | 0.70 | 8.37 | 0.06 | 8.57 | 0.67 | 9.77 | 0.90 | 9.86 | 0.99 | 12.10 | 0.46 | 12.04 | 0.65 |
| 26 | 5.73 | 0.23 | 5.73 | 0.23 | 8.00 | 0.10 | 8.07 | 0.81 | 9.57 | 0.15 | 9.91 | 0.71 | 12.00 | 0.48 | 12.10 | 0.78 |
| 27 | 6.63 | 0.32 | 6.57 | 0.45 | 8.20 | 0.35 | 7.13 | 0.12 | 9.70 | 0.10 | 9.83 | 0.29 | 12.07 | 0.32 | 12.00 | 0.47 |
| 28 | 6.00 | 0.32 | 6.70 | 0.66 | 8.13 | 0.93 | 8.80 | 0.25 | 9.63 | 0.81 | 10.20 | 0.26 | 11.87 | 0.15 | 11.85 | 0.76 |
| 29 | 6.83 | 0.90 | 6.00 | 0.36 | 8.27 | 0.31 | 8.13 | 0.21 | 9.73 | 0.51 | 10.03 | 0.42 | 11.75 | 0.25 | 11.71 | 0.45 |
| 30 | 6.17 | 0.67 | 6.10 | 0.00 | 7.57 | 0.45 | 8.43 | 0.12 | 9.93 | 0.06 | 9.93 | 0.00 | 11.73 | 0.15 | 11.70 | 0.17 |
| 31 | 6.50 | 0.26 | 6.30 | 0.00 | 8.03 | 0.23 | 8.37 | 0.40 | 9.64 | 0.42 | 9.97 | 0.21 | 11.72 | 1.01 | 11.71 | 0.64 |
| mean | 6.11 | | 6.20 | | 8.18 | | 8.17 | | 9.67 | | 10.10 | | 11.97 | | 11.96 | |

6.3. The centroid of midstance contact surface for the simulated body weight

For all 31 subjects, the x-axis coordinate of midstance changes when moving. The maximum and minimum distributions of the left heel in the midstance phase are similar to those observed for the right foot, as shown in **Table 6.3**.

The greater mean x-axis coordinate value for both left and right midstance within the SBW group (10, 20, 30 kg) were found compared with the mean x-axis coordinate in midstance phase for both left and right value with NBW (0 kg) group.

6.3.1. X-axis coordinate of midstance

The mean values of the maximum x-axis coordinate for both left and right feet within the SBW sets were 4.53 and 4.46 mm for the 10 kg; 5.48 and 5.48 mm for 20 kg; and 6.60 and 6.58 mm with 30 kg. The mean values for x-axis coordinate, both left and right value, with NBW were 2.99 and 2.98 mm, respectively.

CHAPTER 6: RESULTS OF EXPERIMENT 3

| | Midstance- Left and right foot X-axis | | | | | | | | | | | | | | | |
|--------------|---------------------------------------|------|---------|------|---------|------|---------|------|---------|------|---------|------|---------|------|---------|------|
| Participants | | 01 | ĸg | | | 10 | lkg | | | 20 | kg | | | 30 | kg | |
| Id | Mean Lf | SD | Mean Rf | SD | Mean Lf | SD | Mean Rf | SD | Mean Lf | SD | Mean Rf | SD | Mean Lf | SD | Mean Rf | SD |
| 1 | 3.63 | 0.96 | 3.63 | 0.96 | 4.67 | 0.75 | 4.62 | 0.75 | 5.70 | 0.17 | 5.70 | 0.06 | 6.90 | 0.95 | 6.90 | 0.95 |
| 2 | 3.07 | 0.15 | 3.07 | 0.15 | 4.80 | 0.10 | 4.80 | 0.10 | 5.33 | 0.40 | 5.33 | 0.06 | 6.73 | 0.38 | 6.73 | 0.38 |
| 3 | 2.83 | 0.23 | 2.83 | 0.23 | 4.70 | 0.10 | 4.70 | 0.10 | 5.23 | 0.29 | 5.54 | 0.21 | 6.87 | 0.31 | 6.87 | 0.31 |
| 4 | 2.33 | 0.06 | 2.33 | 0.15 | 4.53 | 0.15 | 4.41 | 0.12 | 5.43 | 0.06 | 5.43 | 0.10 | 6.87 | 0.31 | 6.80 | 0.38 |
| 5 | 2.27 | 0.25 | 2.27 | 0.25 | 4.40 | 0.53 | 4.38 | 0.53 | 5.07 | 0.21 | 5.07 | 0.95 | 6.97 | 0.15 | 6.97 | 0.62 |
| 6 | 2.50 | 0.82 | 2.50 | 0.09 | 4.13 | 0.38 | 4.13 | 0.15 | 5.10 | 0.10 | 5.10 | 0.15 | 6.70 | 0.20 | 6.70 | 0.44 |
| 7 | 3.90 | 0.26 | 3.90 | 0.96 | 4.20 | 0.95 | 4.11 | 0.75 | 5.07 | 0.06 | 5.07 | 0.06 | 6.83 | 0.27 | 6.79 | 0.95 |
| 8 | 2.83 | 0.31 | 2.83 | 0.35 | 4.30 | 0.00 | 4.30 | 0.00 | 5.57 | 0.32 | 5.57 | 0.10 | 6.47 | 0.15 | 6.47 | 0.06 |
| 9 | 2.63 | 1.96 | 2.61 | 0.96 | 4.47 | 0.75 | 4.47 | 0.75 | 5.70 | 0.17 | 5.70 | 0.06 | 6.90 | 0.95 | 6.90 | 0.95 |
| 10 | 2.83 | 0.23 | 2.80 | 0.23 | 4.74 | 0.10 | 4.74 | 0.56 | 5.23 | 0.29 | 5.23 | 0.46 | 6.87 | 0.31 | 6.87 | 0.40 |
| 11 | 2.83 | 0.23 | 2.93 | 0.23 | 4.70 | 0.10 | 4.77 | 0.10 | 5.23 | 0.29 | 5.21 | 0.21 | 6.87 | 0.31 | 6.79 | 0.31 |
| 12 | 2.23 | 0.40 | 2.22 | 0.06 | 4.40 | 0.53 | 4.40 | 0.40 | 5.53 | 0.06 | 5.53 | 0.29 | 6.17 | 0.40 | 6.17 | 0.52 |
| 13 | 3.23 | 0.12 | 3.03 | 0.12 | 4.87 | 0.31 | 4.07 | 0.31 | 5.70 | 0.17 | 5.67 | 0.70 | 6.10 | 0.40 | 6.12 | 0.40 |
| 14 | 2.83 | 0.23 | 2.83 | 0.23 | 4.70 | 0.10 | 4.70 | 0.10 | 5.23 | 0.29 | 5.20 | 0.21 | 6.87 | 0.31 | 6.87 | 0.31 |
| 15 | 3.43 | 0.72 | 3.43 | 0.00 | 4.60 | 0.10 | 4.60 | 0.80 | 5.70 | 0.10 | 5.87 | 1.81 | 6.83 | 0.25 | 6.83 | 0.10 |
| 16 | 2.83 | 0.23 | 2.90 | 0.23 | 4.70 | 0.10 | 4.70 | 0.10 | 5.23 | 0.29 | 5.23 | 0.21 | 6.87 | 0.31 | 6.80 | 0.31 |
| 17 | 3.53 | 0.06 | 3.53 | 0.00 | 4.27 | 0.25 | 4.22 | 0.38 | 5.90 | 0.00 | 5.90 | 0.10 | 6.53 | 0.06 | 6.53 | 0.06 |
| 18 | 2.30 | 0.44 | 2.28 | 0.42 | 4.53 | 0.64 | 4.53 | 0.23 | 5.63 | 0.46 | 5.63 | 0.37 | 6.33 | 0.15 | 6.33 | 0.31 |
| 19 | 2.60 | 0.40 | 2.60 | 0.96 | 4.73 | 0.12 | 4.73 | 0.75 | 5.97 | 0.72 | 5.91 | 0.06 | 6.33 | 0.68 | 6.33 | 0.95 |
| 20 | 3.63 | 0.96 | 3.60 | 0.96 | 4.07 | 0.75 | 4.07 | 0.75 | 5.65 | 0.17 | 5.60 | 0.06 | 6.89 | 0.95 | 6.89 | 0.95 |
| 21 | 2.83 | 0.23 | 2.73 | 0.23 | 4.00 | 0.40 | 4.00 | 0.40 | 5.17 | 0.38 | 5.17 | 0.83 | 6.17 | 0.32 | 6.17 | 0.32 |
| 22 | 3.07 | 0.15 | 3.07 | 0.15 | 4.80 | 0.10 | 4.80 | 0.10 | 5.53 | 0.06 | 5.53 | 0.06 | 6.73 | 0.38 | 6.73 | 0.38 |
| 23 | 3.53 | 0.06 | 3.53 | 0.06 | 4.27 | 0.25 | 4.27 | 0.25 | 5.90 | 0.00 | 5.90 | 0.50 | 6.53 | 0.06 | 6.43 | 0.06 |
| 24 | 3.23 | 0.12 | 3.23 | 0.31 | 4.87 | 0.31 | 4.87 | 0.23 | 5.70 | 0.17 | 5.70 | 0.21 | 6.10 | 0.40 | 6.10 | 0.23 |
| 25 | 2.60 | 0.40 | 2.60 | 0.06 | 4.73 | 0.12 | 4.65 | 0.61 | 5.97 | 0.72 | 5.97 | 0.64 | 6.33 | 0.68 | 6.33 | 1.81 |
| 26 | 2.80 | 1.48 | 2.80 | 0.56 | 4.23 | 0.41 | 4.23 | 0.06 | 5.43 | 0.74 | 5.23 | 0.14 | 6.93 | 0.24 | 6.93 | 0.47 |
| 27 | 3.63 | 0.96 | 3.60 | 0.62 | 4.07 | 0.75 | 4.03 | 0.55 | 5.70 | 0.17 | 5.68 | 0.79 | 6.90 | 0.95 | 6.88 | 0.67 |
| 28 | 2.77 | 0.12 | 2.72 | 0.06 | 4.87 | 0.32 | 4.87 | 0.81 | 5.00 | 0.17 | 5.00 | 0.73 | 6.27 | 0.06 | 6.27 | 1.16 |
| 29 | 3.87 | 0.15 | 3.87 | 0.10 | 4.27 | 0.15 | 4.17 | 0.15 | 5.87 | 0.15 | 5.87 | 0.53 | 6.17 | 0.06 | 6.17 | 0.31 |
| 30 | 2.83 | 0.23 | 2.83 | 0.50 | 4.88 | 0.40 | 4.08 | 0.10 | 5.17 | 0.38 | 5.17 | 0.36 | 6.17 | 0.32 | 6.17 | 0.38 |
| 31 | 3.23 | 0.38 | 3.23 | 0.15 | 4.93 | 0.06 | 4.93 | 0.42 | 5.23 | 0.21 | 5.21 | 0.58 | 6.40 | 0.40 | 6.11 | 0.85 |
| mean | 2.99 | | 2.98 | | 4.53 | | 4.46 | | 5.48 | | 5.48 | | 6.60 | | 6.58 | |

Table 6.3. The mean value and SD for left and right foot with midstance phase (x-axis coordinate, mm)

6.3.2. Y-axis coordinate of midstance

The y-axis coordinate of midstance changes with movement for all 31 subjects. The maximum and minimum distributions of the left heel in the midstance phase are similar to those observed for the right foot, as shown in **Table 6.4**.

The greatest mean y-axis coordinate of values for left and right midstance were found in the SBW group (10, 20, 30 kg) compared with the y-axis coordinate in midstance phase for both left and right values with the NBW (0 kg) group.

The mean values of the maximum y-axis coordinate for both left and right feet in the SBW sets were 4.43 and 4.39 mm for the 10 kg; 5.42 and 5.40 mm for 20 kg, and 6.41 and 6.36 mm for 30 kg. The mean values for the y-axis coordinate, both left and right values, with for NBW were 3.05, 3.05 mm.

CHAPTER 6: RESULTS OF EXPERIMENT 3

| Table 6.4. The of mean value and SD for least sector. | ft and right foot in midstance | e phase (y- axis coordinate, mm) |
|---|--------------------------------|----------------------------------|
|---|--------------------------------|----------------------------------|

| | Midstance- Left and right foot Y-axis | | | | | | | | | | | | | | | |
|--------------|---------------------------------------|------|---------|------|---------|------|---------|------|---------|------|---------|------|---------|------|---------|------|
| Participants | | 01 | g | | | 10 | kg | | | 20 | kg | | | 30 | kg | |
| Id | Mean Lf | SD | Mean Rf | SD | Mean Lf | SD | Mean Rf | SD | Mean Lf | SD | Mean Rf | SD | Mean Lf | SD | Mean Rf | SD |
| 1 | 3.76 | 0.12 | 3.76 | 0.12 | 4.23 | 0.06 | 4.03 | 0.06 | 5.17 | 0.23 | 5.16 | 0.23 | 6.73 | 0.67 | 6.45 | 0.67 |
| 2 | 3.40 | 0.61 | 3.38 | 0.61 | 4.17 | 0.06 | 4.17 | 0.06 | 5.30 | 0.53 | 5.30 | 0.53 | 6.00 | 0.35 | 6.10 | 0.35 |
| 3 | 2.00 | 0.55 | 2.00 | 0.55 | 4.93 | 0.21 | 4.88 | 0.21 | 5.23 | 0.40 | 5.23 | 0.40 | 6.07 | 0.12 | 6.01 | 0.12 |
| 4 | 3.67 | 0.31 | 3.67 | 0.12 | 4.70 | 0.35 | 4.70 | 0.10 | 5.43 | 0.12 | 5.43 | 0.46 | 6.07 | 0.12 | 6.12 | 0.35 |
| 5 | 3.00 | 0.17 | 3.00 | 0.17 | 4.03 | 0.95 | 4.03 | 0.95 | 5.07 | 0.32 | 5.02 | 0.17 | 6.73 | 0.84 | 6.65 | 0.12 |
| 6 | 3.80 | 0.36 | 3.78 | 0.15 | 4.30 | 0.17 | 4.30 | 0.15 | 5.83 | 0.12 | 5.83 | 0.10 | 6.77 | 0.57 | 6.67 | 0.50 |
| 7 | 3.10 | 0.46 | 3.10 | 0.12 | 4.10 | 0.86 | 4.12 | 0.06 | 5.40 | 0.10 | 5.37 | 0.23 | 6.10 | 0.31 | 5.46 | 0.67 |
| 8 | 2.37 | 0.81 | 2.37 | 0.33 | 4.57 | 0.15 | 4.50 | 0.00 | 5.97 | 0.42 | 5.97 | 0.46 | 6.87 | 0.06 | 6.87 | 0.21 |
| 9 | 3.76 | 0.12 | 3.72 | 0.12 | 4.23 | 0.06 | 4.23 | 0.06 | 5.17 | 0.23 | 5.17 | 0.23 | 6.73 | 0.67 | 6.73 | 0.67 |
| 10 | 2.00 | 0.52 | 2.00 | 0.42 | 4.93 | 0.21 | 4.93 | 0.46 | 5.23 | 0.40 | 5.23 | 0.62 | 6.07 | 0.12 | 6.02 | 0.95 |
| 11 | 2.00 | 0.55 | 2.20 | 1.55 | 4.93 | 0.21 | 4.93 | 0.21 | 5.23 | 0.40 | 5.20 | 0.40 | 6.07 | 0.12 | 6.07 | 0.12 |
| 12 | 2.73 | 0.04 | 2.73 | 0.69 | 4.03 | 0.95 | 4.03 | 0.29 | 5.07 | 0.15 | 5.07 | 0.87 | 6.67 | 0.81 | 6.67 | 0.23 |
| 13 | 3.93 | 0.38 | 3.91 | 0.38 | 4.20 | 0.70 | 4.10 | 0.70 | 5.90 | 0.69 | 5.89 | 0.69 | 6.00 | 0.15 | 5.89 | 0.15 |
| 14 | 2.00 | 0.55 | 1.99 | 0.55 | 4.93 | 0.21 | 4.93 | 0.21 | 5.23 | 0.40 | 5.34 | 0.40 | 6.07 | 0.12 | 6.07 | 0.12 |
| 15 | 3.53 | 5.86 | 3.53 | 0.18 | 4.20 | 0.08 | 4.11 | 1.81 | 5.20 | 6.08 | 5.20 | 0.40 | 6.57 | 0.17 | 6.55 | 0.91 |
| 16 | 2.00 | 0.55 | 2.00 | 0.55 | 4.93 | 0.21 | 4.83 | 0.21 | 5.23 | 0.40 | 5.23 | 0.40 | 6.07 | 0.12 | 6.07 | 0.12 |
| 17 | 3.67 | 0.06 | 3.63 | 0.06 | 4.17 | 0.50 | 4.17 | 0.10 | 5.33 | 1.67 | 5.31 | 0.42 | 6.33 | 0.15 | 6.33 | 0.06 |
| 18 | 2.33 | 0.49 | 2.33 | 0.25 | 4.07 | 0.06 | 4.02 | 0.37 | 5.90 | 0.10 | 5.90 | 0.06 | 6.90 | 0.17 | 6.90 | 1.40 |
| 19 | 3.30 | 0.52 | 3.32 | 0.12 | 4.93 | 0.10 | 4.93 | 0.06 | 5.30 | 0.26 | 5.02 | 0.23 | 6.40 | 0.36 | 6.40 | 0.67 |
| 20 | 3.76 | 0.12 | 3.71 | 0.12 | 4.23 | 0.06 | 4.23 | 0.06 | 5.19 | 0.23 | 5.19 | 0.23 | 6.73 | 0.67 | 6.73 | 0.67 |
| 21 | 2.00 | 0.55 | 2.00 | 0.55 | 4.33 | 0.83 | 4.04 | 0.83 | 5.80 | 0.10 | 5.80 | 0.10 | 6.70 | 0.66 | 6.69 | 0.66 |
| 22 | 3.40 | 0.61 | 3.40 | 0.61 | 4.17 | 0.06 | 4.17 | 0.06 | 5.43 | 0.23 | 5.33 | 0.53 | 6.00 | 0.35 | 6.00 | 0.35 |
| 23 | 3.67 | 0.06 | 3.67 | 0.06 | 4.17 | 0.50 | 4.17 | 0.50 | 5.33 | 1.67 | 5.33 | 0.67 | 6.33 | 0.15 | 6.11 | 0.15 |
| 24 | 3.93 | 0.38 | 3.93 | 0.12 | 4.20 | 0.70 | 4.20 | 0.21 | 5.90 | 0.69 | 5.90 | 0.98 | 6.00 | 1.15 | 6.00 | 0.35 |
| 25 | 3.30 | 0.52 | 3.30 | 0.06 | 4.93 | 0.10 | 4.93 | 0.64 | 5.30 | 0.26 | 5.30 | 0.81 | 6.40 | 0.36 | 6.40 | 0.72 |
| 26 | 3.53 | 0.69 | 3.53 | 0.53 | 4.20 | 0.01 | 4.20 | 0.14 | 5.07 | 0.68 | 5.07 | 0.66 | 6.07 | 0.21 | 6.07 | 0.74 |
| 27 | 3.76 | 0.12 | 3.72 | 0.00 | 4.23 | 0.06 | 4.20 | 0.79 | 5.17 | 0.23 | 5.12 | 0.55 | 6.73 | 0.67 | 6.68 | 0.63 |
| 28 | 2.43 | 0.24 | 2.41 | 0.15 | 4.97 | 1.24 | 4.97 | 0.73 | 5.20 | 0.70 | 5.20 | 0.21 | 6.57 | 0.06 | 6.52 | 0.08 |
| 29 | 3.73 | 0.31 | 3.73 | 0.98 | 4.55 | 0.53 | 4.55 | 0.53 | 5.73 | 0.31 | 5.68 | 0.59 | 6.67 | 0.23 | 6.67 | 0.12 |
| 30 | 2.00 | 0.55 | 2.00 | 0.44 | 4.33 | 0.83 | 4.11 | 0.36 | 5.80 | 0.10 | 5.80 | 0.55 | 6.70 | 0.66 | 6.70 | 0.78 |
| 31 | 2.63 | 0.64 | 2.63 | 0.76 | 4.50 | 0.17 | 4.50 | 0.58 | 5.80 | 0.36 | 5.80 | 0.62 | 6.57 | 0.88 | 6.57 | 0.71 |
| mean | 3.05 | | 3.05 | | 4.43 | | 4.39 | | 5.42 | | 5.40 | | 6.41 | | 6.36 | |

6.4. The centroid of toe-off contact surface for simulated body weight

The objective was to investigate the influence of SBW on the participants' centre of toe-off zone by adding 10, 20, 30 kg. Samples of 31 adult males were tested. These subjects were grouped into four groups according to SBW and NBW: 0, 10, 20, and 30 kg. Participant characteristics are shown in **Table 3.3**.

6.4.1. X-axis coordinate of toe-off

The results show that the x-axis coordinate values increased at toe-off for the 31 subjects moving with simulated body weight. In all four groups (0, 10, 20, and 30 kg) left foot and right foot values were almost the same values for each condition. However, left foot and right foot values did show differences between NBW (0 kg) and SBW with 10, 20, and 30 kg weights. The results for x-axis coordinate value for both feet are given in **Table 6.5**.

Compared with the other simulated weights (20 and 10 kg) and the NBW (0 kg), the mean x-axis coordinate values under SBW of 30 kg loads were higher for both feet. The x-axis coordinate mean values for NBW (0 kg) and SBW of 10, 20, 30 kg for the left foot were 6.05, 7.97, 9.72, and 11.58 mm, respectively. X-axis coordinate mean values for the right foot were 6.04, 7.96, 9.71, and 11.57 mm, respectively based on the NBW (0 kg) and SBW of 10, 20, 30 kg.

CHAPTER 6: RESULTS OF EXPERIMENT 3

| | | | | | | Тое | -off- Left and | l right fo | ot X-axis | | | | | | | |
|--------------|---------|------|---------|------|---------|------|----------------|------------|-----------|------|---------|------|---------|-------|---------|------|
| Participants | | 01 | кg | | | 10 |)kg | | | 20 | kg | | | 30 | kg | |
| Id | Mean Lf | SD | Mean Rf | SD | Mean Lf | SD | Mean Rf | SD | Mean Lf | SD | Mean Rf | SD | Mean Lf | SD | Mean Rf | SD |
| 1 | 6.13 | 0.92 | 6.13 | 0.74 | 8.83 | 1.01 | 8.83 | 0.98 | 9.78 | 0.64 | 9.70 | 0.06 | 11.67 | 0.81 | 11.67 | 0.85 |
| 2 | 6.80 | 1.82 | 6.80 | 0.26 | 7.03 | 0.15 | 7.33 | 0.46 | 9.57 | 0.81 | 9.52 | 0.55 | 11.80 | 0.92 | 11.80 | 0.32 |
| 3 | 6.80 | 1.82 | 6.07 | 0.25 | 7.03 | 0.15 | 7.30 | 0.46 | 9.57 | 0.81 | 9.53 | 0.55 | 11.33 | 0.68 | 11.33 | 0.83 |
| 4 | 6.07 | 0.55 | 6.80 | 0.26 | 7.30 | 0.26 | 7.03 | 0.46 | 9.53 | 0.64 | 9.37 | 0.55 | 11.70 | 0.44 | 11.69 | 0.35 |
| 5 | 6.83 | 0.75 | 6.79 | 0.15 | 7.93 | 0.06 | 7.93 | 0.21 | 9.57 | 0.32 | 9.57 | 0.46 | 11.20 | 0.26 | 11.19 | 0.38 |
| 6 | 6.73 | 0.32 | 6.73 | 0.21 | 7.23 | 0.42 | 7.23 | 0.38 | 9.63 | 0.31 | 9.61 | 0.47 | 11.77 | 0.42 | 11.77 | 0.15 |
| 7 | 5.30 | 0.35 | 5.21 | 0.85 | 7.13 | 0.51 | 7.10 | 0.95 | 10.30 | 0.92 | 10.28 | 0.15 | 11.97 | 36.84 | 11.96 | 0.26 |
| 8 | 6.17 | 0.70 | 6.17 | 0.15 | 7.23 | 0.83 | 7.20 | 0.30 | 9.79 | 0.26 | 9.79 | 0.23 | 11.70 | 0.53 | 11.70 | 0.31 |
| 9 | 5.43 | 0.76 | 5.40 | 0.76 | 8.47 | 0.25 | 8.41 | 0.25 | 10.77 | 0.29 | 10.77 | 0.29 | 11.53 | 0.42 | 11.52 | 0.35 |
| 10 | 5.47 | 0.67 | 5.47 | 0.14 | 8.10 | 0.10 | 8.10 | 0.67 | 9.27 | 0.40 | 9.17 | 0.46 | 11.37 | 0.65 | 11.32 | 0.71 |
| 11 | 6.63 | 0.21 | 6.63 | 0.21 | 8.90 | 0.10 | 8.90 | 0.10 | 9.97 | 0.38 | 9.96 | 0.38 | 11.67 | 0.21 | 11.67 | 0.21 |
| 12 | 6.97 | 0.35 | 6.97 | 0.31 | 7.63 | 0.49 | 7.63 | 0.17 | 9.57 | 0.12 | 9.57 | 0.75 | 11.50 | 0.82 | 11.50 | 1.00 |
| 13 | 6.47 | 0.21 | 6.24 | 0.21 | 7.77 | 0.15 | 7.72 | 0.15 | 9.47 | 0.40 | 9.47 | 0.40 | 11.87 | 0.29 | 11.84 | 0.29 |
| 14 | 5.63 | 0.21 | 5.61 | 0.21 | 8.90 | 0.10 | 8.87 | 0.10 | 9.97 | 0.38 | 9.97 | 0.38 | 11.67 | 0.21 | 11.67 | 0.21 |
| 15 | 5.67 | 0.85 | 5.67 | 0.21 | 7.70 | 0.66 | 7.70 | 1.06 | 10.00 | 0.61 | 10.00 | 0.06 | 11.33 | 0.49 | 11.33 | 0.83 |
| 16 | 5.63 | 0.21 | 5.63 | 0.21 | 8.90 | 0.10 | 8.90 | 0.10 | 9.97 | 0.38 | 9.97 | 0.38 | 11.67 | 0.21 | 11.63 | 0.21 |
| 17 | 5.17 | 0.57 | 5.17 | 0.07 | 8.33 | 0.55 | 8.31 | 0.78 | 9.13 | 0.45 | 9.13 | 0.76 | 11.67 | 0.21 | 11.65 | 0.51 |
| 18 | 5.60 | 0.17 | 5.60 | 0.07 | 7.60 | 0.52 | 7.56 | 1.51 | 9.03 | 0.35 | 9.03 | 0.85 | 11.90 | 0.35 | 11.90 | 1.07 |
| 19 | 6.20 | 0.61 | 6.20 | 0.40 | 7.97 | 0.15 | 7.93 | 0.70 | 9.83 | 0.12 | 9.83 | 0.51 | 11.43 | 0.38 | 11.43 | 0.66 |
| 20 | 6.33 | 0.06 | 6.33 | 0.06 | 7.53 | 0.55 | 7.52 | 0.55 | 10.30 | 0.17 | 10.30 | 0.17 | 11.47 | 0.35 | 11.47 | 0.35 |
| 21 | 6.63 | 0.21 | 6.63 | 0.21 | 8.90 | 0.10 | 8.78 | 0.10 | 9.97 | 0.38 | 9.97 | 0.38 | 11.67 | 0.21 | 11.67 | 0.21 |
| 22 | 5.03 | 0.32 | 5.13 | 0.47 | 8.03 | 0.78 | 8.03 | 0.38 | 9.77 | 0.93 | 9.77 | 0.23 | 11.50 | 0.56 | 11.23 | 0.55 |
| 23 | 5.73 | 0.35 | 5.77 | 0.50 | 7.50 | 0.70 | 7.50 | 1.01 | 9.50 | 0.69 | 9.45 | 0.14 | 11.57 | 0.35 | 11.57 | 0.70 |
| 24 | 5.27 | 0.47 | 5.27 | 0.80 | 8.00 | 0.89 | 8.10 | 1.15 | 9.90 | 0.87 | 9.90 | 0.92 | 11.07 | 0.58 | 11.07 | 1.01 |
| 25 | 6.47 | 0.67 | 6.40 | 0.24 | 8.13 | 0.59 | 8.13 | 0.47 | 9.33 | 0.93 | 9.32 | 0.23 | 11.60 | 0.00 | 11.59 | 0.60 |
| 26 | 5.57 | 0.96 | 5.57 | 0.25 | 7.23 | 0.15 | 7.23 | 0.72 | 9.53 | 0.47 | 9.53 | 0.47 | 11.53 | 0.12 | 11.53 | 0.84 |
| 27 | 6.43 | 0.76 | 6.38 | 0.12 | 8.47 | 0.25 | 8.42 | 0.60 | 9.77 | 0.29 | 9.72 | 0.10 | 11.90 | 0.35 | 11.88 | 0.56 |
| 28 | 6.57 | 0.12 | 6.57 | 0.65 | 8.43 | 0.29 | 8.43 | 0.66 | 9.70 | 0.17 | 9.89 | 0.46 | 11.50 | 0.82 | 11.50 | 0.10 |
| 29 | 6.47 | 0.21 | 6.41 | 0.60 | 7.77 | 0.15 | 7.72 | 0.17 | 9.47 | 0.40 | 9.47 | 0.46 | 11.87 | 0.29 | 11.87 | 0.23 |
| 30 | 5.73 | 0.74 | 5.73 | 0.46 | 8.13 | 0.31 | 8.11 | 0.26 | 9.37 | 0.15 | 9.37 | 1.22 | 11.47 | 0.25 | 11.47 | 0.21 |
| 31 | 5.63 | 0.21 | 5.62 | 0.06 | 8.90 | 0.10 | 8.89 | 0.30 | 9.97 | 0.38 | 9.97 | 0.35 | 11.17 | 0.67 | 11.16 | 0.44 |
| mean | 6.05 | | 6.04 | | 7.97 | | 7.96 | | 9.72 | | 9.71 | | 11.58 | | 11.57 | |

Table 6.5. The mean value and SD for left and right foot in toe-off phase (x-axis coordinate, mm)

6.4.2. Y-axis coordinate of toe-off

The results show that, for all 31 subjects moving with simulated body weight, the yaxis coordinate values increased at toe off. In all four groups (SBW and NBW), left foot and right foot values were almost the same values for each condition (0, 10, 20, and 30 kg). However, left and right foot values did show the differences for NBW (0 kg) and simulated (10, 20, and 30 kg) weights. The results for y-axis coordinate values for both feet, are given in **Table 6.6**.

For both feet, the NBW group (0 kg) was lower at the y-axis coordinate mean values than the SBW group (10, 20, 30 kg). The y-axis coordinates mean values for NBW (0 kg) and SBW of 10, 20, 30 kg for the left foot were 5.84, 8.03, 9.40, and 11.53 mm respectively. In addition, y-axis coordinate mean value of the right foot were 5.83, 8.02, 9.39, and 11.52 mm, respectively for the NBW (0 kg) and SBW of 10, 20, 30 kg.

CHAPTER 6: RESULTS OF EXPERIMENT 3

Table 6.6. The mean value and SD for left and right foot in toe-off phase (y-axis coordinate, mm)

| | Toe-off- Left and right foot X-axis | | | | | | | | | | | | | | | |
|--------------|-------------------------------------|------|---------|------|---------|------|---------|------|---------|------|---------|------|---------|------|---------|------|
| Participants | | 01 | cg | | | 10 |)kg | | | 20 | kg | | | 30 | kg | |
| Id | Mean Lf | SD | Mean Rf | SD | Mean Lf | SD | Mean Rf | SD | Mean Lf | SD | Mean Rf | SD | Mean Lf | SD | Mean Rf | SD |
| 1 | 5.43 | 0.80 | 5.43 | 0.36 | 8.67 | 0.31 | 8.62 | 1.82 | 9.57 | 0.85 | 9.57 | 0.00 | 11.20 | 0.10 | 11.19 | 0.12 |
| 2 | 5.53 | 0.21 | 5.53 | 0.15 | 7.63 | 0.51 | 7.63 | 0.40 | 9.09 | 0.52 | 9.09 | 0.44 | 11.73 | 0.68 | 11.73 | 0.52 |
| 3 | 5.53 | 0.21 | 5.53 | 0.31 | 7.63 | 0.51 | 7.70 | 0.42 | 9.09 | 0.52 | 9.50 | 0.32 | 11.73 | 0.68 | 11.73 | 0.67 |
| 4 | 5.53 | 0.32 | 5.53 | 0.15 | 7.70 | 0.26 | 7.60 | 0.40 | 9.53 | 0.06 | 9.09 | 0.44 | 11.73 | 0.21 | 11.73 | 0.49 |
| 5 | 5.77 | 0.60 | 5.76 | 0.12 | 7.47 | 0.31 | 7.47 | 0.42 | 9.17 | 1.72 | 9.17 | 0.21 | 11.73 | 0.40 | 11.67 | 0.61 |
| 6 | 5.90 | 0.78 | 5.90 | 0.21 | 7.47 | 0.58 | 7.42 | 0.90 | 9.37 | 0.38 | 9.37 | 0.36 | 11.13 | 0.06 | 11.13 | 0.06 |
| 7 | 6.37 | 0.80 | 6.37 | 0.60 | 7.70 | 0.92 | 7.66 | 0.70 | 9.37 | 0.61 | 9.29 | 0.52 | 11.87 | 0.81 | 11.87 | 0.81 |
| 8 | 5.97 | 0.84 | 5.97 | 0.95 | 7.17 | 0.67 | 7.12 | 2.89 | 9.57 | 0.15 | 9.52 | 0.15 | 11.60 | 0.35 | 11.60 | 0.20 |
| 9 | 5.20 | 0.72 | 5.20 | 0.72 | 8.60 | 0.44 | 8.60 | 0.44 | 9.27 | 0.86 | 9.22 | 0.86 | 11.27 | 0.59 | 11.27 | 0.59 |
| 10 | 6.30 | 0.42 | 6.30 | 0.65 | 8.17 | 0.12 | 8.17 | 0.83 | 9.30 | 1.48 | 9.30 | 0.45 | 11.47 | 0.23 | 11.47 | 0.17 |
| 11 | 6.17 | 0.06 | 6.17 | 0.06 | 8.87 | 0.25 | 8.81 | 0.25 | 9.27 | 0.23 | 9.27 | 0.23 | 11.77 | 0.58 | 11.71 | 0.58 |
| 12 | 6.00 | 0.53 | 6.00 | 0.35 | 7.43 | 0.58 | 7.43 | 0.32 | 9.97 | 0.60 | 9.97 | 0.12 | 11.10 | 0.56 | 11.10 | 0.25 |
| 13 | 6.00 | 0.66 | 5.97 | 0.66 | 7.07 | 0.38 | 7.12 | 0.38 | 9.47 | 0.15 | 9.42 | 0.15 | 11.67 | 0.68 | 11.62 | 0.68 |
| 14 | 6.17 | 0.06 | 6.17 | 0.06 | 8.87 | 0.25 | 8.82 | 0.25 | 9.27 | 0.23 | 9.37 | 0.23 | 11.77 | 0.58 | 11.75 | 0.58 |
| 15 | 5.60 | 0.52 | 5.54 | 0.55 | 7.40 | 0.17 | 7.40 | 0.40 | 9.90 | 0.44 | 9.88 | 0.40 | 11.00 | 0.61 | 11.00 | 0.67 |
| 16 | 6.17 | 0.06 | 6.17 | 0.06 | 8.87 | 0.25 | 8.87 | 0.25 | 9.27 | 0.23 | 9.27 | 0.23 | 11.77 | 0.58 | 11.77 | 0.58 |
| 17 | 5.20 | 0.62 | 5.20 | 0.98 | 8.57 | 0.32 | 8.52 | 0.38 | 9.40 | 0.36 | 9.40 | 0.67 | 11.67 | 0.32 | 11.67 | 0.37 |
| 18 | 5.47 | 0.97 | 5.47 | 0.67 | 7.77 | 0.21 | 7.77 | 0.17 | 9.77 | 0.21 | 9.67 | 0.15 | 11.53 | 0.15 | 11.53 | 0.23 |
| 19 | 5.97 | 0.38 | 5.97 | 0.23 | 7.30 | 0.10 | 7.32 | 1.08 | 9.60 | 0.30 | 9.59 | 0.53 | 11.47 | 0.38 | 11.42 | 0.26 |
| 20 | 6.73 | 0.23 | 6.73 | 0.23 | 7.07 | 0.81 | 7.07 | 0.81 | 9.13 | 0.71 | 9.13 | 0.71 | 11.50 | 0.78 | 11.49 | 0.78 |
| 21 | 6.17 | 0.06 | 6.15 | 0.06 | 8.87 | 0.25 | 8.87 | 0.25 | 9.27 | 0.23 | 9.23 | 0.23 | 11.77 | 0.58 | 11.73 | 0.58 |
| 22 | 5.23 | 0.38 | 5.23 | 0.40 | 8.57 | 0.12 | 8.51 | 0.35 | 9.90 | 0.17 | 9.90 | 0.21 | 11.00 | 0.10 | 11.00 | 0.15 |
| 23 | 5.63 | 0.59 | 5.63 | 0.67 | 7.50 | 0.00 | 7.50 | 0.25 | 9.03 | 0.38 | 9.03 | 0.46 | 11.03 | 0.23 | 11.20 | 0.10 |
| 24 | 5.67 | 0.50 | 5.67 | 0.23 | 8.83 | 0.57 | 8.83 | 0.25 | 9.83 | 0.29 | 9.78 | 0.26 | 11.83 | 0.40 | 11.69 | 0.10 |
| 25 | 5.63 | 0.62 | 5.61 | 0.70 | 8.10 | 0.36 | 8.10 | 0.23 | 9.07 | 0.31 | 9.07 | 0.45 | 11.50 | 0.30 | 11.50 | 0.84 |
| 26 | 6.77 | 0.87 | 6.67 | 0.85 | 7.97 | 0.42 | 7.97 | 0.78 | 9.33 | 0.32 | 9.33 | 0.21 | 11.53 | 0.12 | 11.55 | 0.35 |
| 27 | 6.20 | 0.72 | 6.11 | 0.81 | 8.60 | 0.44 | 8.49 | 0.26 | 9.27 | 0.86 | 9.24 | 0.38 | 11.27 | 0.59 | 11.12 | 0.53 |
| 28 | 5.47 | 0.06 | 5.47 | 0.36 | 8.50 | 0.53 | 8.50 | 0.26 | 9.00 | 0.10 | 8.92 | 0.23 | 11.90 | 0.17 | 11.88 | 1.60 |
| 29 | 6.00 | 0.66 | 6.00 | 0.50 | 7.07 | 0.38 | 7.07 | 0.15 | 9.47 | 0.15 | 9.47 | 0.61 | 11.67 | 1.68 | 11.65 | 1.08 |
| 30 | 5.13 | 0.42 | 5.10 | 0.32 | 8.77 | 0.98 | 8.77 | 1.84 | 9.53 | 0.06 | 9.66 | 0.40 | 11.47 | 0.18 | 11.47 | 0.25 |
| 31 | 6.17 | 0.06 | 6.15 | 0.15 | 8.87 | 0.25 | 8.87 | 0.21 | 9.27 | 0.23 | 9.27 | 0.26 | 11.77 | 0.58 | 11.77 | 0.15 |
| mean | 5.84 | | 5.83 | | 8.03 | | 8.02 | | 9.40 | | 9.39 | | 11.53 | | 11.52 | |

6.5. Discussion

For the simulated body weight, a positive correlation was found between the increase in body weight and mean value for both x-axis and y-axis coordinates. The results of this experiment suggest that the increased weight by SBW can significantly increase x axis and y-axis coordinate mean values. So, the developed system does offer a new innovative technique for the calculation of angles and centroid of the area of contact. The results and analysis are limited to three significant stages of the load phase: 1) Heel strike, where the heel is in initial contact with the substrate. 2) Midstance or the instant in the gait cycle where the swing leg passes the support leg. 3) Toe-off or the instant before the plantar surface leaves the substrate.

Best on our knowledge, this is the first study to examine the relationship between dynamic foot alignment and centroid plantar pressure patterns in adults. The investigation used an insole system to capture the plantar surface data of 31 subjects with differing SBW. Participants were divided into three categories. Based on our best knowledge there is no research that studies simulated body weight to evaluate the x-location and y-location of the foot plantar during gait, thus, it is hard to compare these results with those of other researchers. Founded on the limited data for x-locational change, it seems that the NBW (0 kg) subjects' foot pressure compressed the soft plantar tissue more than the SBW of subjects (10, 20, 30 kg) when a load was applied.

6.6. Centroid of the foot plantar during gait

Changing the magnitude of the weight on the moving foot is a method that can help foot specialists to determine the cause of increased pressure of the plantar surface. In addition, the data can be used for medicine, sport and applications associated with footwear design. Thus, this research introduces a novel procedure to match increased body weight by simulation to determine the foot x-axis and y-axis coordinate values in three phases (heel strike, midstance and toe off).

6.6.1. Centroid of heel strike contact surface for the simulated body weight

This result indicates that increased body weight will lead to increased heel strike for both x-axis and y-axis coordinate mean values for both feet. The greater mean value for both coordinates (x-axis and y-axis) were with the 30kg, which were 12.59 and 12.18 mm for both feet.

6.6.2. The centroid of midstance contact surface for the simulated body weight

This result indicates that increased body weight will lead to an increase in both xaxis and y-axis coordinate mean value for both feet in midstance phase. The larger mean value for both coordinates (x-axis and y-axis) was 30kg for both feet, which were 6.59 and 6.90 mm.

6.6.3. The centroid of toe-off contact surface within the simulation body weight

This result indicates that increased body weight will lead to an increase in both xaxis and y-axis coordinate mean values for both feet. The largest mean values for both x-axis and y-axis coordinates were 30kg for both feet, which were 11.57 and 11.52 mm.

The results of the SBW indicate that a higher prevalence of overweight and obesity leads to an effect on the sole of the foot when other complicating agents, like the attendance of deformities and finite joint movability are controlled. The rationale for such an occurrence is that the soft tissue of overweight and obese subjects is constantly subjected to heavier weight thus having a small change in weight no longer had a greater impact.

As the data was normalised according to the total time taken for the loading phase of the gait, the y-locational change was due partly to the extra weight, which could increase the time of lifting the foot. Therefore, the results showed that the xlocate and y-locate change can help to calculate the change in the rotation of the ankle joint. The developed techniques could be used to demonstrate the effect of walking along hilly walkways.

6.7. Chapter summary

This chapter provides a better understanding of the connection between the simulated body weights and the centroid of the foot plantar during gait, and foot angles (heel strike angle and toe angle) for the loading phase of human gait. The participants were 31 males from a Southern Queensland University. They were instructed to walk freely, in a three-step gait trial on a level surface with a pressure system. Walking data was recorded by an insole capturing system.

The x and y-location means for the left and right sides were comparable and the combined values have presented the relationship between dynamic foot alignment and foot pressure patterns in participants with different loads.

The test data shows different mean x-axis and y-axis coordinate values under simulated body weight in all foot phases. The research results also show that the characteristics of the impact of body weight loading on the plantar surface varies between individuals. These conclusions allow the creation of a set of practical recommendations for future work that can be done in relation to the system. The utilisation of an innovative pre-marked medial foot reference and the advanced insole foot pressure sensor have enabled the demonstration of the effect of overweight on ankle motion. Preliminary tests show that the technique is practical for clinical studies and demonstration. Further testing will be needed to establish the suitability of the techniques for developing an indexing scheme for the study of ankle injury in obese people.

The next chapter presents a summary and final conclusion for the thesis and identifies the possibilities for further research.

7. CONCLUSIONS

7.1. Introduction

This chapter summarises the overall conclusions of this research. Detailed conclusions corresponding to the experiments and the result of each objective can be found in **chapter 4**, **to chapter 6**, respectively. Based on these conclusions, a set of practical recommendations is provided in **chapter 7**.

Advancing the knowledge of human foot function and bodyweight loading impact on foot performance is a crucial area of human health research. This knowledge is particularly important for the study and treatment of pathological injured and deformed feet. Therefore, a number of investigations were carried out to determine whether the continuous capture by an insole system during walking could assist in understanding the loading characteristics and structural behaviour of the foot. The study discusses the various methods and the approaches used for achieving the effects of body weight or simulation of body weight and correlation. The findings of these studies are summarised below. The study discusses the various methods and the approaches used for achieving this correlation. Three objectives were established to accomplish the research aim. These objectives were: 1) assessing dynamic foot plantar pressure characteristics in adults, who were of normal weight, overweight and obese, during level walking using the insole pressure system, and then comparing those values (findings) with those of a healthy control group, 2) studying the impact of simulated body weight (SBW) to increase the BMI from healthy to overweight and obese on gait measures, by comparing large simulated BMI (wearing a vest with 10, 20, and 30 kg weight) with the control group (who was tested without any weight or un-simulated body weight group, NBW), and 3) evaluating the spatial relationship between the trace of the centroid of the area of contact with heel strike, midstance, and toe-off phases with simulated body weight (SBW) groups, using the in-shoe plantar pressure system. Below are the conclusions of the study based on each research objective.

Objective 1:

This objective was achieved by classifying participants into three groups (healthy, overweight and obese) according to BMI to capture plantar foot pressure. Results derived from this research show that the obese group has identified significant impact compare to the other two groups: healthy and overweight within the foot plantar pressure parameter features (F, CA, CP, PTI, and PP) within ten different foot regions. It has been noted that the highest regions of F measures were found with the obese and overweight groups under the H, 1MH, 1stT, followed by the 2MH when compared to the controls. CA was highest under the H, 1MH, MF, and 2MH areas. CP was higher beneath the 2MH, 3MH, H, 1stT, and 1MH regions. PTI was highest under the H, 1stT, 2MH, and 3MH regions. And PP was highest under the H, 1stT, and 2MH areas. The forefoot appears to be more sensitive to weight-related pressure under the foot than the rearfoot. Results also suggest that obese participants, to higher BMI, experienced increased plantar foot pressure when compared to healthy weight and overweight participants because obese participants have greater body mass. Notably, the feet of obese individuals are wider than those of healthy and overweight adults. Findings from this study indicate that being overweight or obese increases foot pressure measures even for individuals with similar body features. These findings have implications for pain and discomfort in the lower extremity in the obese, and in the predisposition to participate in activities of daily living, such as walking.

Objective 2:

To accomplish this objective, an additional weight was added to increase a participant's body weight (10, 20, 30 kg), to foot plantar measures. This method allows a reduction in resources, such as time for increasing or reducing participants' weights. The results derived from the SBW showed that SBW is positively correlated to foot plantar measure values compared to the NBW group. Additional weight added (SBW) to increase the participants' body weight allowing them to reach a higher BMI), indicated that there were statistically significant increases in force measure between the SBW (10, 20, 30 kg) compared with the NBW group within the H, 1MH, 2MH, and MF regions, respectively. And, for contact area, H, 1MH, and MF were three main regions that made the largest CA measures with SBW (30, 20,

CHAPTER 7: CONCLUSIONS

10 kg). Moreover, 2MH, H, and 3MH displayed statistically significant increases in mean CP value in all SBW sessions. In addition, H, 2MH, and 3MH regions displayed statistically significant increases in mean PTI in the SBW groups compared to the NSBW group. Furthermore, a statistically significant increase in PP between the 30, 20, 10 kg and NSBW was evident in all other regions. The regions of 3rd-5thT and 2ndT, in contracts did not display any statistically significant differences in changes with SBW compared to the NSBW group in which all five most clinically relevant parameters were obtained: F, CA, CP, PTI and PP. The values or results of CV and ICC for analysis showed a generally good level of reliability, the quality of which is dependent on the region of the foot and the variable investigated with SBW loads.

Objective 3:

To achieve the last objective, the centroid trace of the plantar contact area of thirtyone subjects (identification: 1 to 31) were compared the heel strike, midstance, and toe-off phases with SBW groups to the other NBW group, using the in-shoe plantar pressure system. As discussed, this objective provides an advanced understanding of the spatial relationship between the area of contact during heel strike, midstance, and toe-off phases, noting the SBW groups of human gait. This objective also provides a better understanding of the plantar soft tissue mechanics during walking. The tests showed that the technique is practical for both a clinical study and clinical demonstration. The centroid coordinates x-axis and y-axis with coordinate values have demonstrated a significant increase in simulated bodyweight groups within the three stages of the load phase: heel strike, midstance; and toe-off, for both feet. A change in the magnitude of the load on the moving foot is a method that can help foot specialists determine the cause of increased pressure on the plantar surface.

7.2. Overall conclusions

Overweight and obesity are growing problems throughout the world and adults suffering from these conditions are particularly at risk of suffering health issues. Furthermore, the associated comorbidities of obesity are increasingly evident.

CHAPTER 7: CONCLUSIONS

Biomedical, physiotherapy and clinical studies show that humans can be distinguished by unique limb movement patterns and plantar pressure patterns if they are obese or non-obese. Being overweight and obese restrict mobility and physical exercise, which are essential for maintaining good health. Body weight is the main factor for increasing results in the plantar pressures foot measure. Advancing the knowledge of human foot function and body weight loading impact on foot performance has been a crucial area of research into human health. This knowledge is particularly important for the study and treatment of pathologically injured and deformed feet. The pressure system is becoming popular for capturing accurate plantar pressure data for overweight, obese and healthy adults and children.

Presently, there are few studies on the alterations in gait features of overweight and obese adults with small differences in BMI over a short time, particularly precision tracking of the lower limbs, including ankles and feet. The purpose of this study is to determine the relationship between BMI and gait characteristics with the aim of evaluating the relationship between lower limb and foot biomechanical measures and plantar pressure measures in adults aged 24 to 50. Changes in plantar pressure distribution parameters were evaluated using an insole shoe system during comfortable walking speed under simulated body weight conditions. Clinically, force, contact area, contact pressure, plantar pressure integral and peak pressure are the foot parameters that are most relied upon.

Pressure distribution measurement techniques are useful in analysing the mechanical behaviour of the human foot during dynamic loading situations in adult subjects. Hence, there is a need for a complete analysis of the variation of pressure distribution patterns under both feet while body weight is simulated through three different weights. Additionally, it is important to make and analyse a comparison of these weight pressure distribution patterns under the foot.

7.3. Limitations of the research

As with any experimental study, due to the operating conditions and laboratory environment, this study has some limitations, which should be acknowledged. These are:

CHAPTER 7: CONCLUSIONS

- 1. Although all testing was conducted within university hours (9:00 am 4:00 pm), results may have varied among adults due to differences in the time of day the assessments were conducted;
- 2. Participants were selected from the available population at university;
- The tests were conducted in a laboratory setting which does not represent the gait environment of the real world;
- Male participants presented at various body weights (68-142.5) kg and BMI (20.1- 46.0 kg/m²);
- 5. The participants recruited for these studies were aged between 24 to 50 years;
- 6. These findings are restricted to a generalisation for adults within this age range;
- 7. Participants walked at self-selected speeds;
- 8. In-shoe techniques can only provide information about forefoot conditions. Therefore, the results cannot reveal the impact of the barefoot plantar surface;
- Walking at speed and running were not tested during this investigation due to limitations;
- 10. The current study examined and provided results for males, so these conditions cannot be generalized to females;
- 11. Despite familiarizing the adults with each protocol, adults may have altered the way they typically performed a task because they were aware that they were being assessed;
- 12. Participants were wearing vests with different weights (10, 20 and 30kg) which was a limitation that could have influenced the gait parameters while walking;
- 13. Limited analyses of the GRF and PP were available for the determination of stress and strain on the foot.

7.4. Future research

The topics of this study may open doors for new and further investigations. As an extension of this research, investigations could include the determination of the plantar surface tendency at different walking speeds and the development of effective methods of monitoring a large sample of the diabetic population in both urban and rural communities. The investigation might supply vital information about the characteristics of the underlying plantar muscles for high surface loading.

Further work could also include developing a comprehensive human plantar evaluation system utilising plantar 3D ultrasound data photorealistic textured 3D plantar surface and plantar surface pressure data of the dynamic foot during gait. Photographic examination may reveal attributes of the foot that could cause increased loading of the plantar soft tissue and skin breakdown.

Further research is required to investigate the relationship between obesity and foot structure during different walking speeds. In future studies, we will investigate a larger number of subjects and will also consider foot arch height and postural sway of all subjects to understand plantar pressure distribution variations on both feet due to external perturbation. Moreover, this would provide a direct measure of soft tissue features such as plantar fascia length, elongation/strain, and 3D shape changes and, as a consequence, would enable direct assessment of in vivo mechanical properties. It is necessary to examine factors that may influence the function of the quadriceps and walking speed and to longitudinally evaluate the influence of the foot.

The following recommendations for further study depend on the results of the current study suggesting that overweight and obesity have a negative effect on adults. Whether the increased pressures are associated with foot discomfort to prevent an obese adult from participating in physical activity requires further investigation. In addition, further investigation of the effects of obesity on the musculoskeletal structure and biomechanical function of young adults is also suggested. Encouraging overweight and obese individuals to lose weight even by a modest amount, is important to improve their healthy status. The decrease in body weight can significantly reduce the risk of developing neuropathic foot ulcers through the reduction of pressure. Noticeably, this association can play a significant role in the diabetic population.

REFERENCES

- Aaboe, J., Bliddal, H., Messier, S.P., Alkjaer, T. and Henriksen, M., 2011. Effects of an intensive weight loss program on knee joint loading in obese adults with knee osteoarthritis. Osteoarthritis and Cartilage, 19(7), pp.822-828.
- Abboud, R.J., 2002. (i) Relevant foot biomechanics. *Current Orthopaedics*, 16(3), pp.165-179.
- ABo, S., 2015. National Health Survey: First Results, 2014–2015. Canberra: ACT: Australian Bureau of Statistics (ABS).
- Adesida, Y., Papi, E. and McGregor, A.H., 2019. Exploring the Role of Wearable Technology in Sport Kinematics and Kinetics: A Systematic Review. Sensors, 19(7), p.1597.
- Agrawal, S.K., Banala, S.K., Fattah, A., Sangwan, V., Krishnamoorthy, V., Scholz, J.P. and Hsu, W.L., 2007. Assessment of motion of a swing leg and gait rehabilitation with a gravity balancing exoskeleton. *IEEE Transactions on Neural Systems and Rehabilitation Engineering*, 15(3), pp.410-420.
- Alaqtash, M., Yu, H., Brower, R., Abdelgawad, A. and Sarkodie-Gyan, T., 2011. Application of wearable sensors for human gait analysis using fuzzy computational algorithm. *Engineering Applications of Artificial Intelligence*, 24(6), pp.1018-1025.
- AL-Baghdadi, J.A.A., Chong, A.K. and Milburn, P.D., 2015. Fabrication and testing of a low-cost foot pressure sensing system. *Proceedings of the 3rd International Conference on Industrial Application Engineering 2015 at Kitakyushu, Japan.*
- Al-Baghdadi, J.A.A., Chong, A.K., McDougall, K., Alshadli, D., Milburn, P. and Newsham-West, R., 2011. A dense surface modelling technique for foot surface imaging. In Proceedings of the Surveying and Spatial Sciences Biennial Conference (SSSC 2011) (pp. 295-302). Scion.
- Alexander, I.J., Chao, E.Y. and Johnson, K.A., 1990. The assessment of dynamic footto-ground contact forces and plantar pressure distribution: a review of the evolution of current techniques and clinical applications. *Foot & ankle*, 11(3), pp.152-167.
- Al-Magsoosi, S.K. and Chong, A.K., 2019, March. Advanced Foot Pressure Technology for Demonstrating the Influence of Over-weight and Obesity on Ankle Joint. In Proceedings of the 2019 2nd International Conference on Information Science and Systems (pp. 195-199). ACM.
- Amiri, P., Hubley-Kozey, C.L., Landry, S.C., Stanish, W.D. and Wilson, J.A., 2015. Obesity is associated with prolonged activity of the quadriceps and gastrocnemii during gait. *Journal of Electromyography and Kinesiology*, 25(6), pp.951-958.
- Anandacoomarasamy, A., Caterson, I., Sambrook, P., Fransen, M. and March, L., 2007. The impact of obesity on the musculoskeletal system. *Internal Medicine Journal*, 37, p.A35.
- Antunes, P.J., Dias, G.R., Coelho, A.T., Rebelo, F. and Pereira, T., 2008. Non-linear finite element modelling of anatomically detailed 3D foot model. *Report paper*, pp.1-11.
- Armstrong, D.G., Lavery, L.A., Vela, S.A., Quebedeaux, T.L. and Fleischli, J.G., 1998. Choosing a practical screening instrument to identify patients at risk for diabetic foot ulceration. *Archives of internal medicine*, 158(3), pp.289-292.

- Arndt, A., Westblad, P., Winson, I., Hashimoto, T. and Lundberg, A., 2004. Ankle and subtalar kinematics measured with intracortical pins during the stance phase of walking. Foot & ankle international, 25(5), pp.357-364.
- Arts, M.L.J. and Bus, S.A., 2011. Twelve steps per foot are recommended for valid and reliable in-shoe plantar pressure data in neuropathic diabetic patients wearing custom made footwear. Clinical biomechanics, 26(8), pp.880-884.
- Australian Government (2017). "A picture of overweight and obesity in Australia." Retrieved 10 Aug, 2018, from https://www.aihw.gov.au/getmedia/172fba28-785e-4a08-ab37-2da3bbae40b8/aihw-phe-216.pdf.aspx?inline=true.
- Australian Institute of Health and Welfare 2016. Australia's health 2016. Australia's health series no. 15. Cat. no. AUS 199. *Canberra*: AIHW.
- Australian Institute of Health and Welfare AIHW 2017. Weight loss surgery in Australia 2014–15: Australian hospital statistics. Cat. no. HSE 186.Canberra: AIHW.
- Awale, A., Dufour, A.B., Katz, P., Menz, H.B. and Hannan, M.T., 2016. Link between foot pain severity and prevalence of depressive symptoms. *Arthritis care & research*, 68(6), pp.871-876.
- Aydos, L., Uzun, A., Kaya, M., Kanatli, U., Esen, E. and Uslu, S., 2012. The effect of volleyball on the sole contact areas and maximal forces of Female volleyball players. *Scientific Journal Of Sport And Physical Education*, p.4.
- Ayyappa, E., 1997. Normal human locomotion, part 1: Basic concepts and terminology. *JPO: Journal of Prosthetics and Orthotics*, 9(1), pp.10-17.
- Baan, H., Dubbeldam, R., Nene, A.V. and van de Laar, M.A., 2012, June. Gait analysis of the lower limb in patients with rheumatoid arthritis: a systematic review. In *Seminars in arthritis and rheumatism* (Vol. 41, No. 6, pp. 768-788). WB Saunders.
- Badlissi, F., Dunn, J.E., Link, C.L., Keysor, J.J., McKinlay, J.B. and Felson, D.T., 2005. Foot musculoskeletal disorders, pain, and foot-related functional limitation in older persons. *Journal of the American Geriatrics Society*, 53(6), pp.1029-1033.
- Bamberg, S.J.M., Benbasat, A.Y., Scarborough, D.M., Krebs, D.E. and Paradiso, J.A., 2008. Gait analysis using a shoe-integrated wireless sensor system. *IEEE* transactions on information technology in biomedicine, 12(4), pp.413-423.
- Bankoff, A.D.P., Zamai, C.A., Schimdt, A., Ciol, P. and Barros, D.D., 2003. Estudo das alterações morfológicas do sistema locomotor: postura corporal x obesidade. *Journal of Physical Education*, 14(2), pp.41-48.
- Barisch-Fritz, B., Schmeltzpfenning, T., Plank, C., Hein, T. and Grau, S., 2014. The effects of gender, age, and body mass on dynamic foot shape and foot deformation inchildren and adolescents. *Footwear Science*, 6(1), pp.27-39.
- Battaglia, P.J., Maeda, Y., Welk, A., Hough, B. and Kettner, N., 2014. Reliability of the Goutallier classification in quantifying muscle fatty degeneration in the lumbar multifidus using magnetic resonance imaging. *Journal of manipulative* and physiological therapeutics, 37(3), pp.190-197.
- Becker, H., Rosenbaum, D., Claes, L. and Gerngro, H., 1997. Measurement of plantar pressure distribution during gait for diagnosis of functional lateral ankle instability. *Clinical biomechanics (Bristol, Avon)*, 12(3), pp.S19-S19.
- Bedi, H.S. and Love, B.R., 1998. Differences in impulse distribution patterns in patients with plantar fasciitis. *Foot & ankle international*, 19(3), pp.153-156.

- Bell, J.A., Kivimaki, M. and Hamer, M., 2014. Metabolically healthy obesity and risk of incident type 2 diabetes: a meta-analysis of prospective cohort studies. *Obesity reviews*, *15*(6), pp.504-515.
- BenAbdelkader, C. and Davis, L., 2002, May. Detection of people carrying objects: A motion-based recognition approach. In *Proceedings of Fifth IEEE International Conference on Automatic Face Gesture Recognition* (pp. 378-383). IEEE.
- Bennett, P.J. and Duplock, L.R., 1993. Pressure distribution beneath the human foot. *Journal of the American Podiatric Medical Association*, 83(12), pp.674-678.
- Birrell, S.A. and Haslam, R.A., 2009. The effect of military load carriage on 3-D lower limb kinematics and spatiotemporal parameters. *Ergonomics*, 52(10), pp.1298-1304.
- Birtane, M. and Tuna, H., 2004. The evaluation of plantar pressure distribution in obese and non-obese adults. *Clinical Biomechanics*, 19(10), pp.1055-1059.
- Bland, J.M. and Altman, D., 1986. Statistical methods for assessing agreement between two methods of clinical measurement. *The lancet*, *327*(8476), pp.307-310.
- Błaszczyk, J.W., Plewa, M., Cieślinska-Swider, J., Bacik, B., Zahorska-Markiewicz,
 B. and Markiewicz, A., 2011. Impact of excess body weight on walking at the preferred speed. *Acta neurobiologiae experimentalis*, 71(4), pp.528-540.
- Bosch, K., Gerss, J. and Rosenbaum, D., 2007. Preliminary normative values for foot loading parameters of the developing child. Gait & posture, 26(2), pp.238-247.
- Bosch, K., Nagel, A., Weigend, L. and Rosenbaum, D., 2009. From "first" to "last" steps in life–pressure patterns of three generations. Clinical Biomechanics, 24(8), pp.676-681.
- Bosy-Westphal, A. and Müller, M.J., 2015. Identification of skeletal muscle mass depletion across age and BMI groups in health and disease—there is need for a unified definition. *International journal of obesity*, *39*(3), p.379.
- Boulgouris, N.V., Hatzinakos, D. and Plataniotis, K.N., 2005. Gait recognition: a challenging signal processing technology for biometric identification. *IEEE* signal processing magazine, 22(6), pp.78-90.
- Boulton, A.J., 2014. Diabetic neuropathy and foot complications. In *Handbook of clinical neurology* (Vol. 126, pp. 97-107). Elsevier.
- Browning, R.C. and Kram, R., 2007. Effects of obesity on the biomechanics of walking at different speeds. *Medicine & Science in Sports & Exercise*, 39(9), pp.1632-1641.
- Browning, R.C., McGowan, C.P. and Kram, R., 2009. Obesity does not increase external mechanical work per kilogram body mass during walking. *Journal of biomechanics*, 42(14), pp.2273-2278.
- Brownrigg, J.R.W., Apelqvist, J., Bakker, K., Schaper, N.C. and Hinchliffe, R.J., 2013. Evidence-based management of PAD & the diabetic foot. *European Journal of Vascular and Endovascular Surgery*, 45(6), pp.673-681.
- Bryant, A., Singer, K. and Tinley, P., 1999. Comparison of the reliability of plantar pressure measurements using the two-step and midgait methods of data collection. *Foot & ankle international*, 20(10), pp.646-650.
- Bryant, A.R., Tinley, P. and Singer, K.P., 2000. Normal values of plantar pressure measurements determined using the EMED-SF system. *Journal of the American Podiatric Medical Association*, *90*(6), pp.295-299.
- Bryant, J., Stevenson, J., Pelot, R., Morin, E. and Deakin, J., 2005. Development of a Dynamic Biomechanical Model for Load Carriage: Phase 4, Part C3: Dynamic

Assessment of Pressure Measurement Systems for Use in Human Load Carriage, Queen's Univ Kingston (Ontario). *Ergonomics Research Group, Toronto*, pp.1-12.

- Burnfield, M., 2010. Gait analysis: normal and pathological function. *Journal of Sports Science and Medicine*, 9(2), p.353.
- Burns, J., Crosbie, J., Hunt, A. and Ouvrier, R., 2005. The effect of pes cavus on foot pain and plantar pressure. *Clinical Biomechanics*, 20(9), pp.877-882.
- Bus, S.A. and de Lange, A., 2005. A comparison of the 1-step, 2-step, and 3-step protocols for obtaining barefoot plantar pressure data in the diabetic neuropathic foot. *Clinical biomechanics*, 20(9), pp.892-899.
- Bus, S.A. and Waaijman, R., 2013. The value of reporting pressure-time integral data in addition to peak pressure data in studies on the diabetic foot: a systematic review. *Clinical Biomechanics*, 28(2), pp.117-121.
- Bus, S.A., Haspels, R. and Busch-Westbroek, T.E., 2011. Evaluation and optimization of therapeutic footwear for neuropathic diabetic foot patients using in-shoe plantar pressure analysis. Diabetes care, 34(7), pp.1595-1600.
- Bus, S.A., Maas, M., de Lange, A., Michels, R.P. and Levi, M., 2005. Elevated plantar pressures in neuropathic diabetic patients with claw/hammer toe deformity. *Journal of biomechanics*, 38(9), pp.1918-1925.
- Butler, R.J., Crowell III, H.P. and Davis, I.M., 2003. Lower extremity stiffness: implications for performance and injury. *Clinical biomechanics*, *18*(6), pp.511-517.
- Butterworth, P.A., Landorf, K.B., Smith, S.E. and Menz, H.B., 2012. The association between body mass index and musculoskeletal foot disorders: a systematic review. *Obesity reviews*, *13*(7), pp.630-642.
- Butterworth, P.A., Urquhart, D.M., Landorf, K.B., Wluka, A.E., Cicuttini, F.M. and Menz, H.B., 2015. Foot posture, range of motion and plantar pressure characteristics in obese and non-obese individuals. *Gait & posture*, 41(2), pp.465-469.
- Castano, L.M. and Flatau, A.B., 2014. Smart fabric sensors and e-textile technologies: a review. *Smart Materials and Structures*, 23(5), p.053001.
- Castro, M., Abreu, S., Sousa, H., Machado, L., Santos, R. and Vilas-Boas, J.P., 2013. Ground reaction forces and plantar pressure distribution during occasional loaded gait. *Applied ergonomics*, 44(3), pp.503-509.
- Castro, M., Abreu, S., Sousa, H., Machado, L., Santos, R. and Vilas-Boas, J.P., 2013. In-shoe plantar pressures and ground reaction forces during obese adults'overground walking. *Biomechanics of The Loaded Gait (Backpackers & Obese People), Testing of Pressure Relief Insoles and Reliability of A New Gait Analysis Device*, p.33.
- Castro, M., Soares, D. and Machado, L., 2011. Comparison of vertical GRF obtained from force plate, pressure plate and insole pressure system. In *International Society of Biomechanics in Sports (ISBS)-Conference Proceedings Archive* (Vol. 1, No. 1).
- Castro, M.P.D., Meucci, M., Soares, D.P., Fonseca, P., Borgonovo-Santos, M., Sousa, F., Machado, L. and Vilas-Boas, J.P., 2014. Accuracy and repeatability of the gait analysis by the WalkinSense system. *BioMed research international*, 2014.
- Cau, N., Cimolin, V., Galli, M., Precilios, H., Tacchini, E., Santovito, C. and Capodaglio, P., 2014. Center of pressure displacements during gait initiation in

individuals with obesity. *Journal of neuroengineering and rehabilitation*, *11*(1), p.82.

- Cavanagh, P.R. and Ulbrecht, J.S., 1994. Clinical plantar pressure measurement in diabetes: rationale and methodology. *The foot*, 4(3), pp.123-135.
- Cavanagh, P.R., Hennig, E.M., Rodgers, M.M. and Sanderson, D.J., 1985. The measurement of pressure distribution on the plantar surface of diabetic feet. *Biomechanics*, pp.159-166.
- Cavanagh, P.R., Hewitt Jr, F.G. and Perry, J.E., 1992. In-shoe plantar pressure measurement: a review. *The foot*, 2(4), pp.185-194.
- Centers for Disease Control and Prevention (CDC, 2011. Prevalence of obesity among adults with arthritis---United States, 2003--2009. *MMWR*. *Morbidity and mortality weekly report*, 60(16), p.509.
- Centers for Disease Control and Prevention (Centers for Disease Control and Prevention) 2015. Strategies to prevent obesity. Atlanta: CDC.Viewed 23 June 2017, <www.cdc.gov/obesity/strategies>.
- Centers for Disease Control and Prevention. Centers for Disease Control and Prevention, 20 Nov. 2014. Web. 5 Dec. 2014.
- Chakravarthy, Y.K., Ranganayakulu, A.B.R., Shanmuk, D., Narasimha, V.L., Praveen, J.V. and Srinath, A., 2017. Data capturing techniques for gait analysis using virtual interface. *International Journal of Pure and Applied Mathematics*, 116(5), pp.45-50.
- Chang, A., Hayes, K., Dunlop, D., Hurwitz, D., Song, J., Cahue, S., Genge, R. and Sharma, L., 2004. Thrust during ambulation and the progression of knee osteoarthritis. *Arthritis & Rheumatism*, 50(12), pp.3897-3903.
- Chang, B.C., Wang, J.Y., Huang, B.S., Lin, H.Y. and Lee, W.C., 2012. Dynamic impression insole in rheumatoid foot with metatarsal pain. *Clinical biomechanics*, 27(2), pp.196-201.
- Chang, J.H., Wang, S.H., Kuo, C.L., Shen, H.C., Hong, Y.W. and Lin, L.C., 2010. Prevalence of flexible flatfoot in Taiwanese school-aged children in relation to obesity, gender, and age. *European journal of pediatrics*, 169(4), pp.447-452.
- Chao, C.Y., Zheng, Y.P. and Cheing, G.L., 2011. Epidermal thickness and biomechanical properties of plantar tissues in diabetic foot. *Ultrasound in medicine & biology*, *37*(7), pp.1029-1038.
- Cheng, H.Y.K., Lin, C.L., Wang, H.W. and Chou, S.W., 2008. Finite element analysis of plantar fascia under stretch—The relative contribution of windlass mechanism and Achilles tendon force. *Journal of biomechanics*, *41*(9), pp.1937-1944.
- Chesnin, K.J., Selby-Silverstein, L. and Besser, M.P., 2000. Comparison of an in-shoe pressure measurement device to a force plate: concurrent validity of center of pressure measurements. *Gait & posture*, *12*(2), pp.128-133.
- Chevalier, T.L., Hodgins, H. and Chockalingam, N., 2010. Plantar pressure measurements using an in-shoe system and a pressure platform: A comparison. *Gait & posture*, 31(3), pp.397-399.
- Chong, A.K. and Milburn, P., 2017. Push-off apparent time delay based on COP trajectory of gait. *Footwear Science*, 9(sup1), pp.S54-S56.
- Chong, A.K., Al-Baghdadi, J.A. and Milburn, P., 2015. Human Plantar Pressure Image and Foot Shape Matching. *Journal of Biosciences and Medicines*, *3*(06), p.36.
- Chong, A.K., Alrikabi, R. and Milburn, P., 2017, July. Gait COP trajectory of left side hip-dislocation and scoliotic patient using ankle-foot orthoses. In *Ninth International Conference on Digital Image Processing (ICDIP 2017)* (Vol. 10420, p. 104204S). International Society for Optics and Photonics.

- Chong, A.K., Alshadli, D., AL-Baghdadi, J.A.A., Milburn, P. and Newsham-West, R., 2013, July. A video-based system for plantar surface acquisition during gait. In *Proceedings of the 31st International Conference on Biomechanics in Sports* (ISBS 2013) (pp. 1-4). National Taiwan Normal University.
- Chow, D.H., Kwok, M.L., Au-Yang, A.C., Holmes, A.D., Cheng, J.C., Yao, F.Y. and Wong, M.S., 2005. The effect of backpack load on the gait of normal adolescent girls. *Ergonomics*, 48(6), pp.642-656.
- Chuckpaiwong, B., Berkson, E.M. and Theodore, G.H., 2009. Extracorporeal shock wave for chronic proximal plantar fasciitis: 225 patients with results and outcome predictors. *The journal of foot and ankle surgery*, 48(2), pp.148-155.
- Chuckpaiwong, B., Nunley, J.A., Mall, N.A. and Queen, R.M., 2008. The effect of foot type on in-shoe plantar pressure during walking and running. *Gait & posture*, 28(3), pp.405-411.
- Clare, H.A., Adams, R. and Maher, C.G., 2003. Reliability of detection of lumbar lateral shift. *Journal of manipulative and physiological therapeutics*, 26(8), pp.476-480.
- Coda, A. and Santos, D., 2015. Repeatability and Reproducibility of the F-Scan System in Healthy Children. *Journal of Orthopedics, Rheumatology and Sports Medicine*.
- Collins, R.T., Gross, R. and Shi, J., 2002, May. Silhouette-based human identification from body shape and gait. In *Proceedings of fifth IEEE international conference on automatic face gesture recognition* (pp. 366-371). IEEE.
- Corbeil, P., Simoneau, M., Rancourt, D., Tremblay, A. and Teasdale, N., 2001. Increased risk for falling associated with obesity: mathematical modeling of postural control. *IEEE Transactions on Neural Systems and Rehabilitation Engineering*, 9(2), pp.126-136.
- Cornwall, M.W. and McPoil, T.G., 1997. Effect of foot orthotics on the initiation of plantar surface loading. *The Foot*, 7(3), pp.148-152.
- Cousins, S.D., Morrison, S.C. and Drechsler, W.I., 2012. The reliability of plantar pressure assessment during barefoot level walking in children aged 7-11 years. *Journal of foot and ankle research*, *5*(1), p.8.
- Cousins, S.D., Morrison, S.C. and Drechsler, W.I., 2013. Foot loading patterns in normal weight, overweight and obese children aged 7 to 11 years. *Journal of foot and ankle research*, 6(1), p.36.
- Cramer, G.D., Cantu, J.A., Pocius, J.D., Cambron, J.A. and McKinnis, R.A., 2010. Reliability of zygapophysial joint space measurements made from magnetic resonance imaging scans of acute low back pain subjects: comparison of 2 statistical methods. *Journal of manipulative and physiological therapeutics*, 33(3), pp.220-225.
- da Rocha, E.S., Bratz, D.T.K., Gubert, L.C., de David, A. and Carpes, F.P., 2014. Obese children experience higher plantar pressure and lower foot sensitivity than non-obese. *Clinical Biomechanics*, 29(7), pp.822-827.
- da Silva-Hamu, T.C.D., Formiga, C.K.M.R., Gervásio, F.M., Ribeiro, D.M., Christofoletti, G. and de França Barros, J., 2013. The impact of obesity in the kinematic parameters of gait in young women. *International journal of general medicine*, 6, p.507.
- Daentzer, D., Wülker, N. and Zimmermann, U., 1997. Observations concerning the transverse metatarsal arch. *Foot and Ankle Surgery*, *3*(1), pp.15-20.

- Dananberg, H.J., 2000. Sagittal plane biomechanics. American Diabetes Association. Journal of the American Podiatric Medical Association, 90(1), pp.47-50.
- Dario, A.B., Ferreira, M.L., Refshauge, K.M., Lima, T.S., Ordonana, J.R. and Ferreira, P.H., 2015. The relationship between obesity, low back pain, and lumbar disc degeneration when genetics and the environment are considered: a systematic review of twin studies. *The spine journal*, 15(5), pp.1106-1117.
- de Castro, M.P., Abreu, S.C., Sousa, H., Machado, L., Santos, R. and Vilas-Boas, J.P., 2014. In-shoe plantar pressures and ground reaction forces during overweight adults' overground walking. *Research quarterly for Exercise and Sport*, 85(2), pp.188-197.
- de Sá Pinto, A.L., de Barros Holanda, P.M., Radu, A.S., Villares, S.M. and Lima, F.R., 2006. Musculoskeletal findings in obese children. *Journal of paediatrics and child health*, 42(6), pp.341-344.
- De Souza, S.A.F., Faintuch, J., Valezi, A.C., Sant'Anna, A.F., Gama-Rodrigues, J.J., de Batista Fonseca, I.C. and de Melo, R.D., 2005. Postural changes in morbidly obese patients. *Obesity surgery*, *15*(7), pp.1013-1016.
- de Souza, V.A., 2007. Design of insole using image base analysis. *The University of Tokyo*.
- Debbi, E.M., Wolf, A., Goryachev, Y., Yizhar, Z., Luger, E., Debi, R. and Haim, A., 2012. In-shoe center of pressure: Indirect force plate vs. direct insole measurement. *The Foot*, 22(4), pp.269-275.
- DeVita, P. and Hortobágyi, T., 2003. Obesity is not associated with increased knee joint torque and power during level walking. *Journal of biomechanics*, *36*(9), pp.1355-1362.
- Donner, A. and Koval, J.J., 1980. The estimation of intraclass correlation in the analysis of family data. Biometrics, pp.19-25.
- Dowling, A.M., Steele, J.R. and Baur, L.A., 2001. Does obesity influence foot structure and plantar pressure patterns in prepubescent children?. *International journal of obesity*, 25(6), p.845.
- Duarte, M. and Freitas, S.M., 2010. Revision of posturography based on force plate for balance evaluation. *Brazilian Journal of physical therapy*, *14*(3), pp.183-192.
- Dufek, J.S., Currie, R.L., Gouws, P.L., Candela, L., Gutierrez, A.P., Mercer, J.A. and Putney, L.G., 2012. Effects of overweight and obesity on walking characteristics in adolescents. *Human movement science*, 31(4), pp.897-906.
- Dufour, A.B., Broe, K.E., Nguyen, U.S.D., Gagnon, D.R., Hillstrom, H.J., Walker, A.H., Kivell, E. and Hannan, M.T., 2009. Foot pain: is current or past shoewear a factor?. Arthritis Care & Research, 61(10), pp.1352-1358.
- Ehlen, K.A., RAOUL F REISER, I.I. and Browning, R.C., 2011. Energetics and biomechanics of inclined treadmill walking in obese adults. *Medicine & Science in Sports & Exercise*, 43(7), pp.1251-1259.
- Elftman, H., 1969. Dynamic structure of the human foot. Artif Limbs, 13(1), pp.49-58.
- England, S.A. and Granata, K.P., 2007. The influence of gait speed on local dynamic stability of walking. *Gait & posture*, 25(2), pp.172-178.
- Fabris, S.M., Valezi, A.C., de Souza, S.A.F., Faintuch, J., Cecconello, I. and Junior, M.P., 2006. Computerized baropodometry in obese patients. *Obesity* surgery, 16(12), pp.1574-1578.
- Fallon, E.M., Tanofsky-Kraff, M., Norman, A.C., McDuffie, J.R., Taylor, E.D., Cohen, M.L., Young-Hyman, D., Keil, M., Kolotkin, R.L. and Yanovski, J.A.,

2005. Health-related quality of life in overweight and nonoverweight black and white adolescents. *The Journal of pediatrics*, *147*(4), pp.443-450.

- Femery, V., Hespel, J.M., Thevenon, A. and Moretto, P., 2003. A real time plantar pressure feedback device for foot unloading. *Journal of Medical Informatics & Technologies*, 6, pp.IT103-106.
- Filippin, N.T., Barbosa, V.L.P., Sacco, I.C.N. and Lobo da Costa, P.H., 2007. Effects of obesity on plantar pressure distribution in children. *Brazilian Journal of Physical Therapy*, *11*(6), pp.495-501.
- Firat, O.F., 2017. Characterization and fabrication of MWCNTs pressure sensors for pressure ulcers monitoring (Doctoral dissertation, University of Delaware).
- Foti, T., Davids, J.R. and Bagley, A., 2000. A biomechanical analysis of gait during pregnancy. *JBJS*, 82(5), pp.625-625.
- Frey, C. and Zamora, J., 2007. The effects of obesity on orthopaedic foot and ankle pathology. *Foot & ankle international*, 28(9), pp.996-999.
- Fryar, C.D., Carroll, M.D. and Ogden, C.L., 2014. Prevalence of overweight and obesity among children and adolescents: United States, 1963–1965 through 2011–2012. Atlanta, GA: National Center for Health Statistics.
- Fukano, M. and Fukubayashi, T., 2009. Motion characteristics of the medial and lateral longitudinal arch during landing. *European journal of applied physiology*, 105(3), pp.387-392.
- Fuller, E., 1996. Computerised gait evaluation. In *Clinical biomechanics of the lower extremities* (pp. 179-205). Mosby, London.
- Galli, M., Crivellini, M., Sibella, F., Montesano, A., Bertocco, P. and Parisio, C., 2000. Sit-to-stand movement analysis in obese subjects. *International journal of* obesity, 24(11), p.1488.
- Ganea, D., Mereuta, E., Veresiu, S., Rus, M. and Amortila, V., 2017. Experimental tests for foot pressure analysis during orthostatic position and gait. In *MATEC Web of Conferences*(Vol. 112, p. 08009). EDP Sciences.
- Garrow, A.P., Silman, A.J. and Macfarlane, G.J., 2004. The Cheshire Foot Pain and Disability Survey: a population survey assessing prevalence and associations. *Pain*, *110*(1-2), pp.378-384.
- Giacomozzi, C. and Martelli, F., 2006. Peak pressure curve: an effective parameter for early detection of foot functional impairments in diabetic patients. *Gait & posture*, 23(4), pp.464-470.
- Giacomozzi, C., Macellari, V., Leardini, A. and Benedetti, M.G., 2000. Integrated pressure-force-kinematics measuring system for the characterisation of plantar foot loading during locomotion. *Medical and Biological Engineering and Computing*, 38(2), pp.156-163.
- Gill, S.V., 2015. The impact of weight classification on safety: timing steps to adapt to external constraints. *Journal of musculoskeletal & neuronal interactions*, *15*(1), p.103.
- Gilleard, W. and Smith, T., 2007. Effect of obesity on posture and hip joint moments during a standing task, and trunk forward flexion motion. *International journal of obesity*, *31*(2), p.267.
- Gilleard, W., Crosbie, J. and Smith, R., 2002. Effect of pregnancy on trunk range of motion when sitting and standing. *Acta Obstetricia et Gynecologica Scandinavica*, 81(11), pp.1011-1020.
- Gilleard, W., Crosbie, J. and Smith, R., 2008. A longitudinal study of the effect of pregnancy on rising to stand from a chair. *Journal of Biomechanics*, 41(4), pp.779-787.
- Gilmour, J.C. and Burns, Y., 2001. The measurement of the medial longitudinal arch in children. *Foot & ankle international*, 22(6), pp.493-498.
- Goffar, S.L., Reber, R.J., Christiansen, B.C., Miller, R.B., Naylor, J.A., Rodriguez, B.M., Walker, M.J. and Teyhen, D.S., 2013. Changes in dynamic plantar pressure during loaded gait. *Physical therapy*, 93(9), pp.1175-1184.
- Gravante, G., Russo, G., Pomara, F. and Ridola, C., 2003. Comparison of ground reaction forces between obese and control young adults during quiet standing on a baropodometric platform. *Clinical Biomechanics*, *18*(8), pp.780-782.
- Greve, J., Alonso, A., Bordini, A.C.P. and Camanho, G.L., 2007. Correlation between body mass index and postural balance. *Clinics*, 62(6), pp.717-720.
- Guh, D.P., Zhang, W., Bansback, N., Amarsi, Z., Birmingham, C.L. and Anis, A.H., 2009. The incidence of co-morbidities related to obesity and overweight: a systematic review and meta-analysis. *BMC public health*, *9*(1), p.88.
- Guiotto, A., Sawacha, Z., Guarneri, G., Cristoferi, G., Avogaro, A. and Cobelli, C., 2013. The role of foot morphology on foot function in diabetic subjects with or without neuropathy. *Gait & posture*, 37(4), pp.603-610.
- Gurney, J.K., Kersting, U.G. and Rosenbaum, D., 2008. Between-day reliability of repeated plantar pressure distribution measurements in a normal population. *Gait & posture*, 27(4), pp.706-709.
- Gushue, D.L., Houck, J. and Lerner, A.L., 2005. Effects of childhood obesity on threedimensional knee joint biomechanics during walking. *Journal of Pediatric Orthopaedics*, 25(6), pp.763-768.
- Hageman, P.A., Pullen, C.H., Dizona, P., Schmidt, K. and Boeckner, L.S., 2013. Obesity and physical function in rural women who enroll in a lifestyle modification intervention for reducing blood pressure. *Cardiopulmonary physical therapy journal*, 24(4), p.4.
- Hahn, S.L., Borton, K.A. and Sonneville, K.R., 2018. Cross-sectional associations between weight-related health behaviors and weight misperception among US adolescents with overweight/obesity. *BMC public health*, *18*(1), p.514.
- Haidar, Y.M. and Cosman, B.C., 2011. Obesity epidemiology. *Clinics in colon and rectal surgery*, 24(04), pp.205-210.
- Haight, D.J., Lerner, Z.F., Board, W.J. and Browning, R.C., 2014. A comparison of slow, uphill and fast, level walking on lower extremity biomechanics and tibiofemoral joint loading in obese and nonobese adults. *Journal of Orthopaedic Research*, 32(2), pp.324-330.
- Hailey, D. and Tomie, J.A., 2000. An assessment of gait analysis in the rehabilitation of children with walking difficulties. *Disability and Rehabilitation*, 22(6), pp.275-280.
- Hamill, J., Caldwell, G.E. and Derrick, T.R., 1997. Reconstructing digital signals using Shannon's sampling theorem. Journal of Applied Biomechanics, 13(2), pp.226-238.
- Hamilton, W.J., 1967. Anatomy of the Foot. British medical journal, 1(5532), p.101.
- Hannink, J., Kautz, T., Pasluosta, C.F., Gaßmann, K.G., Klucken, J. and Eskofier, B.M., 2017. Sensor-based gait parameter extraction with deep convolutional neural networks. *IEEE journal of biomedical and health informatics*, 21(1), pp.85-93.
- Harada, T., Mori, T., Nishida, Y., Yoshimi, T. and Sato, T., 1999. Body parts positions and posture estimation system based on pressure distribution image. In Proceedings 1999 IEEE International Conference on Robotics and Automation (Cat. No. 99CH36288C) (Vol. 2, pp. 968-975). IEEE.

- Hayafune, N., Hayafune, Y. and Jacob, H.A.C., 1999. Pressure and force distribution characteristics under the normal foot during the push-off phase in gait. *The foot*, 9(2), pp.88-92.
- Haykin, S. and Van Veen, B., 2007. Signals and systems. John Wiley & Sons.
- Healy, A., Burgess-Walker, P., Naemi, R. and Chockalingam, N., 2012. Repeatability of WalkinSense® in shoe pressure measurement system: A preliminary study. *The Foot*, 22(1), pp.35-39.
- Hennig, E.M., Staats, A. and Rosenbaum, D., 1994. Plantar pressure distribution patterns of young school children in comparison to adults. *Foot & ankle international*, 15(1), pp.35-40.
- Hergenroeder, A.L., Brach, J.S., Otto, A.D., Sparto, P.J. and Jakicic, J.M., 2011. The influence of body mass index on self-report and performance-based measures of physical function in adult women. *Cardiopulmonary physical therapy journal*, 22(3), p.11.
- Hessert, M.J., Vyas, M., Leach, J., Hu, K., Lipsitz, L.A. and Novak, V., 2005. Foot pressure distribution during walking in young and old adults. *BMC geriatrics*, 5(1), p.8.
- Hicks, J.H., 1954. The mechanics of the foot: II. The plantar aponeurosis and the arch. *Journal of anatomy*, 88(Pt 1), p.25.
- Hill Jr, J.J. and Cutting, P.J., 1989. Heel pain and body weight. *Foot & ankle*, 9(5), pp.254-256.
- Hill, C.L., Gill, T.K., Menz, H.B. and Taylor, A.W., 2008. Prevalence and correlates of foot pain in a population-based study: the North West Adelaide health study. *Journal of foot and ankle research*, 1(1), p.2.
- Hills, A.P. and Parker, A.W., 1991. Gait characteristics of obese children. Archives of physical medicine and rehabilitation, 72(6), pp.403-407.
- Hills, A.P., Hennig, E.M., Byrne, N.M. and Steele, J.R., 2002. The biomechanics of adiposity–structural and functional limitations of obesity and implications for movement. *Obesity reviews*, 3(1), pp.35-43.
- Hills, A.P., Hennig, E.M., McDonald, M. and Bar-Or, O., 2001. Plantar pressure differences between obese and non-obese adults: a biomechanical analysis. *International journal of obesity*, 25(11), p.1674.
- Hita-Contreras, F., Martínez-Amat, A., Lomas-Vega, R., Alvarez, P., Mendoza, N., Romero-Franco, N. and Aránega, A., 2013. Relationship of body mass index and body fat distribution with postural balance and risk of falls in Spanish postmenopausal women. *Menopause*, 20(2), pp.202-208.
- Ho, I.J., Hou, Y.Y., Yang, C.H., Wu, W.L., Chen, S.K. and Guo, L.Y., 2010. Comparison of plantar pressure distribution between different speed and incline during treadmill jogging. *Journal of sports science & medicine*, 9(1), p.154.
- Hopkins, W.G., 2000. Measures of reliability in sports medicine and science. Sports medicine, 30(1), pp.1-15.
- Horak, F.B., 2006. Postural orientation and equilibrium: what do we need to know about neural control of balance to prevent falls? *Age and ageing*, *35*(suppl_2), pp.ii7-ii11.
- Houweling, T., Bolton, J. and Newell, D., 2014. Comparison of two methods of collecting healthcare usage data in chiropractic clinics: patient-report versus documentation in patient files. *Chiropractic & manual therapies*, 22(1), p.32.
- Huang, L., Zhuang, J. and Zhang, Y.X., 2013. A method of speed control during overground walking: Using a digital light-emitting diode light strip. In Advanced Materials Research(Vol. 718, pp. 1371-1376). Trans Tech Publications.

- Hughes J, Pratt L, Linge K, et al. Reliability of pressure measurements: the EMED F system. Clin Biomech. 1991;6:14–18.
- Hulens, M., Vansant, G., Lysens, R., Claessens, A.L., Muls, E. and Brumagne, S., 2001. Study of differences in peripheral muscle strength of lean versus obese women: an allometric approach. *International journal of obesity*, 25(5), p.676.
- Irving, D.B., Cook, J.L., Young, M.A. and Menz, H.B., 2007. Obesity and pronated foot type may increase the risk of chronic plantar heel pain: a matched case-control study. *BMC musculoskeletal disorders*, 8(1), p.41.
- Jahss, M.H., 1982. Disorders of the Foot (Vol. 1). WB Saunders Company.
- Jegede, J.A., Adegoke, B.O. and Olagbegi, O.M., 2017. Effects of a Twelve-Week Weight Reduction Exercise Programme on Selected Spatiotemporal Gait Parameters of Obese Individuals. *Journal of obesity*, 2017.
- Jiang, L., Tian, W., Wang, Y., Rong, J., Bao, C., Liu, Y., Zhao, Y. and Wang, C., 2012. Body mass index and susceptibility to knee osteoarthritis: a systematic review and meta-analysis. *Joint Bone Spine*, 79(3), pp.291-297.
- Johnson, A.Y. and Bobick, A.F., 2001, June. A multi-view method for gait recognition using static body parameters. In *International Conference on Audio-and Video-Based Biometric Person Authentication* (pp. 301-311). Springer, Berlin, Heidelberg.
- Kadaba, M.P., Ramakrishnan, H.K., Wootten, M.E., Gainey, J., Gorton, G. and Cochran, G.V.B., 1989. Repeatability of kinematic, kinetic, and electromyographic data in normal adult gait. *Journal of Orthopaedic Research*, 7(6), pp.849-860.
- Kalamdani, A.A., 2006. Development and characterization of a high-spatial-temporalresolution foot-sole-pressure measurement system. *Master's thesis, The Robotics Institute Carnegie Mellon University Pittsburgh.*
- Kanatli, U., Yetkin, H. and Bolukbasi, S., 2003. Evaluation of the transverse metatarsal arch of the foot with gait analysis. *Archives of orthopaedic and trauma surgery*, *123*(4), pp.148-150.
- Kanatli, U., Yetkin, H., Simsek, A., Ozturk, A.M., Esen, E. and Besli, K., 2008. Pressure distribution patterns under the metatarsal heads in healthy individuals. *Acta Orthop Traumatol Turc*, 42(1), pp.26-30.
- Kavanagh, J. J. & Menz, H. B. (2008). Accelerometry: a technique for quantifying movement patterns during walking. Gait & Posture, 28(1), 1-15.
- Keijsers, N.L.W., Stolwijk, N.M. and Pataky, T.C., 2010. Linear dependence of peak, mean, and pressure–time integral values in plantar pressure images. *Gait & posture*, *31*(1), pp.140-142.
- Ker, R.F., Bennett, M.B., Bibby, S.R., Kester, R.C. and Alexander, R.M., 1987. The spring in the arch of the human foot. *Nature*, *325*(6100), p.147.
- Kernozek, T.W., LaMott, E.E. and Dancisak, M.J., 1996. Reliability of an in-shoe pressure measurement system during treadmill walking. Foot & Ankle International, 17(4), pp.204-209.
- Ki, S.W., Leung, A.K.L. and Li, A.N.M., 2008. Comparison of plantar pressure distribution patterns between foot orthoses provided by the CAD-CAM and foam impression methods. *Prosthetics and orthotics international*, 32(3), pp.356-362.
- King, W.C., Belle, S.H., Eid, G.M., Dakin, G.F., Inabnet, W.B., Mitchell, J.E., Patterson, E.J., Courcoulas, A.P., Flum, D.R., Chapman, W.H. and Wolfe, B.M., 2008. Physical activity levels of patients undergoing bariatric surgery in the

Longitudinal Assessment of Bariatric Surgery study. Surgery for Obesity and Related Diseases, 4(6), pp.721-728.

- Kirby, K.A., 2000. Biomechanics of the normal and abnormal foot. *Journal of the American Podiatric Medical Association*, 90(1), pp.30-34.
- Kirtley, C., 2006. *Clinical gait analysis: theory and practice*. Elsevier Health Sciences.
- Ko, M., Hughes, L. and Lewis, H., 2012. Walking speed and peak plantar pressure distribution during barefoot walking in persons with diabetes. *Physiotherapy Research International*, 17(1), pp.29-35.
- Ko, S.U., Stenholm, S. and Ferrucci, L., 2010. Characteristic gait patterns in older adults with obesity—Results from the Baltimore Longitudinal Study of Aging. *Journal of biomechanics*, 43(6), pp.1104-1110.
- Koo, T.K. and Li, M.Y., 2016. A guideline of selecting and reporting intraclass correlation coefficients for reliability research. *Journal of chiropractic medicine*, 15(2), pp.155-163.
- Koo, T.K., Cohen, J.H. and Zheng, Y., 2011. A mechano-acoustic indentor system for in vivo measurement of nonlinear elastic properties of soft tissue. *Journal of Manipulative and Physiological Therapeutics*, 34(9), pp.584-593.
- Kopelman, P.G., 2000. Obesity as a medical problem. Nature, 404(6778), p.635.
- Kortt, M. and Baldry, J., 2002. The association between musculoskeletal disorders and obesity. *Australian Health Review*, 25(6), pp.207-214.
- Kraushaar, D., 2005, 'Finding your running shoe matching foot type to shoe type', Hughston Research & Education Journal of Athletic Training, <http://www.hughston.com/hha/a_17_2_4.htm>.
- Krul, M., van der Wouden, J., Schellevis, F.G., van Suijlekom-Smit, L.W. and Koes, B.W., 2010. Musculoskeletal problems in overweight and obese children. *Swiss Medical Weekly*, 139(33-34), pp.78S-78S.
- Kunter, P.E. and Fleiss, J.L., 1979. Intraclass correlations: uses in assessing rater reliability. *Psychological bulletin*, 86(2), p.420.
- Kushner, R.F. and Roth, J.L., 2003. Assessment of the obese patient. *Endocrinology* and metabolism clinics of North America, 32(4), pp.915-933.
- Kutner, M.H., Nachtsheim, C.J., Neter, J. and Li, W., 2005. Applied linear statistical models (Vol. 103). Boston: McGraw-Hill Irwin.
- LaFiandra, M., Wagenaar, R.C., Holt, K.G. and Obusek, J.P., 2003. How do load carriage and walking speed influence trunk coordination and stride parameters?. *Journal of biomechanics*, *36*(1), pp.87-95.
- Lai, P.P., Leung, A.K., Li, A.N. and Zhang, M., 2008. Three-dimensional gait analysis of obese adults. *Clinical biomechanics*, 23, pp.S2-S6.
- Laroche, D., Duval, A., Morisset, C., Beis, J.N., d'Athis, P., Maillefert, J.F. and Ornetti, P., 2011. Test–retest reliability of 3D kinematic gait variables in hip osteoarthritis patients. *Osteoarthritis and cartilage*, 19(2), pp.194-199.
- Leach, R.A., Parker, P.L. and Veal, P.S., 2003. PulStar differential compliance spinal instrument: a randomized interexaminer and intraexaminer reliability study. *Journal of manipulative and physiological therapeutics*, 26(8), pp.493-501.
- Leardini, A., Benedetti, M.G., Catani, F., Simoncini, L. and Giannini, S., 1999. An anatomically based protocol for the description of foot segment kinematics during gait. *Clinical Biomechanics*, 14(8), pp.528-536.
- Ledoux, W.R. and Hillstrom, H.J., 2002. The distributed plantar vertical force of neutrally aligned and pes planus feet. *Gait & posture*, 15(1), pp.1-9.

- Lee, G.H., Park, S.B. and Lee, S.G., 1996. Analysis of the stance phase by measurement of plantar pressure. *Journal Korean Acad Rehab Med*, 20, pp.524-531.
- Lee, P.Y., Landorf, K.B., Bonanno, D.R. and Menz, H.B., 2014. Comparison of the pressure-relieving properties of various types of forefoot pads in older people with forefoot pain. *Journal of foot and ankle research*, 7(1), p.18.
- Lee, R. and Kean, W.F., 2012. Obesity and knee osteoarthritis. *Inflammopharmacology*, 20(2), pp.53-58.
- Lee, Y.S., 2009. The role of genes in the current obesity epidemic. *Annals Academy of Medicine Singapore*, 38(1), p.45.
- Lelas, J.L., Merriman, G.J., Riley, P.O. and Kerrigan, D.C., 2003. Predicting peak kinematic and kinetic parameters from gait speed. *Gait & posture*, 17(2), pp.106-112.
- Leong, K.S. and Wilding, J.P., 1999. Obesity and diabetes. *Best Practice & Research Clinical Endocrinology & Metabolism*, 13(2), pp.221-237.
- Lerner, Z.F., Board, W.J. and Browning, R.C., 2014. Effects of obesity on lower extremity muscle function during walking at two speeds. *Gait & posture*, *39*(3), pp.978-984.
- Levin, M.E. and O'Neal, L.W., 1988. The diabetic foot: pathophysiology, evaluation, and treatment. *The diabetic foot*, pp.1-50.
- Ling, C.G., Brotherton, S.S. and Smith, S.O., 2009. Review of the Literature Regarding Gait and Class III Obesity. *Journal of Exercise Physiology Online*, 12(5).
- Liu, T., Inoue, Y. and Shibata, K., 2009. A small and low-cost 3-D tactile sensor for a wearable force plate. *IEEE Sensors Journal*, *9*(9), pp.1103-1110.
- Liu, T., Inoue, Y. and Shibata, K., 2010. A wearable force plate system for the continuous measurement of triaxial ground reaction force in biomechanical applications. *Measurement Science and Technology*, 21(8), p.085804.
- Liu, T., Inoue, Y. and Shibata, K., 2010. A wearable ground reaction force sensor system and its application to the measurement of extrinsic gait variability. *Sensors*, *10*(11), pp.10240-10255.
- Lobet, S., Hermans, C., Bastien, G.J., Massaad, F. and Detrembleur, C., 2012. Impact of ankle osteoarthritis on the energetics and mechanics of gait: the case of hemophilic arthropathy. *Clinical Biomechanics*, 27(6), pp.625-631.
- Lochner, S.J., 2013. Computer aided engineering in the foot orthosis development process.
- Lord, M., 1997. Spatial resolution in plantar pressure measurement. *Medical* engineering & physics, 19(2), pp.140-144.
- Luo, Z.P., Berglund, L.J. and An, K.N., 1998. Validation of F-Scan pressure sensor system: a technical note. *Journal of rehabilitation research and development*, 35, pp.186-186.
- Lynn, S.K., Reid, S.M. and Costigan, P.A., 2007. The influence of gait pattern on signs of knee osteoarthritis in older adults over a 5–11 year follow-up period: a case study analysis. *The Knee*, *14*(1), pp.22-28.
- Lyytinen, T., Bragge, T., Liikavainio, T., Vartiainen, P., Karjalainen, P.A. and Arokoski, J.P., 2014. The impact of obesity and weight loss on gait in adults. In *The Mechanobiology of Obesity and Related Diseases* (pp. 125-147). Springer, Cham.
- Lyytinen, T., Liikavainio, T., Paakkonen, M., Gylling, H. and Arokoski, J.P., 2013. Physical function and properties of quadriceps femoris muscle after bariatric

surgery and subsequent weight loss. Journal Musculoskelet Neuronal Interact, 13(3), pp.329-38. -38.

- MacWilliams, B.A., Cowley, M. and Nicholson, D.E., 2003. Foot kinematics and kinetics during adolescent gait. *Gait & posture*, *17*(3), pp.214-224.
- Maetzler, M., Bochdansky, T. and Abboud, R.J., 2010. Normal pressure values and repeatability of the Emed® ST2 system. *Gait & Posture*, *32*(3), pp.391-394.
- Malatesta, D., Vismara, L., Menegoni, F., Galli, M., Romei, M. and Capodaglio, P., 2009. Mechanical external work and recovery at preferred walking speed in obese subjects. *Medicine & Science in Sports & Exercise*, 41(2), pp.426-434.
- Mao, D.W., Li, J.X. and Hong, Y., 2006. The duration and plantar pressure distribution during one-leg stance in Tai Chi exercise. *Clinical biomechanics*, 21(6), pp.640-645.
- Martin, P.E. and Nelson, R.C., 1986. The effect of carried loads on the walking patterns of men and women. *Ergonomics*, 29(10), pp.1191-1202.
- Martinez-Nova, A., Huerta, J.P. and Sanchez-Rodriguez, R., 2008. Cadence, age, and weight as determinants of forefoot plantar pressures using the Biofoot in-shoe system. *Journal of the American Podiatric Medical Association*, 98(4), pp.302-310.
- Mathie, M.J., Coster, A.C., Lovell, N.H. and Celler, B.G., 2004. Accelerometry: providing an integrated, practical method for long-term, ambulatory monitoring of human movement. *Physiological measurement*, 25(2), p.R1.
- Mathieson, I., Upton, D. and Birchenough, A., 1999. Comparison of footprint parameters calculated from static and dynamic footprints. *The Foot*, 9(3), pp.145-149.
- Mauch, M., Mickle, K.J., Munro, B.J., Dowling, A.M., Grau, S. and Steele, J.R., 2008. Do the feet of German and Australian children differ in structure? Implications for children's shoe design. *Ergonomics*, 51(4), pp.527-539.
- McAllister, E.J., Dhurandhar, N.V., Keith, S.W., Aronne, L.J., Barger, J., Baskin, M., Benca, R.M., Biggio, J., Boggiano, M.M., Eisenmann, J.C. and Elobeid, M., 2009. Ten putative contributors to the obesity epidemic. *Critical reviews in food science and nutrition*, 49(10), pp.868-913.
- McClean, K.M., Kee, F., Young, I.S. and Elborn, J.S., 2008. Obesity and the lung: 1. Epidemiology. *Thorax*, 63(7), pp.649-654.
- McCrory, J.L., Young, M.J., Boulton, A.J.M. and Cavanagh, P.R., 1997. Arch index as a predictor of arch height. *The Foot*, 7(2), pp.79-81.
- McGoey, B.V., Deitel, M., Saplys, R.J. and Kliman, M.E., 1990. Effect of weight loss on musculoskeletal pain in the morbidly obese. *The Journal of bone and joint surgery. British volume*, 72(2), pp.322-323.
- McKay, M.J., Baldwin, J.N., Ferreira, P., Simic, M., Vanicek, N., Wojciechowski, E., Mudge, A., Burns, J. and 1000 Norms Project Consortium, 2017. Spatiotemporal and plantar pressure patterns of 1000 healthy individuals aged 3–101 years. *Gait* & posture, 58, pp.78-87.
- McMaster, D.T., Gill, N., Cronin, J. and McGuigan, M., 2014. A brief review of strength and ballistic assessment methodologies in sport. Sports Medicine, 44(5), pp.603-623.
- McMillan, A.G., Auman, N.L., Collier, D.N. and Williams, D.S.B., 2009. Frontal plane lower extremity biomechanics during walking in boys who are overweight versus healthy weight. *Pediatric Physical Therapy*, 21(2), pp.187-193.

- McMillan, A.G., Pulver, A.M.E., Collier, D.N. and Williams, D.B., 2010. Sagittal and frontal plane joint mechanics throughout the stance phase of walking in adolescents who are obese. *Gait & posture*, *32*(2), pp.263-268.
- McPoil, T.G., Cornwall, M.W., Dupuis, L. and Cornwell, M., 1999. Variability of plantar pressure data. A comparison of the two-step and midgait methods. Journal of the American Podiatric Medical Association, 89(10), pp.495-501.
- McPoil, T.G., Yamada, W., Smith, W. and Cornwall, M., 2001. The distribution of plantar pressures in American Indians with diabetes mellitus. *Journal of the American Podiatric Medical Association*, 91(6), pp.280-287.
- Melai, T., IJzerman, T.H., Schaper, N.C., de Lange, T.L., Willems, P.J., Meijer, K., Lieverse, A.G. and Savelberg, H.H., 2011. Calculation of plantar pressure time integral, an alternative approach. *Gait & posture*, 34(3), pp.379-383.
- Menz, H.B. and Morris, M.E., 2006. Clinical determinants of plantar forces and pressures during walking in older people. *Gait & posture*, 24(2), pp.229-236.
- Menz, H.B., Morris, M.E. and Lord, S.R., 2006. Footwear characteristics and risk of indoor and outdoor falls in older people. *Gerontology*, 52(3), pp.174-180.
- Messier, S.P., 2010. Diet and exercise for obese adults with knee osteoarthritis. *Clinics in geriatric medicine*, 26(3), pp.461-477.
- Messier, S.P., Davies, A.B., Moore, D.T., Davis, S.E., Pack, R.J. and Kazmar, S.C., 1994. Severe obesity: effects on foot mechanics during walking. *Foot & ankle international*, 15(1), pp.29-34.
- Mezhov, V., Ciccutini, F.M., Hanna, F.S., Brennan, S.L., Wang, Y.Y., Urquhart, D.M. and Wluka, A.E., 2014. Does obesity affect knee cartilage? A systematic review of magnetic resonance imaging data. *Obesity reviews*, *15*(2), pp.143-157.
- Mickle, K.J. and Steele, J.R., 2015. Obese older adults suffer foot pain and footrelated functional limitation. *Gait & posture*, 42(4), pp.442-447.
- Mickle, K.J., Munro, B.J., Lord, S.R., Menz, H.B. and Steele, J.R., 2010. Foot pain, plantar pressures, and falls in older people: a prospective study. *Journal of the American Geriatrics Society*, 58(10), pp.1936-1940.
- Mickle, K.J., Munro, B.J., Lord, S.R., Menz, H.B. and Steele, J.R., 2011. Gait, balance and plantar pressures in older people with toe deformities. *Gait & posture*, *34*(3), pp.347-351.
- Mickle, K.J., Steele, J.R. and Munro, B.J., 2006. Does excess mass affect plantar pressure in young children?. *International Journal of Pediatric Obesity*, 1(3), pp.183-188.
- Mickle, K.J., Steele, J.R. and Munro, B.J., 2006. The feet of overweight and obese young children: are they flat or fat?. *Obesity*, *14*(11), pp.1949-1953.
- Mittlemeier, T.W.F. and Morlock, M., 1993, February. Pressure distribution measurements in gait analysis: dependency on measurement frequency. In 39th annual meeting of the Orthopaedic Research Society.
- Miyazaki, T., Wada, M., Kawahara, H., Sato, M., Baba, H. and Shimada, S., 2002. Dynamic load at baseline can predict radiographic disease progression in medial compartment knee osteoarthritis. *Annals of the rheumatic diseases*, *61*(7), pp.617-622.
- Moghadam, M., Ashayeri, H., Salavati, M., Sarafzadeh, J., Taghipoor, K.D., Saeedi, A. and Salehi, R., 2011. Reliability of center of pressure measures of postural stability in healthy older adults: effects of postural task difficulty and cognitive load. *Gait & Posture*, 33(4), pp.651-655.

- Mohd Said, A., Justine, M. and Manaf, H., 2016. Plantar pressure distribution among older persons with different types of foot and its correlation with functional reach distance. *Scientifica*, 2016.
- Mokdad, A.H., Ford, E.S., Bowman, B.A., Dietz, W.H., Vinicor, F., Bales, V.S. and Marks, J.S., 2003. Prevalence of obesity, diabetes, and obesity-related health risk factors, 2001. *Jama*, 289(1), pp.76-79.
- Monteiro, M., Gabriel, R., Aranha, J., e Castro, M.N., Sousa, M. and Moreira, M., 2010. Influence of obesity and sarcopenic obesity on plantar pressure of postmenopausal women. *Clinical Biomechanics*, 25(5), pp.461-467.
- Morag, E. and Cavanagh, P.R., 1999. Structural and functional predictors of regional peak pressures under the foot during walking. *Journal of biomechanics*, *32*(4), pp.359-370.
- Mueller, S., Carlsohn, A., Mueller, J., Baur, H. and Mayer, F., 2016. Influence of obesity on foot loading characteristics in gait for children aged 1 to 12 years. *PloS one*, 11(2), p.e0149924.
- Müller, S., Carlsohn, A., Müller, J., Baur, H. and Mayer, F., 2012. Static and dynamic foot characteristics in children aged 1–13 years: a cross-sectional study. Gait & posture, 35(3), pp.389-394.
- Munoz-Organero, M., Parker, J., Powell, L. and Mawson, S., 2016. Assessing walking strategies using insole pressure sensors for stroke survivors. *Sensors*, 16(10), p.1631.
- Muro-De-La-Herran, A., Garcia-Zapirain, B. and Mendez-Zorrilla, A., 2014. Gait analysis methods: An overview of wearable and non-wearable systems, highlighting clinical applications. *Sensors*, *14*(2), pp.3362-3394.
- Murray, H.J., Young, M.J., Hollis, S. and Boulton, A.J., 1996. The association between callus formation, high pressures and neuropathy in diabetic foot ulceration. *Diabetic Medicine*, 13(11), pp.979-982.
- Nair, B.M. and Kendricks, K.D., 2016. Deep network for analyzing gait patterns in low resolution video towards threat identification. *Electronic Imaging*, 2016(11), pp.1-8.
- Natali, A.N., Fontanella, C.G. and Carniel, E.L., 2010. Constitutive formulation and analysis of heel pad tissues mechanics. *Medical engineering & physics*, 32(5), pp.516-522.
- Netter, F.H. and Colacino, S., 1989. *Atlas of human anatomy*. Second ed. Basel: Ciba-Geigy Corporation. 1991:471.
- Noce, H.P., 2005. An in-shoe instrument for acquisition and storage of plantar pressure.MSc thesis, Oregon State University, USA.
- Nutt, J.G., Marsden, C.D. and Thompson, P.D., 1993. Human walking and higher-level gait disorders, particularly in the elderly. *Neurology*, 43(2), pp.268-268.
- Nyquist, H., 2002. Certain topics in telegraph transmission theory. Proceedings of the IEEE, 90(2), pp.280-305.
- Nyska, M., Linge, K., McCabe, C. and Klenerman, L., 1997. The adaptation of the foot to heavy loads: plantar foot pressures study. *Clinical biomechanics (Bristol, Avon)*, *12*(3), pp.S8-S8.
- Nyska, M., McCabe, C., Linge, K., Laing, P. and Klenerman, L., 1995. Effect of the shoe on plantar foot pressures. *Acta Orthopaedica Scandinavica*, 66(1), pp.53-56.
- Oatis, C.A., 2004. *Kinesiology: the mechanics and pathomechanics of human movement* (pp. 688-9). Philadelphia: Lippincott Williams & Wilkins.

- Obesity Rates & Trends Overview. (2015). Retrieved February 17, 2015, from http://stateofobesity.org/obesity-rates-trends-overview/
- Ogden, C.L., Carroll, M.D., Kit, B.K. and Flegal, K.M., 2014. Prevalence of childhood and adult obesity in the United States, 2011-2012. *Jama*, *311*(8), pp.806-814.
- Ong, K.L., Wu, B.J., Cheung, B.M., Barter, P.J. and Rye, K.A., 2013. Arthritis: its prevalence, risk factors, and association with cardiovascular diseases in the United States, 1999 to 2008. *Annals of epidemiology*, 23(2), pp.80-86.
- Orendurff, M.S., Rohr, E.S., Sangeorzan, B.J., Weaver, K. and Czerniecki, J.M., 2006. An equinus deformity of the ankle accounts for only a small amount of the increased forefoot plantar pressure in patients with diabetes. *The Journal of bone and joint surgery. British volume*, 88(1), pp.65-68.
- Orlin, M.N. and McPoil, T.G., 2000. Plantar pressure assessment. *Physical therapy*, 80(4), pp.399-409.
- Owens Jr, E.F., Hart, J.F., Donofrio, J.J., Haralambous, J. and Mierzejewski, E., 2004. Paraspinal skin temperature patterns: an interexaminer and intraexaminer reliability study. *Journal of manipulative and physiological therapeutics*, 27(3), pp.155-159.
- Özdemir, H., Söyüncü, Y., Ozgorgen, M. and Dabak, K., 2004. Effects of changes in heel fat pad thickness and elasticity on heel pain. Journal of the American Podiatric Medical Association, 94(1), pp.47-52.
- Ozdemir, H., Yilmaz, E., Murat, A., Karakurt, L., Poyraz, A.K. and Ogur, E., 2005. Sonographic evaluation of plantar fasciitis and relation to body mass index. *European journal of radiology*, *54*(3), pp.443-447.
- Pan, X., Bai, J.J., Sun, J., Ming, Y., Chen, L.R. and Wang, Z., 2016. The characteristics of walking strategy in elderly patients with type 2 diabetes. *International Journal of Nursing Sciences*, 3(2), pp.185-189.
- Park, W., Ramachandran, J., Weisman, P. and Jung, E.S., 2010. Obesity effect on male active joint range of motion. *Ergonomics*, 53(1), pp.102-108.
- Pataky, T.C., 2012. Spatial resolution in plantar pressure measurement revisited. *Journal of biomechanics*, 45(12), pp.2116-2124.
- Pataky, T.C., Caravaggi, P., Savage, R. and Crompton, R.H., 2008. Regional peak plantar pressures are highly sensitive to region boundary definitions. *Journal of Biomechanics*, 41(12), pp.2772-2775.
- Patry, J., Belley, R., Côté, M. and Chateau-Degat, M.L., 2013. Plantar pressures, plantar forces, and their influence on the pathogenesis of diabetic foot ulcers: a review. *Journal of the American Podiatric Medical Association*, 103(4), pp.322-332.
- Periyasamy, R., Mishra, A., Anand, S. and Ammini, A.C., 2012. Foot pressure distribution variation in pre-obese and non-obese adult subject while standing. *The Foot*, 22(4), pp.276-282.
- Perju-Dumbrava, D., Chiroban, O. and Radu, C.C., 2017. Obesity and overweight risk factors in sudden death due to cardiovascular causes: A case series. *Iranian journal of public health*, 46(6), p.856.
- Perry, J. and Davids, J.R., 1992. Gait analysis: normal and pathological function. *Journal of Pediatric Orthopaedics*, 12(6), p.815.
- Perry, J.E., Ulbrecht, J.S., Derr, J.A.P.D. and Cavanagh, P.R.P.D., 1995. The use of running shoes to reduce plantar pressures in patients who have diabetes. *JBJS*, 77(12), pp.1819-1828.
- Perttunen, J., 2002. Foot loading in normal and pathological walking (No. 83). University of Jyväskylä.

- Phethean, J. and Nester, C., 2012. The influence of body weight, body mass index and gender on plantar pressures: results of a cross-sectional study of healthy children's feet. *Gait & posture*, 36(2), pp.287-290.
- Phethean, J., Pataky, T.C., Nester, C.J. and Findlow, A.H., 2014. A cross-sectional study of age-related changes in plantar pressure distribution between 4 and 7 years: a comparison of regional and pixel-level analyses. *Gait & posture*, 39(1), pp.154-160.
- Pirozzi, K., McGuire, J. and Meyr, A.J., 2014. Effect of variable body mass on plantar foot pressure and off-loading device efficacy. *The Journal of Foot and Ankle Surgery*, 53(5), pp.588-597.
- Pollo, F.E., Otis, J.C., Backus, S.I., Warren, R.F. and Wickiewicz, T.L., 2002. Reduction of medial compartment loads with valgus bracing of the osteoarthritic knee. *The American Journal of Sports Medicine*, 30(3), pp.414-421.
- Ponta, M.L., Gozza, M., Giacinto, J., Gradaschi, R. and Adami, G.F., 2014. Effects of obesity on posture and walking: study prior to and following surgically induced weight loss. *Obesity surgery*, 24(11), pp.1915-1920.
- Prakash, C., Gupta, K., Mittal, A., Kumar, R. and Laxmi, V., 2015. Passive marker based optical system for gait kinematics for lower extremity. *Procedia Computer Science*, 45, pp.176-185.
- Prakash, C., Kumar, R. and Mittal, N., 2018. Recent developments in human gait research: parameters, approaches, applications, machine learning techniques, datasets and challenges. *Artificial Intelligence Review*, 49(1), pp.1-40.
- Price, C. and Nester, C., 2016. Foot dimensions and morphology in healthy weight, overweight and obese males. *Clinical Biomechanics*, *37*, pp.125-130.
- Price, C., Parker, D. and Nester, C.J., 2014, April. Validity and repeatability of three commercially available in-shoe pressure measurement systems. *In Journal of foot and ankle research*, 7 (1), p. A67. BioMed Central.
- Prichasuk, S., 1994. The heel pad in plantar heel pain. *The Journal of bone and joint surgery. British volume*, 76(1), pp.140-142.
- Putti, A.B., Arnold, G.P. and Abboud, R.J., 2010. Differences in foot pressures between Caucasians and Indians. *Foot and Ankle Surgery*, *16*(4), pp.195-198.
- Putti, A.B., Arnold, G.P., Cochrane, L.A. and Abboud, R.J., 2008. Normal pressure values and repeatability of the Emed® ST4 system. *Gait & posture*, 27(3), pp.501-505.
- Putz, R. and Pabst, R., 2006. Sobotta-Atlas of Human Anatomy: Head, Neck, Upper Limb, Thorax, Abdomen, Pelvis, Lower Limb; Two-volume set.
- Queen, R.M., Mall, N.A., Nunley, J.A. and Chuckpaiwong, B., 2009. Differences in plantar loading between flat and normal feet during different athletic tasks. *Gait & posture*, 29(4), pp.582-586.
- Ramachandra, P., Maiya, A.G. and Kumar, P., 2012. Test-retest reliability of the Win-Track platform in analyzing the gait parameters and plantar pressures during barefoot walking in healthy adults. *Foot & ankle specialist*, *5*(5), pp.306-312.
- Ramanathan, A.K., John, M.C., Arnold, G.P., Cochrane, L. and Abboud, R.J., 2008. The effects of off-the-shelf in-shoe heel inserts on forefoot plantar pressure. *Gait & posture*, 28(4), pp.533-537.
- Ramanathan, A.K., Kiran, P., Arnold, G.P., Wang, W. and Abboud, R.J., 2010. Repeatability of the Pedar-X[®] in-shoe pressure measuring system. *Foot and Ankle Surgery*, 16(2), pp.70-73.
- Rana, N.K., 2009, December. Application of force sensing resistor (FSR) in design of pressure scanning system for plantar pressure measurement. In 2009 Second

International Conference on Computer and Electrical Engineering (Vol. 2, pp. 678-685). IEEE.

- Randolph, A.L., Nelson, M., Akkapeddi, S., Levin, A. and Alexandrescu, R., 2000. Reliability of measurements of pressures applied on the foot during walking by a computerized insole sensor system. *Archives of physical medicine and rehabilitation*, 81(5), pp.573-578.
- Rao, S., Baumhauer, J.F. and Nawoczenski, D.A., 2011. Is barefoot regional plantar loading related to self-reported foot pain in patients with midfoot osteoarthritis. *Osteoarthritis and cartilage*, *19*(8), pp.1019-1025.
- Rao, S., Riskowski, J.L. and Hannan, M.T., 2012. Musculoskeletal conditions of the foot and ankle: assessments and treatment options. *Best Practice & Research Clinical Rheumatology*, 26(3), pp.345-368.
- Rathnayaka, A.M.A.I., Perera, W.N.D., Savindu, H.P., Madarasingha, K.C.M., Ranasinghe, S.P., Thuduwage, H.G.T.V., Kulathilaka, A.U., Silva, P., Jayasinghe, S., Kahaduwa, K.T.D. and De Silva, A.C., 2018, July. A Customized System to Assess Foot Plantar Pressure: A Case Study on Calloused and Normal Feet. In 2018 IEEE Region Ten Symposium (Tensymp) (pp. 202-206). IEEE.
- Razak, A., Hadi, A., Zayegh, A., Begg, R.K. and Wahab, Y., 2012. Foot plantar pressure measurement system: A review. *Sensors*, *12*(7), pp.9884-9912.
- Redmond, A.C., Crane, Y.Z. and Menz, H.B., 2008. Normative values for the foot posture index. *Journal of Foot and Ankle research*, *1*(1), p.6.
- Reeves, G.K., Pirie, K., Beral, V., Green, J., Spencer, E. and Bull, D., 2007. Cancer incidence and mortality in relation to body mass index in the Million Women Study: cohort study. *Bmj*, 335(7630), p.1134.
- Reiber, G., Lipsky, B. and Gibbons, G., 1998. The burden of diabetic foot ulcers. *The American journal of surgery*, *176*(2), pp.5S-10S.
- Richie Jr, D.H., 2007. Biomechanics and clinical analysis of the adult acquired flatfoot. *Clinics in podiatric medicine and surgery*, 24(4), pp.617-644.
- Riddiford-Harland, D.L., 2010. Childhood foot structure and function: is this influenced by obesity?.
- Riddiford-Harland, D.L., Steele, J.R. and Baur, L.A., 2011. Are the feet of obese children fat or flat? Revisiting the debate. *International Journal of Obesity*, 35(1), p.115.
- Riddle, D.L., Pulisic, M., Pidcoe, P. and Johnson, R.E., 2003. Risk factors for plantar fasciitis: a matched case-control study. *JBJS*, 85(5), pp.872-877.
- Riskowski, J.L., Hagedorn, T.J. and Hannan, M.T., 2011. Measures of foot function, foot health, and foot pain. *Arthritis care & research*, 63(0 11), p.S229.
- Robertson, D.G.E., Caldwell, G.E., Hamill, J., Kamen, G. and Whittlesey, S.N., 2013. *Research methods in biomechanics 2nd edition*. Human Kinetics.
- Robertson, G.E., Caldwell, G.E., Hamill, J., Kamen, G. and Whittlesey, S., 2013. Research methods in biomechanics. Human kinetics.
- Rodrigo, A.S., Goonetilleke, R.S. and Xiong, S., 2013. Load distribution to minimise pressure-related pain on foot: a model. *Ergonomics*, *56*(7), pp.1180-1193.
- Rosenbaum, D. and BECKER, H.P., 1997. Plantar pressure distribution measurements. Technical background and clinical applications. *Foot and ankle surgery*, *3*(1), pp.1-14.
- Rossouw, H.A., Grant, C.C. and Viljoen, M., 2012. Overweight and obesity in children and adolescents: The South African problem. *South African journal of science*, *108*(5-6), pp.31-37.

- Runhaar, J., Koes, B.W., Clockaerts, S. and Bierma-Zeinstra, S.M.A., 2011. A systematic review on changed biomechanics of lower extremities in obese individuals: a possible role in development of osteoarthritis. *Obesity reviews*, 12(12), pp.1071-1082.
- Rupérez Moreno, MJ 2011, Multidisciplinary techniques for the simulation of the contact between the foot and the shoe upper in gait: virtual reality, computational biomechanics, and artificial neural networks, PhD thesis, Polytechnic University of Valencia, Spain.
- Rupérez, M.J., Monserrat, C. and Alcañíz, M., 2009. A study of the viability of obtaining a generic animation of the foot while walking for the virtual testing of footwear using dorsal pressures. *Journal of Biomechanics*, 42(13), pp.2040-2046.
- Russell, B.S., Muhlenkamp, K.A., Hoiriis, K.T. and DeSimone, C.M., 2012. Measurement of lumbar lordosis in static standing posture with and without high-heeled shoes. *Journal of chiropractic medicine*, 11(3), pp.145-153.
- Russell, B.S., Muhlenkamp, K.A., Hoiriis, K.T. and DeSimone, C.M., 2012. Measurement of lumbar lordosis in static standing posture with and without high-heeled shoes. *Journal of chiropractic medicine*, 11(3), pp.145-153.
- Russell, E.M. and Hamill, J., 2011. Lateral wedges decrease biomechanical risk factors for knee osteoarthritis in obese women. *Journal of biomechanics*, 44(12), pp.2286-2291.
- Russell, E.M., Braun, B. and Hamill, J., 2010. Does stride length influence metabolic cost and biomechanical risk factors for knee osteoarthritis in obese women?. *Clinical Biomechanics*, 25(5), pp.438-443.
- Sainsbury, A. and Zhang, L., 2012. Role of the hypothalamus in the neuroendocrine regulation of body weight and composition during energy deficit. *Obesity reviews*, 13(3), pp.234-257.
- Sarkar, A.P.A.R.N.A., Singh, M.E.E.N.A.K.S.H.I., Bansal, N.I.T.E.S.H. and Kapoor, S.E.E.M.A., 2011. Effects of obesity on balance and gait alterations in young adults. *Indian journal Physiol. Pharmacol.*, 55, pp.227-233.
- Schaff, P.S., 1993. An overview of foot pressure measurement systems. Clinics in podiatric medicine and surgery, 10(3), pp.403-415.
- Scheinfeld, N.S., 2004. Obesity and dermatology. *Clinics in dermatology*, 22(4), pp.303-309.
- Scott, G., Menz, H.B. and Newcombe, L., 2007. Age-related differences in foot structure and function. Gait & posture, 26(1), pp.68-75.
- Segal, N.A., Yack, H.J. and Khole, P., 2009. Weight, rather than obesity distribution, explains peak external knee adduction moment during level gait. American journal of physical medicine & rehabilitation/Association of Academic Physiatrists, 88(3), p.180.
- Sheehan, K.J. and Gormley, J., 2013. The influence of excess body mass on adult gait. *Clinical Biomechanics*, 28(3), pp.337-343.
- Sheffler, L.R., Bailey, S.N., Gunzler, D. and Chae, J., 2014. Effect of body mass index on hemiparetic gait. *Physical Medicine and Rehabilitation*, 6(10), pp.908-913.
- Shultz, S.P., D'Hondt, E., Lenoir, M., Fink, P.W. and Hills, A.P., 2014. The role of excess mass in the adaptation of children's gait. *Human movement science*, *36*, pp.12-19.
- Shultz, S.P., D'hondt, E., Fink, P.W., Lenoir, M. and Hills, A.P., 2014. The effects of pediatric obesity on dynamic joint malalignment during gait. *Clinical biomechanics*, 29(7), pp.835-838.

- Shultz, S.P., Sitler, M.R., Tierney, R.T., Hillstrom, H.J. and Song, J., 2009. Effects of pediatric obesity on joint kinematics and kinetics during 2 walking cadences. Archives of physical medicine and rehabilitation, 90(12), pp.2146-2154.
- Sibella, F., Galli, M., Romei, M., Montesano, A. and Crivellini, M., 2003. Biomechanical analysis of sit-to-stand movement in normal and obese subjects. *Clinical biomechanics*, 18(8), pp.745-750.
- Silvernail, J.F., Milner, C.E., Thompson, D., Zhang, S. and Zhao, X., 2013. The influence of body mass index and velocity on knee biomechanics during walking. *Gait & posture*, *37*(4), pp.575-579.
- Simon, S.R., 2004. Quantification of human motion: gait analysis—benefits and limitations to its application to clinical problems. *Journal of biomechanics*, *37*(12), pp.1869-1880.
- Singh, B., Yack, H.J., Francis, S.L. and Janz, K.F., 2015. Biomechanical loads during common rehabilitation exercises in obese individuals. *International journal of sports physical therapy*, 10(2), p.189.
- Singh, D., Park, W., Levy, M.S. and Jung, E.S., 2009. The effects of obesity and standing time on postural sway during prolonged quiet standing. *Ergonomics*, 52(8), pp.977-986.
- Slim, F.J., van Schie, C.H., Keukenkamp, R., Faber, W.R. and Nollet, F., 2012. Increased plantar foot pressure in persons affected by leprosy. *Gait & posture*, 35(2), pp.218-224.
- Smith, B., Ashton, K.M., Bohl, D., Clark, R.C., Metheny, J.B. and Klassen, S., 2006. Influence of carrying a backpack on pelvic tilt, rotation, and obliquity in female college students. *Gait & posture*, 23(3), pp.263-267.
- Song, J., Kane, R., Tango, D.N., Vander Veur, S.S., Furmato, J., Komaroff, E. and Foster, G.D., 2015. Effects of weight loss on foot structure and function in obese adults: a pilot randomized controlled trial. *Gait & posture*, *41*(1), pp.86-92.
- Speksnijder, C.M., vd Munckhof, R.J., Moonen, S.A. and Walenkamp, G.H., 2005. The higher the heel the higher the forefoot-pressure in ten healthy women. *The foot*, 15(1), pp.17-21.
- Spyropoulos, P., Pisciotta, J.C., Pavlou, K.N., Cairns, M.A. and Simon, S.R., 1991. Biomechanical gait analysis in obese men. Archives of physical medicine and rehabilitation, 72(13), pp.1065-1070.
- Stefanyshyn, D.J. and Nigg, B.M., 1997. Mechanical energy contribution of the metatarsophalangeal joint to running and sprinting. *Journal of biomechanics*, 30(11-12), pp.1081-1085.
- Stenholm, S., Alley, D., Bandinelli, S., Griswold, M.E., Koskinen, S., Rantanen, T., Guralnik, J.M. and Ferrucci, L., 2009. The effect of obesity combined with low muscle strength on decline in mobility in older persons: results from the InCHIANTI study. *International journal of obesity*, 33(6), p.635.
- Stovitz, S.D., Pardee, P.E., Vazquez, G., Duval, S. and Schwimmer, J.B., 2008. Musculoskeletal pain in obese children and adolescents. *Acta paediatrica*, 97(4), pp.489-493.
- Stuck, R.M., Sohn, M.W., Budiman-Mak, E., Lee, T.A. and Weiss, K.B., 2008. Charcot arthropathy risk elevation in the obese diabetic population. *The American journal of medicine*, *121*(11), pp.1008-1014.
- Sumiya, T., Suzuki, Y., Kasahara, T. and Ogata, H., 1998. Sensing stability and dynamic response of the F-Scan in-shoe sensing system: a technical note. *Journal of rehabilitation research and development*, *35*(2), p.192.

REFERENCES

- Switaj, T.L. and O'connor, F.G., 2008. Chapter 43-Gait Analysis. *The Sports Medicine Resource Manual. Philadelphia: WB Saunders.*
- Taylor, A.J., Menz, H.B. and Keenan, A.M., 2004. The influence of walking speed on plantar pressure measurements using the two-step gait initiation protocol. *The foot*, *14*(1), pp.49-55.
- Thing, J., Maruthappu, M. and Rogers, J., 2012. Diagnosis and management of plantar fasciitis in primary care. *Br J Gen Pract*, 62(601), pp.443-444.
- Thomas, M.J., Roddy, E., Zhang, W., Menz, H.B., Hannan, M.T. and Peat, G.M., 2011. The population prevalence of foot and ankle pain in middle and old age: a systematic review. *Pain*, *152*(12), pp.2870-2880.
- Titianova, E.B., Mateev, P.S. and Tarkka, I.M., 2004. Footprint analysis of gait using a pressure sensor system. *Journal of Electromyography and Kinesiology*, 14(2), pp.275-281.
- Tong, J.W. and Kong, P.W., 2013. Reliability of footprint geometric and plantar loading measurements in children using the Emed® M system. *Gait & posture*, 38(2), pp.281-286.
- Urry, S., 1999. Plantar pressure-measurement sensors. *Measurement Science and Technology*, 10(1), p.R16.
- Valderrabano, V., Nigg, B.M., von Tscharner, V., Stefanyshyn, D.J., Goepfert, B. and Hintermann, B., 2007. Gait analysis in ankle osteoarthritis and total ankle replacement. *Clinical Biomechanics*, 22(8), pp.894-904.
- van der Leeden, M., Dekker, J.H., Siemonsma, P.C., Lek-Westerhof, S.S. and Steultjens, M.P., 2004. Reproducibility of plantar pressure measurements in patients with chronic arthritis: a comparison of one-step, two-step, and three-step protocols and an estimate of the number of measurements required. *Foot & ankle international*, 25(10), pp.739-744.
- Vela, S.A., Lavery, L.A., Armstrong, D.G. and Anaim, A.A., 1998. The effect of increased weight on peak pressures: implications for obesity and diabetic foot pathology. *The Journal of foot and ankle surgery*, 37(5), pp.416-420.
- Vereecke, E., D'Août, K., De Clercq, D., Van Elsacker, L. and Aerts, P., 2003. Dynamic plantar pressure distribution during terrestrial locomotion of bonobos (Pan paniscus). American Journal of Physical Anthropology: The Official Publication of the American Association of Physical Anthropologists, 120(4), pp.373-383.
- Vieira, A., Ribeiro, B. and Ferreira, J.P., 2017. GAIT analysis: methods & data review. *Cisuc tecnhical report TR-2017–004*.
- Vijayakumar, K., Kumar, S.S. and Subramanian, R., 2016. A study on relationship between BMI and prevalence of flat foot among the adults using foot print parameters. *International Journal of Advanced Research*, 4(5), pp.1428-1431.
- Villarroya, M.A., Esquivel, J.M., Tomás, C., Moreno, L.A., Buenafé, A. and Bueno, G., 2009. Assessment of the medial longitudinal arch in children and adolescents with obesity: footprints and radiographic study. *European journal of pediatrics*, 168(5), pp.559-567.
- Vincent, H.K., Ben-David, K., Cendan, J., Vincent, K.R., Lamb, K.M. and Stevenson, A., 2010. Effects of bariatric surgery on joint pain: a review of emerging evidence. Surgery for Obesity and Related Diseases, 6(4), pp.451-460.
- Vismara, L., Romei, M., Galli, M., Montesano, A., Baccalaro, G., Crivellini, M. and Grugni, G., 2007. Clinical implications of gait analysis in the rehabilitation of adult patients with" Prader-Willi" Syndrome: a cross-sectional comparative

study (" Prader-Willi" Syndrome vs matched obese patients and healthy subjects). *Journal of neuroengineering and rehabilitation*, 4(1), p.14.

- Waaijman, R. and Bus, S.A., 2012. The interdependency of peak pressure and pressure–time integral in pressure studies on diabetic footwear: no need to report both parameters. *Gait & Posture*, 35(1), pp.1-5.
- Waaijman, R., Arts, M.L.J., Haspels, R., Busch-Westbroek, T.E., Nollet, F. and Bus, S.A., 2012. Pressure-reduction and preservation in custom-made footwear of patients with diabetes and a history of plantar ulceration. *Diabetic Medicine*, 29(12), pp.1542-1549.
- Wafai, L., Zayegh, A., Woulfe, J., Aziz, S. and Begg, R., 2015. Identification of foot pathologies based on plantar pressure asymmetry. *Sensors*, 15(8), pp.20392-20408.
- Walker, M. and Fan, H.J., 1998. Relationship between foot pressure pattern and foot type. *Foot & ankle international*, *19*(6), pp.379-383.
- Walsh, T.P., Butterworth, P.A., Urquhart, D.M., Cicuttini, F.M., Landorf, K.B., Wluka, A.E., Shanahan, E.M. and Menz, H.B., 2017. Increase in body weight over a two-year period is associated with an increase in midfoot pressure and foot pain. *Journal of foot and ankle research*, 10(1), p.31.
- Watkins, M.P. and Portney, L., 2009. *Foundations of clinical research: applications to practice*. Upper Saddle River, NJ: Pearson/Prentice Hall.
- Wearing, S.C., Grigg, N.L., Lau, H.C. and Smeathers, J.E., 2012. Footprint-based estimates of arch structure are confounded by body composition in adults. *Journal of Orthopaedic Research*, *30*(8), pp.1351-1354.
- Wearing, S.C., Hennig, E.M., Byrne, N.M., Steele, J.R. and Hills, A.P., 2006. Musculoskeletal disorders associated with obesity: a biomechanical perspective. *Obesity reviews*, 7(3), pp.239-250.
- Wearing, S.C., Hennig, E.M., Byrne, N.M., Steele, J.R. and Hills, A.P., 2006. The biomechanics of restricted movement in adult obesity. *Obesity reviews*, 7(1), pp.13-24.
- Wearing, S.C., Urry, S., Smeathers, J.E. and Battistutta, D., 1999. A comparison of gait initiation and termination methods for obtaining plantar foot pressures. *Gait* & posture, 10(3), pp.255-263.
- Wearing, S.C., Urry, S., Smeathers, J.E. and Battistutta, D., 1999. A comparison of gait initiation and termination methods for obtaining plantar foot pressures. Gait & posture, 10(3), pp.255-263.
- Whittle, M.W., 1999. Generation and attenuation of transient impulsive forces beneath the foot: a review. *Gait & posture*, *10*(3), pp.264-275.
- Whittle, M.W., 2007. Normal gait, Gait analysis: an introduction. in Gait Analysis (Fourth Edition), ed Edinburgh: Butterworth-Heinemann, pp. 47-100.
- Wills, M., 2004. Orthopedic complications of childhood obesity. *Pediatric physical therapy*, *16*(4), pp.230-235.
- Winter, D. A. (2009). *Biomechanics and Motor Control of Human Movement*. (Fourth Edition). Hoboken: John Wiley & Sons.
- Winter, D.A., 1995. Human balance and posture control during standing and walking. Gait & posture, 3(4), pp.193-214.
- World health organization (2016). "Obesity and overweight." Retrieved 10 June, 2018, from < http://www.who.int/news-room/fact-sheets/detail/obesity-andoverweight>.
- World Health Organization. Global strategy on diet, physical activity and health. What is overweight and obesity? [homepage on the Internet]. No date [cited 2011 Dec

11].Availablefrom:http://www.who.int/dietphysicalactivity/childhood_what/en/index.html.

- Wu, X. and Madigan, M.L., 2014. Impaired plantar sensitivity among the obese is associated with increased postural sway. *Neuroscience letters*, 583, pp.49-54.
- Xu, C., Wen, X.X., Huang, L.Y., Shang, L., Cheng, X.X., Yan, Y.B. and Lei, W., 2017. Normal foot loading parameters and repeatability of the Footscan® platform system. *Journal of foot and ankle research*, 10(1), p.30.
- Yakimovich, T., Kofman, J. and Lemaire, E., 2006, January. Design, construction and evaluation of an electromechanical stance-control knee-ankle-foot orthosis. In 2005 IEEE Engineering in Medicine and Biology 27th Annual Conference(pp. 2333-2340). IEEE.
- Yan, S., Ren, W., Liang, X. and Zhang, K., 2014, October. Gait characteristics of overweight and obese children with different ages. In 2014 7th International Conference on Biomedical Engineering and Informatics (pp. 184-188). IEEE.
- Yan, S.H., Zhang, K., Tan, G.Q., Yang, J. and Liu, Z.C., 2013. Effects of obesity on dynamic plantar pressure distribution in Chinese prepubescent children during walking. *Gait & posture*, 37(1), pp.37-42.
- Youdas, J.W., Carey, J.R. and Garrett, T.R., 1991. Reliability of measurements of cervical spine range of motion—comparison of three methods. *Physical therapy*, 71(2), pp.98-104.
- Yu, H., Qin, S.F., Wight, D.K. and Kang, J.S., 2007, September. Generation of 3D human models with different levels of detail through point-based simplification. In EUROCON 2007-The International Conference on" Computer as a Tool" (pp. 1982-1986).
- Yümin, E.T., Şimşek, T.T., Sertel, M. and Ankaralı, H., 2016. The effect of age and body mass index on plantar cutaneous sensation in healthy women. *Journal of physical therapy science*, 28(9), pp.2587-2595.
- Zammit, G.V., Menz, H.B. and Munteanu, S.E., 2010. Reliability of the TekScan MatScan® system for the measurement of plantar forces and pressures during barefoot level walking in healthy adults. *Journal of foot and ankle research*, *3*(1), p.11.
- Zeng, X., Ma, L., Lin, Z., Huang, W., Huang, Z., Zhang, Y. and Mao, C., 2017. Relationship between Kellgren-Lawrence score and 3D kinematic gait analysis of patients with medial knee osteoarthritis using a new gait system. *Scientific reports*, 7(1), p.4080.
- Zhu, H., Wertsch, J.J., Harris, G.F., Loftsgaarden, J.D. and Price, M.B., 1991. Foot pressure distribution during walking and shuffling. *Archives of Physical Medicine and Rehabilitation*, 72(6), pp.390-397.
- Zimmet, P.Z., Magliano, D.J., Herman, W.H. and Shaw, J.E., 2014. Diabetes: a 21st century challenge. *The lancet Diabetes & endocrinology*, 2(1), pp.56-64.
- Zoico, E., Di Francesco, V., Guralnik, J.M., Mazzali, G., Bortolani, A., Guariento, S., Sergi, G., Bosello, O. and Zamboni, M., 2004. Physical disability and muscular strength in relation to obesity and different body composition indexes in a sample of healthy elderly women. *International journal of obesity*, 28(2), p.234.

APPENDICES

[Appendices A - Human Research Ethics Committee]

OFFICE OF RESEARCH

Human Research Ethics Committee PHONE +61 7 4631 2690| FAX +61 7 4631 5555 EMAIL human.ethics@usg.edu.au

3 May 2018



Miss Suhad Kareem Rahi Al-Magsoosi Darling Heights 4350

Dear Suhad

The USQ Human Research Ethics Committee has recently reviewed your responses to the conditions placed upon the ethical approval for the project outlined below. Your proposal is now deemed to meet the requirements of the *National Statement on Ethical Conduct in Human Research (2007)* and full ethical approval has been granted.

| Approval No. | H18REA065 |
|---------------|---|
| Project Title | BMI variation and its effect on gait in overweight and obese adults |
| Approval date | 3 May 2018 |
| Expiry date | 3 May 2021 |
| Status | Approved with standard conditions |

The standard conditions of this approval are:

- responsibly conduct the project strictly in accordance with the proposal submitted and granted ethics approval, including any amendments made to the proposal;
- (b) advise the University (email: ResearchIntegrity@usq.edu.au) immediately of any complaint pertaining to the conduct of the research or any other issues in relation to the project which may warrant review of the ethical approval of the project;
- promptly report any adverse events or unexpected outcomes to the University (email: <u>ResearchIntegrity@usq.edu.au</u>) and take prompt action to deal with any unexpected risks;
- (d) make submission for any amendments to the project and obtain approval prior to implementing such changes;
- provide a progress 'milestone report' when requested and at least for every year of approval;
- (f) provide a final 'milestone report' when the project is complete;
- (g) promptly advise the University if the project has been discontinued, using a final 'milestone report'.

For (d) to (g) forms are available on the USQ ethics website: <u>https://www.usq.edu.au/current-students/academic/higher-degree-by-research-students/conducting-research/human-ethics/forms-resources</u>

Please note that failure to comply with the conditions of approval and the National Statement (2007), may result in withdrawal of approval for the project.

Yours sincerely,

Mrs Nikita Kok Ethics Officer

[Appendices B - Consent Form]

The University of Southern Queensland Consent Form To: Participants Full Project Title: BMI variation and its effect on gait in overweight and obese adults. Principal Researcher: Dr. Albert Chong Student Researcher: Miss Suhad Kareem Rahi Al-Magsoosi ID: H18REA065

I have read the Participant Information Sheet and the nature and purpose of the research project has been explained to me. I understand and agree to take part.

• I understand the purpose of the research project and my involvement in it.

 $\bullet \Box I$ understand that I may withdraw from the research project at any stage and that this will not affect my status now or in the future.

• \Box I confirm that I am over 18 years of age.

 $\bullet \Box I$ understand that while information gained during the study may be published, I will not be identified, and my personal results will remain confidential.

 $\bullet \Box I$ understand that the video footage recorded of my foot during the research will be stored in a password protected computer at the University of Southern

Queensland and access will only be granted to the researchers involved in the study.

• I understand that only my lower limbs (knee and foot) will be videotaped during the study.

Name of

Participant..... Phone...

Signed......Date.....

If you have any ethical concerns with the research

OFFICE OF RESEARCH

Human Research Ethics Committee

PHONE +61 7 4631 2690| FAX +61 7 4631 5555

EMAIL <u>human.ethics@usq.edu.au</u>

[Appendices C - The calibration procedure]

Performing a step calibration for a known weight:

The following procedure is used to perform a step calibration on the left and right *F-Scan* sensors. This procedure must be performed on each foot separately. The process below will calibrate the left foot first, and then the right.

- 1. Refer to the "Preparing the Patient" section, and get the subject ready for testing. Open a pair of new Real-time windows for the subject.
- 2. Select the force and pressure units to be use in the calibration. The default units are "pounds (force)" and "PSI (pressure)". If other units are required, select **Measurement Units** from the "Options" menu, and make the changes (see image below).

| Units of Measure | | | |
|---|--|---------------|--|
| Units of Length: | | OK | |
| Units of Force: | Kilogiams 💌 | Cancel | |
| Units of Pressure: | KPa 💌 | Help | |
| Kilograms format 0 0.000 0.0 0.0000 0.0 0.00000 | KPa format Image: 0 0.000 Image: 0.00 0.0000 Image: 0.00 0.00000 | Legend Scroll | |

3. Insert the participant weight:

| Walk Calibration | - Right | | |
|--|--------------|--|--|
| Push OK and then Record to make a movie. At the end of recording, the movie will be | | | |
| calibrated based | Cancel | | |
| Subject Weight: | 81 Kilograms | | |

4. Tools were clicked on the taskbar calibration was selected.



5. We are first going to calibrate the left foot. With the left real-time window highlighted (the title bar will be blue), click the "Step Calibration" icon on the toolbar.

| - Real-time Windows | | | | | |
|---------------------------------------|-------------|-----------|--|--|--|
| Left (un-calibrated) Calibrate | UnCalibrate | Load Save | | | |
| Right (un-calibrated) Calibrate | UnCalibrate | Load Save | | | |
| Extended Calibration For Point Only 🔽 | | | | | |
| Exit | Help | | | | |

6. The "Step Calibration" dialog opens (shown below right) with instructions on how to proceed. Enter the subject's weight in the "Subject Weight" field (shown below Left).

APPENDICES

| | Right Calibration - Left (uscalibrated) |
|--|--|
| | Calibration Curve |
| | Step Calibration - Left |
| Real-time Lett (un-ce Pight (un-ce Exit Exit | Please enter the subject's body weight in units of Kilograms. Subject Weight: Calibration Procedure Stand on the inght hoot and press Start. When prompted by the display, shift to standing on the left toot and hold this postion in as steady a memore as possible, using a vertical surface for balance if necessary, until the step calibration completes. Start by standing on the right foot. Start by standing on the right foot. Start Storp OK Cancel |
| | Adjust Senativity |
| | |

7. If the calibration is correctly performed, the following "Step Calibration" screen is displayed, showing a successful calibration. Click the **OK** button to accept the calibration for the left foot.

| ase ente | er the subject' | s body weig | ght in ur | iits of Kilograms. | | |
|--------------------------------|---|--|----------------------------------|---|--|-------|
| | Subject W | 'eight: 81 | | Kilograms | 🔽 Drift | |
| alibration | Procedure - | - | | | | |
| tand on tanding vertical | the right foot on the left foo surface for bi | and press ' It and hold alance if ne | Start'. \ this pos cessary | When prompted ition in as steady , until the step c | by the display, shift to a manner as possible, i alibration completes. | ising |
| | | | | | | |
| | STEP CAL | IBRATION | SUCC | SSFUL - PRES | S 'OK' TO APPLY | |
| | | | | | | |
| | | | | | | |
| | | | | | | |
| | | | | | | |
| | | | | | | |
| | | | | | | |
| | | | | | | |
| | | | | | | |
| | | | | | | |
| | | | | | | |
| | | | | | | |
| | | Chan | | OK | Connert | |

| Calibration - Left (calibrated | d) | × | |
|--------------------------------|--------------------------|-------------------------|--|
| Calibration Curve | | | |
| 3345.73 KPa (Saturation Pr | essure) | Exp. Legend | |
| 0 Celibration Points | 28 | 55 (Raw/Cell) | |
| (Viloarama' (Pau Cum) | (Number of Leaded Calls) | X17-11. | |
| | (Number of Loaded Cells) | | |
| 2340 | 164 | Step | |
| | | Frame | |
| | | Edit | |
| 1 | | Delete | |
| Units | | | |
| ОК | Cancel | Cell Area: 0.258064 cm2 | |
| Load Cal. File | Save Cal. File | Help | |
| Sensitivity: Default | | | |
| Adjust Sensitivity | | | |

5. Have the participant stand on their right foot (the foot to be off-loaded). Click the **Start** button. A timer bar starts to count down the calibration process. Notice the white bar located about 1/4 the way along this timer bar. This is the point at which the subject must switch feet and stand entirely on their left foot.

| tep Calibration - Right |
|--|
| Please enter the subject's body weight in units of Kilograms. |
| Subject Weight: 81 Kilograms |
| Calibration Procedure |
| Stand on the left foot and press 'Start'. When prompted by the display, shift to standing on the right foot and hold this position in as steady a manner as possible, using a vertical surface for balance if necessary, until the step calibration completes. |
| STATUS: COLLECTING STEP CALIBRATION DATA |
| Switch to standing on the right foot. |
| Start Stop OK Cancel |

6. When the white vertical bar on the timer is reached, have the patient switch feet and stand entirely on their left foot. This will occur rapidly, so care must be taken to have the subject switch at the correct moment. It is important that the subject's weight be entirely borne through the foot being calibrated (here the left foot).

APPENDICES

7. If the calibration is correctly performed, the following "Step Calibration" screen is displayed, showing a successful calibration. Click the **OK** button to accept the calibration for the left foot

| itep Calibration - Right | | | | | |
|---|--|--|--|--|--|
| Please enter the subject's body weight in units of Kilograms. | | | | | |
| Subject Weight: 81 Kilograms 🔽 Drift | | | | | |
| Calibration Procedure | | | | | |
| STEP CALIBRATION SUCCESSFUL - PRESS 'OK' TO APPLY | | | | | |
| | | | | | |
| | | | | | |
| | | | | | |
| | | | | | |
| Redo Stop OK Cancel | | | | | |
| | | | | | |

8. If an error occurs during the calibration process, or the incorrect foot is used during calibration, the following error messages are presented on-screen. The foot calibration will have to be redone

| F-Scan | Research TAM/STAM | |
|--------|---|------------|
| 1 | An error occurred during step calibration. Plea | ase retry. |

If the calibration is correctly performed, the following "Step Calibration" screen is displayed, showing a successful calibration. Click the **OK** button to accept the calibration for the left and right foot.